UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): October 8, 2020

Riviera Resources, Inc.

(Exact name of registrant specified in its charter)

Delaware (State or Other Jurisdiction Of Incorporation) 333-225927 (Commission File Number) 82-5121920 (I.R.S. Employer Identification No.)

717 Texas Avenue, Suite 2000 Houston, Texas (Address of principal executive offices)

77002 (Zip Code)

(281) 840-4000 (Registrant's telephone number, including area code)

NOT APPLICABLE

(Former name or former address, if changed since last report.)			
	ck the appropriate box below if the Form 8-K filing is in the provisions (see General Instruction A.2. below):	ntended to simultaneously satisfy the filin	g obligation of the registrant under any of the
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)		
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)		
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))		
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))		
Securities registered pursuant to Section 12(b) of the Act:			
	Title of each class	Trading Symbol(s)	Name of each exchange on which registered
	None	None	None
Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).			
Emerging growth company \Box			
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.			

Item 2.01 Completion of Acquisition or Disposition of Assets.

On October 8, 2020, Riviera Resources, Inc. ("Riviera" or the "Company"), completed the sale of all of the issued and outstanding equity interests (the "Blue Mountain Interests") of Blue Mountain Midstream LLC ("BMM"), a Delaware limited liability company and wholly owned subsidiary of the Company, contemplated by that certain Membership Interest Purchase Agreement, dated August 23, 2020, (the "Purchase Agreement"), by and between Linn Energy Holdco II, LLC ("Seller"), a wholly owned subsidiary of the Company, Rheinbund, LLC (the "Buyer") and, solely for purposes of Section 11.16 thereof, Citizen Energy Operating, LLC. Pursuant to the Purchase Agreement, the Company divested all of its Blue Mountain Interests for approximately \$111 million in cash (the "Blue Mountain Divestiture"). After customary closing adjustments, transaction costs, and repayment in full of the balance of the BMM senior secured revolving loan facility and accrued interest, the Company received net cash proceeds of approximately \$35 million.

The purchase price was determined based on arm's length negotiations. Prior to the Blue Mountain Divestiture, there were no material relationships outside of the ordinary course of business between the Buyer, on the one hand, and the Company or any of its affiliates, directors, officers or any associate of such directors or officers, on the other hand.

The foregoing description of the Purchase Agreement is qualified in its entirety by reference to the full text of the Purchase Agreement, a copy of which is filed as Exhibit 2.1 hereto and is incorporated herein by reference.

Item 7.01 Regulation FD Disclosure.

On October 13, 2020, the Company filed a press release announcing the Blue Mountain Divestiture, among other things. A copy of the press release is attached as Exhibit 99.1 to this report and is incorporated by reference herein.

Item 9.01 Financial Statements and Exhibits.

(b) Pro Forma financial information.

As previously reported, on October 1, 2020, the Company completed the sale of its interest in Mid-Continent properties located in Oklahoma (the "Mid-Continent Asset Sale") to Staghorn Petroleum II, LLC under a Purchase and Sale Agreement dated August 4, 2020, for net cash proceeds of approximately \$13 million.

The pro forma financial information required by this Item 9.01 is filed as Exhibit 99.2 hereto and is incorporated herein by reference. The unaudited pro forma consolidated and combined balance sheet as of June 30, 2020, has been prepared as if the Blue Mountain Divestiture and the Mid-Continent Asset Sale occurred on that date. The unaudited pro forma consolidated and combined statements of operations for the six months ended June 30, 2020, and for the year ended December 31, 2019, have been prepared as if the Blue Mountain Divestiture and the Mid-Continent Asset Sale occurred on January 1, 2019.

(d) Exhibits.

Exhibit <u>Number</u>	<u>Description</u>
2.1	<u>Purchase and Sale Agreement, dated August 23, 2020, by and between Linn Energy Holdco II, LLC, as seller, Rheinbund, LLC, as buyer, and Citizen Energy Operating, LLC.</u>
99.1	Press Release dated October 13, 2020.
99.2	The Unaudited Pro Forma Condensed Consolidated Balance Sheet of the Company as of June 30, 2020, and Pro Forma Condensed Consolidated Statements of Operations of the Company for the six months ended June 30, 2020, and for the year ended December 31, 2019.
104	Cover Page Interactive Data File (formatted in Inline XBRL and contained in Exhibit 101)

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: October 15, 2020 RIVIERA RESOURCES, INC.

By: /s/ David B. Rottino

Name: David B. Rottino

Title: President and Chief Executive Officer

MEMBERSHIP INTEREST PURCHASE AGREEMENT

DATED AUGUST 23, 2020

BY AND AMONG

RHEINBUND, LLC,

LINN ENERGY HOLDCO II LLC,

AND

SOLELY FOR PURPOSES OF SECTION 11.16 HEREOF,
CITIZEN ENERGY OPERATING, LLC

TABLE OF CONTENTS

ARTICLE I CERTAIN DEFINITIONS		
1.1	Certain Defined Terms	1
ARTICLE	II THE TRANSACTION	1
2.1	Purchase and Sale of Interests	1
2.2	Deposit	2
2.3	Purchase Price	2
2.4	Purchase Price Adjustments	3
2.5	Tax Treatment	5
2.6	Target Equity Awards	6
2.7	Withholding	7
2.8	Indemnity Escrow Fund	7
ARTICLE	III REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE SELLER	8
3.1	Organization, Good Standing, and Authority of the Seller	8
3.2	Title to the Interests	8
3.3	Consents; No Conflicts	9
3.4	Litigation	9
3.5	Broker's or Finder's Fees	9
ARTICLE	IV REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE TARGET COMPANIES	10
4.1	Organization, Good Standing, and Authority; Capitalization of the Target Companies	10
4.2	No Conflicts	11
4.3	Consents and Authorizations	11
4.4	Permits	11
4.5	Taxes	11
4.6	Compliance with Laws	14
4.7	Material Contracts	14
4.8	Intellectual Property	14
4.9	Broker's or Finder's Fees	15
4.10	Employee Benefits	15
4.11	Employee Matters	16
4.12	Financial Statements; Absence of Undisclosed Liabilities	17
4.13	Environmental Matters	18
4.14	Litigation	19
4.15	Property	19
4.16	Affiliate Transactions	21

4.17	Absence of Certain Changes	21
4.18	Insurance	21
4.19	Bank Accounts	22
4.20	Books and Records	22
4.21	Insolvency	22
4.22	Royalties	22
4.23	Mineral Owner Accommodation Requests	22
4.24	Throughput Data	22
4.25	Hydrocarbon Imbalances; Future Delivery of Hydrocarbons	22
4.26	Regulatory Status	22
4.27	Target Company Business	23
4.28	FCC Licenses	23
4.29	SWD Wells	23
4.30	No Other Representations and Warranties	23
ARTICLE V	/ REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE BUYER	24
5.1	Organization, Good Standing, and Authority of the Buyer	24
5.2	No Conflicts	24
5.3	Litigation	25
5.4	Bankruptcy	25
5.5	Broker's or Finder's Fees	25
5.6	Investment Intent	25
5.7	Sufficiency of Funds	25
5.8	Solvency	25
5.9	Independent Evaluation	26
5.10	Business Investigation; Disclaimer Regarding Projections	26
5.11	Debt Financing	27
5.12	Equity Financing	27
5.13	No Other Representations and Warranties	28
ARTICLE V	/I COVENANTS	28
6.1	Conduct of Business	28
6.2	Access to Information; Confidentiality	31
6.3	Contact with Customers, Suppliers and Other Business Relations	32
6.4	Non-Solicitation	33
6.5	Insurance	33
6.6	Efforts	33
6.7	Tax Matters	34
6.8	Continuation of Indemnity; D&O Tail Coverage	37
6.9	R&W Insurance Policy	37
6.10	Employees	38
6.11	Use of Name	40
6.12	Fees and Expenses	41
6.13	Retention of Records by the Seller	42

6.14	Public Announcements	42
6.15	No Solicitation of Other Bids	43
6.16	Payoff Letters	43
6.17	Seller Credit Support; Termination of Agreements and Intercompany Arrangements	43
6.18	Termination and Release of Claims	44
6.19	Litigation Cooperation	44
6.20	Financing Cooperation	45
6.21	Financing Efforts	48
6.22	Services	49
ARTICLE	VII CONDITIONS TO THE CLOSING	49
7.1	Conditions to Each Party's Obligation	49
7.2	Seller's Conditions	49
7.3	Buyer's Conditions	50
7.4	Frustration of Closing Conditions	51
ARTICLE	VIII CLOSING	51
8.1	Time and Place of Closing	51
8.2	Deliveries at the Closing	51
ARTICLE	IX TERMINATION	53
9.1	Termination at or Prior to the Closing	53
9.2	Effect of Termination	54
9.3	Return of Documentation and Confidentiality	56
ARTICLE	X SURVIVAL AND REMEDIES	56
10.1	Survival	56
10.2	Exclusive Remedy	57
10.3	Bold or Capitalized Letters	57
10.4	Disclaimer	57
ARTICLE	XI MISCELLANEOUS PROVISIONS	59
11.1	Assignment	59
11.2	Amendments and Waiver	59
11.3	Entire Agreement	59
11.4	Severability	60
11.5	Counterparts	60
11.6	Governing Law and Dispute Resolution	60
11.7	Notices and Addresses	61
11.8	Conflict Wavier; Attorney-Client Privilege	63
11.9	Rules of Construction; Joint Drafting	64
11.10	Offset	65

11.11	No Partnership; Third-Party Beneficiaries	65
11.12	Specific Performance	65
11.13	Disclosure Schedule	65
11.14	Non-Recourse	66
11.15	Release	67
11.16	Parent Guaranty	68

EXHIBITS

Exhibit A	Target Assignment Agreement
Exhibit B	Transition Services Agreement
Exhibit C	R&W Insurance Policy
Exhibit D	Estimated Closing Statement
Exhibit E	Escrow Agreement

SCHEDULES

Schedule A	Defined Terms
Schedule 1.1(a)	CapEx Budget
Schedule 1.1(b)	Illustrative Working Capital Schedule
Schedule 1.1(c)	Payoff Indebtedness
Schedule 1.1(d)	Liens
Schedule 1.1(e)	Post-Closing Notification
Schedule 1.1(f)	Gathering and Processing Systems
Schedule 1.1(g)	SWD Wells
Schedule 1.1(h)	Target Material Adverse Effect
Schedule 1.1(i)	Reimbursable Expenses
Schedule 1.1(j)	Designated Agreements
Schedule 1.1(k)	Current Tax Assets and Liabilities

Schedule 6.3 Customers

Schedule 6.18 Termination and Release of Claims

DISCLOSURE SCHEDULES

Disclosure Schedule 3.3(a)	Consents (Seller)
Disclosure Schedule 3.3(b)	No Conflicts (Seller)
Disclosure Schedule 3.4	Litigation
Disclosure Schedule 3.5	Broker's or Finder's Fees (Seller)
Disclosure Schedule 4.1(b)	Equity Interests (Target)
Disclosure Schedule 4.1(c)	Target Companies
Disclosure Schedule 4.2	No Conflicts (Target Companies)
Disclosure Schedule 4.3(b)	Required Notifications
Disclosure Schedule 4.3(c)	Required Third-Party Consents
Disclosure Schedule 4.3(d)	Authorizations
Disclosure Schedule 4.4	Permits
Disclosure Schedule 4.5(l)	Tax liability
Disclosure Schedule 4.6	Compliance with Laws

Disclosure Schedule 4.7(a) Material Contracts Disclosure Schedule 4.7(b) Material Contract Exceptions Disclosure Schedule 4.8 Intellectual Property Disclosure Schedule 4.9 Broker's or Finder's Fees Disclosure Schedule 4.10(a) **Business Benefit Plans** Disclosure Schedule 4.10(c) Target Company Benefit Plan Disclosure Schedule 4.10(e)

Disclosure Schedule 4.10(f)

Disclosure Schedule 4.11

Disclosure Schedule 4.12(a)

Disclosure Schedule 4.12(d)

Material Payments

280G Payments

Employee Matters

Financial Statements

Disclosure Schedule 4.12(d)

Indebtedness

Disclosure Schedule 4.13 Environmental Matters

Disclosure Schedule 4.14(a) Proceedings
Disclosure Schedule 4.14(b) Outstanding Orders

Disclosure Schedule 4.14(c) Disputes

Disclosure Schedule 4.15(a)(i) Owned Real Property
Disclosure Schedule 4.15(a)(ii) Leased Real Property

Disclosure Schedule 4.15(e) Personal Property Leases Exceptions

Disclosure Schedule 4.15(f) Assets

Disclosure Schedule 4.16 Affiliate Transactions
Disclosure Schedule 4.17 Absence of Certain Changes

Disclosure Schedule 4.18 Insurance Disclosure Schedule 4.19 **Bank Accounts** Disclosure Schedule 4.24 System Data Disclosure Schedule 4.25 Imbalances Disclosure Schedule 4.26 **Regulatory Status** Disclosure Schedule 4.29(a) SWD Wells (Capacity) SWD Wells (Integrity) Disclosure Schedule 4.29(b) Disclosure Schedule 6.1(a) Conduct of Business

Disclosure Schedule 6.10(f) Buyer Plans

Disclosure Schedule 6.17(a)

Disclosure Schedule 6.17(d)

Disclosure Schedule 7.3(c)

Intercompany Guarantees
Required Intercompany Bonds
Required Third-Party Consent

MEMBERSHIP INTEREST PURCHASE AGREEMENT

This Membership Interest Purchase Agreement (this "*Agreement*"), dated August 23, 2020 (the "*Execution Date*"), is by and among Rheinbund, LLC, a Delaware limited liability company (the "*Buyer*"), Linn Energy Holdco II LLC, a Delaware limited liability company (the "*Seller*"), and, solely for purposes of <u>Section 11.16</u> hereof, Citizen Energy Operating, LLC, a Delaware limited liability company ("*Parent*"). The Buyer and the Seller are sometimes referred to collectively herein as the "*Parties*" and individually as a "*Party*."

RECITALS:

WHEREAS, as of the Execution Date (as defined herein), the Seller directly owns all of the outstanding Equity Interests (other than the Target Equity Awards (as defined herein and which are Equity Interest Equivalents under which no Class B Units have been delivered) to be cancelled upon the Closing (as defined herein)) of Blue Mountain Midstream LLC, a Delaware limited liability company (the "*Target*" and such Equity Interests, the "*Interests*").

WHEREAS, the Target, together with the other Target Companies (as defined herein), own and operate the Business (as defined herein).

WHEREAS, the Buyer desires to purchase, and the Seller desires to sell to the Buyer, the Interests (as defined herein) for the consideration set forth below, subject to the terms and conditions of this Agreement.

AGREEMENT:

NOW, THEREFORE, in consideration of the representations, warranties, covenants and agreements contained in this Agreement, and intending to be legally bound, the Parties agree, and solely for purposes of <u>Section 11.16</u>, Parent agrees, as follows:

ARTICLE I CERTAIN DEFINITIONS

1.1 <u>Certain Defined Terms</u>. Capitalized terms used in this Agreement that are not defined in the text of the body of this Agreement shall have the respective meanings given such terms as set forth in <u>Schedule A</u> attached to this Agreement, and <u>Schedule A</u> is incorporated herein by reference with the same force and effect as if set forth herein in full.

ARTICLE II THE TRANSACTION

2.1 <u>Purchase and Sale of Interests</u>. Subject to and upon the terms and conditions of this Agreement, at the Closing, the Seller shall sell, transfer, assign and deliver to the Buyer, and the Buyer shall purchase and acquire from the Seller, the Interests free and clear of all Liens, except for Permitted Equity Encumbrances. The Interests to be acquired by the Buyer at the Closing shall constitute all of the outstanding Equity Interests in the Target.

2.2 Deposit.

- (a) On the Execution Date, contemporaneously with the execution and delivery of this Agreement by the Parties, the Buyer, the Seller and the Escrow Agent shall enter into the Escrow Agreement. No later than the first (1st) Business Day after the Execution Date, the Buyer shall deposit an amount equal to \$11,100,000 (together with any and all interest accrued thereon, the "*Deposit*") via wire transfer of immediately available funds in to the Deposit Escrow Account to be held in escrow and disbursed by the Escrow Agent only in accordance with the express terms of this Agreement and the Escrow Agreement. If there is any inconsistency between the terms of this Agreement and those of the Escrow Agreement, as between the Buyer and the Seller, the terms of this Agreement shall control.
 - (b) If the Closing occurs, the entirety of the Deposit shall be disbursed to the Seller in accordance with Section 8.2.
 - (c) If this Agreement is terminated prior to Closing, then the Deposit shall be retained or disbursed as provided in Section 9.2.

2.3 Purchase Price.

- (a) The unadjusted purchase price to be delivered by the Buyer to the Seller for the sale, transfer, assignment and delivery of the Interests shall be \$111,000,000 (including the Deposit) (the "*Unadjusted Purchase Price*"), which is payable in accordance with this <u>Section 2.3</u>, subject to adjustment pursuant to <u>Section 2.4</u> (as so adjusted, the "*Adjusted Purchase Price*"), and subject to allocation in accordance with <u>Section 2.5</u>.
 - (b) At the Closing, the Unadjusted Purchase Price shall be paid in the following manner:
 - (i) an amount equal to \$700,000 (the "*Indemnity Escrow Fund*") shall be deposited by the Buyer with the Escrow Agent, by wire transfer of immediately available funds in to the Indemnity Escrow Account to be held in escrow and disbursed by the Escrow Agent only in accordance with the express terms of this Agreement and the Escrow Agreement;
 - (ii) the Escrow Agent shall disburse from the Deposit Escrow Account to the Seller an amount equal to the Deposit in accordance with the joint release instructions to the Escrow Agent provided by the Seller and the Buyer pursuant to <u>Section 8.2(a)(ii)</u> and <u>Section 8.2(b)(iv)</u>, respectively;
 - (iii) the Buyer shall pay the Payoff Amounts to the payees under the Debt Payoff Letters in accordance with Section 8.2(b)(ii); and
 - (iv) the Buyer shall pay to the Seller, by wire transfer of immediately available funds to the account designated in writing by the Seller to the Buyer at Closing, an amount equal to (A) the Unadjusted Purchase Price (net of the Deposit), *minus* (B) the Indemnity Escrow Fund and *plus* (C) the Estimated Adjustment Amount (which shall result in a reduction of the Unadjusted Purchase Price if the Estimated Adjustment Amount is a negative number), in accordance with <u>Section 2.4(a)</u> (such resulting sum, the "Closing Amount").

2.4 Purchase Price Adjustments.

(a) The Seller shall prepare and deliver to the Buyer, at least five (5) Business Days prior to the Closing Date and at the sole expense of the Seller, a written preliminary settlement statement substantially in the form attached to this Agreement as Exhibit D (the "Estimated Closing Statement") setting forth in reasonable detail the Seller's good faith calculation of the Adjusted Purchase Price (including with respect to the components thereof), and the Adjustment Amount (such estimated amount, the "Estimated Adjustment Amount"), together with applicable wiring instructions. For the avoidance of doubt, the Estimated Closing Statement shall include an estimate, with reasonably detailed calculations, of the Measurement Time Working Capital (the "Estimated Working Capital"). The Estimated Closing Statement and the determinations contained therein shall be prepared in accordance with the Accounting Principles and otherwise in accordance with the terms of this Agreement. Within two (2) Business Days after the Buyer's receipt of the draft Estimated Closing Statement pursuant to this Section 2.4(a), the Buyer will deliver to the Seller a written report containing all changes that the Buyer proposes to the Estimated Closing Statement, if any, together with an explanation of such changes. On or before the date that is two (2) Business Days prior to the Closing Date, the Parties shall meet and discuss in good faith any proposed changes to the Estimated Closing Statement and attempt in good faith to agree upon the calculation of the Adjusted Purchase Price. The Estimated Closing Statement, as agreed upon in writing by the Parties, shall control for purposes of the Closing Amount; provided, that if the Buyer and the Seller do not agree in writing upon any or all of the adjustments set forth in the Estimated Closing Statement, then the amount of such disputed adjustment or adjustments used to calculate the Closing Amount shall be that amount set forth in the Estimated Closing Statement with respect to such disputed adjustm

(b) As soon as reasonably practicable after the Closing Date (and, in any event, within ninety (90) days after the Closing Date), the Buyer shall prepare and deliver to the Seller, at the sole expense of the Buyer, a closing statement (the "Post-Closing Statement") setting forth the proposed final calculation of the Adjusted Purchase Price (including with respect to the components thereof) and the Adjustment Amount and reasonable supporting documentation relating to such estimates. During the period between the Closing Date and the delivery of the Post-Closing Statement, the Seller and its Representatives shall cooperate in good faith with the Buyer and its Representatives and provide any additional data and information as the Buyer may reasonably request to prepare the Post-Closing Statement. For the avoidance of doubt, the Post-Closing Statement shall include the proposed final calculation, with reasonably detailed calculations, of the Measurement Time Working Capital (the "Closing Working Capital"). The Post-Closing Statement shall be prepared in a manner consistent with the Accounting Principles and otherwise in accordance with the terms of this Agreement, and the Buyer shall not make any changes to the assumptions underlying the Accounting Principles, except as otherwise agreed to in writing by the Parties. From and after the delivery of the Post-Closing Statement until the Adjustment Amount is finally determined hereunder, the Buyer and its Representatives shall cooperate in good faith with the Seller and its Representatives and provide any additional data and information as the Seller may reasonably request to verify the amounts reflected on the Post-Closing Statement. Not later than the thirtieth (30th) day following receipt of the Post-Closing Statement under this Section 2.4(b), the Seller may deliver to the Buyer a written notice (an "Adjustment Notice") containing any changes the Seller proposes to the Post-Closing Statement.

Such Adjustment Notice will describe the nature and amount of any such disagreement in reasonable detail and shall be accompanied by such supporting documentation as is available to the Seller. If the Seller does not deliver to the Buyer an Adjustment Notice within such thirty (30) day period, then the Seller will be deemed to have irrevocably accepted and agreed to all items in the Post-Closing Statement.

- (c) If an Adjustment Notice is timely delivered to the Buyer in accordance with <u>Section 2.4(b)</u>, then the Buyer and the Seller shall undertake to agree on the final Adjusted Purchase Price no later than fifteen (15) days after the date on which the Seller delivered such Adjustment Notice to the Buyer. If the final Adjusted Purchase Price is mutually agreed upon in writing by the Parties during such fifteen (15) day period, then the final Adjusted Purchase Price as so agreed will be considered conclusive and binding on the Seller and the Buyer.
- (d) In the event that the final Adjusted Purchase Price is not mutually agreed upon by the Parties during the fifteen (15) day period referenced in Section 2.4(c), then Grant Thornton LLP (the "Accounting Firm") will be jointly engaged by the Seller and the Buyer to resolve any disagreements with respect to the components of the final Adjusted Purchase Price (the "Unresolved Objections"). If such accounting firm does not agree to serve as the Accounting Firm within ten (10) days after written request from the Seller and the Buyer to serve, then the Seller and the Buyer will mutually select and engage an alternative internationally-recognized independent accounting firm. If the Seller and the Buyer are unable to agree within ten (10) days upon such firm or such firm declines to serve as the Accounting Firm, then the Buyer and the Seller shall request that the American Arbitration Association (Houston, Texas office) select an internationally-recognized independent accounting firm to serve as the Accounting Firm for resolution of the Unresolved Objections. In connection with the engagement of the Accounting Firm, the Seller and the Buyer will execute such engagement, indemnity and other agreements as the Accounting Firm may reasonably require as a condition to such engagement. As promptly as practicable, but in any event within thirty (30) days after the selection of the Accounting Firm and execution of such engagement, indemnity or other agreements and based solely on (i) a written submission provided by each of the Buyer and the Seller to the Accounting Firm within ten (10) days following the Accounting Firm's selection (and without independent investigation on the part of the Accounting Firm) and (ii) the terms and provisions of this Agreement, the Accounting Firm will determine whether the Post-Closing Statement requires adjustment in accordance with the Accounting Principles and the terms of this Agreement. Once appointed, the Accounting Firm shall have no ex parte communications with any of the Parties concerning the expert determination or the underlying dispute and shall only have communications with the Seller and/or the Buyer as provided in this Section 2.4(d). All communications between the Seller and/or the Buyer, on the one hand, and the Accounting Firm, on the other hand, shall be conducted in writing, with copies sent simultaneously to the Seller and the Buyer, as applicable, or at a meeting involving both the Seller and the Buyer where both the Seller and the Buyer have been provided at least five (5) Business Days' advance notice. The Accounting Firm shall be authorized to determine the Unresolved Objections by only selecting the components of the final Adjusted Purchase Price as proposed by the Seller (on the one hand) or by the Buyer (on the other hand), whichever is more accurate. In resolving the Unresolved Objections, the Accounting Firm will function as an expert and not an arbitrator. The Seller and the Buyer will each bear its own legal fees and other costs of presenting its case to the Accounting Firm. The fees and expenses of the Accounting Firm and of any enforcement of the determination of the Accounting Firm shall be borne by the Seller and the

Buyer in inverse proportion as they may prevail on the matters resolved by the Accounting Firm. The proportionate allocation shall be calculated on an aggregate basis based on the relative dollar values of the amounts in dispute and shall be determined by the Accounting Firm at the time the determination of such firm is rendered on the merits of the matters submitted. The determination of the Accounting Firm with respect to such dispute, including any allocation of fees, will be final, conclusive and binding on the Parties, absent manifest error. The date on which the final Adjusted Purchase Price is finally determined in accordance with this <u>Section 2.4</u> is referred to as the "*Determination Date*."

- (e) The Adjustment Amount, as finally determined pursuant to this <u>Section 2.4</u>, shall be referred to herein as the "*Final Adjustment Amount*". Within five (5) Business Days after the Determination Date, the Seller or the Buyer, as applicable, shall make a true-up payment to the other in accordance with this <u>Section 2.4(e)</u>, so that, after giving effect to such payment, the Seller and the Buyer are in the same position they would have been in had payments at the Closing been based on the finally determined Adjusted Purchase Price (without any interest on such true-up payment) as follows:
 - (i) If the Final Adjustment Amount is *greater* than the Estimated Adjustment Amount, then the Buyer shall promptly (but no later than five (5) Business Days after the Determination Date) tender payment to the Seller, in cash by wire transfer of immediately available funds an amount equal to the amount, in dollars, by which the Final Adjustment Amount is *greater* than the Estimated Adjustment Amount; or
 - (ii) If the Final Adjustment Amount is *less* than the Estimated Adjustment Amount, then the Seller shall promptly (but no later than five (5) Business Days after the Determination Date) tender payment to the Buyer, in cash by wire transfer of immediately available funds an amount equal to the amount, in dollars, by which the Final Adjustment Amount is *less* than the Estimated Adjustment Amount.

2.5 Tax Treatment.

- (a) The Parties agree to treat the sale and purchase of the Interests pursuant to this Agreement for U.S. federal and applicable state and local income tax purposes (but for avoidance of doubt, not for Transfer Tax purposes) as a sale by the Seller to the Buyer, and a purchase by the Buyer, of the Assets.
- (b) The Parties agree that, for U.S. federal income tax purposes (and applicable state and local Tax purposes), the Tax Consideration shall be allocated among the Assets in accordance with Section 1060 of the Code and the Treasury Regulations promulgated thereunder (the "Allocation Schedule"). Within ninety (90) days after the Determination Date, the Buyer shall deliver to the Seller a draft of the Allocation Schedule. Such draft Allocation Schedule shall be deemed final, conclusive and binding (a "Final Allocation Schedule") on the Parties unless the Seller delivers a written objection to the Buyer within thirty (30) days after the Seller's receipt of the draft Allocation Schedule. If the Seller objects in writing within such 30-day period, the Parties shall use their good faith efforts to agree on a Final Allocation Schedule for thirty (30) days after the Buyer's receipt of such objections. If the Parties cannot agree on the Allocation Schedule prior to the end of such thirty (30)-day period, then each Party shall be entitled to adopt its own position

regarding the Allocation Schedule. In the event that the Parties agree (or are deemed to agree) on a Final Allocation Schedule, the Seller and the Buyer shall (and shall cause their respective Affiliates to): (i) be bound by the Final Allocation Schedule agreed upon (or deemed to be agreed upon) under this Section 2.5(b); (ii) use their good faith efforts to update the agreed Final Allocation Schedule to reflect any adjustments to the purchase price (as determined for U.S. federal income tax purposes) in a manner consistent with the agreed Final Allocation Schedule and Section 1060 of the Code and the Treasury Regulations promulgated thereunder; (iii) prepare and file all Tax Returns (including IRS Form 8594) in a manner consistent with the agreed Final Allocation Schedule, as adjusted, for any Tax purpose (whether in any Tax Return, in any audit, examination or other Proceeding related to Taxes, or otherwise), unless required to do so by (x) a determination by IRS Appeals (or any similar level of state Tax administrative determination) or (y) a final ruling by a trial court; provided, however, that no Party or any of their respective Affiliates shall be unreasonably impeded in its ability and discretion to concede, negotiate, compromise, and/or settle any audit, examination or other Proceeding in connection with the agreed Final Allocation Schedule.

2.6 Target Equity Awards.

- (a) <u>Target RSUs</u>. At the Closing, each Target RSU outstanding immediately prior to the Closing, whether vested or unvested, will, without any action on the part of the Buyer, the Seller, the Target or the holder thereof, be cancelled and converted into the right to receive, as the sole consideration due in respect of such Target RSU, an amount in cash, without interest, equal to (i) the Class B Unit Value, multiplied by (ii) the total number of Interests subject to such Target RSU (the "*RSU Consideration*").
- (b) <u>Target PSUs</u>. At the Closing, each Target PSU outstanding immediately prior to the Closing, whether vested or unvested, will, without any action on the part of the Buyer, the Seller, the Target or the holder thereof, be cancelled and converted into the right to receive, as the sole consideration due in respect of such Target PSU, an amount in cash, without interest, equal to (i) the Class B Unit Value, multiplied by (ii) the total number of Interests subject to such Target PSU determined based on actual achievement of the applicable performance-based vesting conditions as of the Closing (the "**PSU Consideration**").
- (c) <u>Payment Procedures</u>. As soon as practicable after the Closing Date, but in any event no later than the date necessary in order to meet the Payment Deadline (as defined below), the Seller will pay the aggregate (i) RSU Consideration owed to all holders of Target RSUs; and (ii) PSU Consideration owed to all holders of Target PSUs, in each case, if any, such that, no later than sixty (60) calendar days following the Closing Date (the "*Payment Deadline*"), each of the applicable holders of Target Equity Awards will receive a payment from the Seller, through its payroll system or payroll provider, of all amounts required to be paid to such holder in respect of his or her Target Equity Awards that are cancelled and converted pursuant to <u>Section 2.6(a)</u> or <u>Section 2.6(b)</u>, as applicable, net of any withholding of Taxes required by applicable Tax Law; *provided*, that if any payment owed to a holder of Target Equity Awards pursuant to <u>Section 2.6(b)</u>, as applicable, cannot be made through the Seller's payroll system or payroll provider, then the Seller will, (A) by wire transfer or direct deposit, or (B) by a check sent by

overnight courier to such holder, provide such payment owed to such holder, net of any withholding of Taxes required by applicable Tax Law, by the Payment Deadline. Notwithstanding the foregoing, to the extent any such amounts relate to a Target RSU or Target PSU that is nonqualified deferred compensation subject to Section 409A of the Code, the Seller shall pay such amounts at the earliest time permitted under the terms of the applicable agreement, plan or arrangement relating to such Target RSU or Target PSU that will not trigger a Tax or penalty under Section 409A of the Code.

- (d) <u>Further Actions</u>. The Seller shall (i) cause the Target to take all actions reasonably necessary to effect the cancellation of the Target Equity Awards upon the Closing and to give effect to <u>Section 2.6</u> and (ii) from and after the Closing, defend, indemnify and hold harmless the Buyer Related Parties from and against any and all Losses, as incurred, arising out of or relating to (A) the cancellation and conversion of the Target Equity Awards, (B) the calculation or payment of, or failure to pay, the RSU Consideration or the PSU Consideration in accordance with the Organizational Documents of the Target and the terms and conditions of the Target Equity Awards, (C) any other failure by the Seller to take all actions reasonably necessary to give effect to <u>Section 2.6</u> or (D) any Claim asserted by any holder of the Target RSUs or Target PSUs with respect to any of the foregoing matters of <u>Section 2.6</u>.
- 2.7 Withholding. Notwithstanding any other provision of this Agreement to the contrary, the Buyer and the Escrow Agent (and any other Person acting on their behalf) shall be entitled to deduct and withhold from the amounts otherwise payable in connection with this Agreement or any Transaction Document such amounts as are required to be deducted or withheld with respect to the making of such payment under applicable Tax Law; provided, however, that the Buyer, the Escrow Agent or any other Person acting on their behalf, as applicable, shall use commercially reasonable efforts to notify the Seller in writing prior to deducting or withholding any amounts pursuant to this Section 2.7 that such deduction or withholding is required and provide the Seller with a reasonable opportunity to contest such deduction and withholding and/or provide appropriate documentation to permit such amounts to be paid without deduction and withholding or at a reduced rate of deduction and withholding. To the extent that amounts are so deducted or withheld and paid over to the appropriate Governmental Authority, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction or withholding was made.

2.8 Indemnity Escrow Fund.

(a) From and after the Closing, the Seller shall defend, indemnify and hold harmless the Buyer Related Parties from and against any and all Losses, as incurred, arising out of or relating to (A) Target's obligations under the Synthetic Equity Interest Agreement, including any payment obligation thereunder, or (B) any Claim asserted by any party to the Synthetic Equity Interest Agreement (or assignee or successor in interest thereto) with respect to such obligations or the transactions contemplated by this Agreement. In connection therewith, the Indemnity Escrow Fund shall be held in escrow, pursuant to the Escrow Agreement, until the Indemnity Escrow Fund Release Date; *provided* that, from and after the Closing (but subject to the provisions of the Escrow Agreement), in the event that any Buyer Related Party incurs any Losses with respect to the matters described in clause (A) or (B) of this Section 2.8(a), the Buyer shall be entitled to recover first from the Indemnity Escrow Fund and the Buyer and the Seller shall execute and deliver a joint written instruction to the Escrow Agent to release proceeds from the Indemnity Escrow Account

to the Buyer in respect of any such Losses. On the Indemnity Escrow Fund Release Date and in accordance with the terms of the Escrow Agreement, the Buyer and the Seller shall deliver a joint written instruction to the Escrow Agent, duly executed by an authorized officer of each such Party authorized to deliver such instructions under the Escrow Agreement, instructing the Escrow Agent to promptly disburse from the Indemnity Escrow Account to the Seller an amount equal to any remainder of the Indemnity Escrow Fund.

(b) If any Party fails to execute and deliver a joint written instruction to the Escrow Agent in accordance with the terms of this Agreement and the Escrow Agreement, the Escrow Agent shall act in accordance with any Order in favor of the Party to which a disbursement is to be made hereunder and thereunder.

ARTICLE III REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE SELLER

The Seller represents and warrants to the Buyer as of the Execution Date and the Closing Date as follows, except as set forth in the Disclosure Schedules:

3.1 Organization, Good Standing, and Authority of the Seller.

- (a) The Seller is a limited liability company duly formed, validly existing, and in good standing under the Laws of the State of Delaware and has all requisite power and authority to own, lease, and operate its properties and to carry on its business as now being conducted.
- (b) The Seller has the full right, power and authority to enter into and perform its obligations under this Agreement and the Transaction Documents to which it is or shall be a party and to sell, assign, transfer and deliver to the Buyer at the Closing the Interests. The execution, delivery, and performance of this Agreement and the Escrow Agreement have been, and the other Transaction Documents to which the Seller will be a party will be, prior to the Closing, duly authorized by all requisite limited liability company action on the part of the Seller. This Agreement and the Escrow Agreement have been, and the other Transaction Documents to which the Seller will be a party will be, prior to the Closing, duly executed and delivered by the Seller, and (assuming due authorization, execution and delivery hereof by the other parties hereto and thereto) such Transaction Documents constitute (in the case of this Agreement and the Escrow Agreement) or will constitute at the Closing (in the case of such other Transaction Documents), legal, valid and binding obligations of the Seller, enforceable against the Seller in accordance with their terms, subject, as to enforceability, to bankruptcy, insolvency, reorganization, moratorium and other laws of general applicability relating to or affecting creditors' rights and to general principles of equity (regardless of whether such enforceability is considered in a Proceeding in equity or at Law).
- 3.2 <u>Title to the Interests</u>. The Seller is the record and beneficial owner of, and has good and valid title to, the Interests, free and clear of any and all Liens, except for Permitted Equity Encumbrances. The Interests owned by the Seller constitute all of the outstanding Equity Interests of the Target (other than the Target Equity Awards to be cancelled upon the Closing). Upon the Closing (should the Closing occur), the Buyer will acquire good and valid title to all of the Interests, as the record and beneficial owner thereof, free and clear of any Liens, except for Permitted Equity Encumbrances. The Seller is not a party

to (a) any Equity Interest Equivalent or other Contract (other than this Agreement) that could require the Seller, the Target or, after the Closing, the Buyer or any of its Affiliates, to sell, transfer, or otherwise dispose of any Equity Interest of the Target (other than the Target Equity Awards to be cancelled upon the Closing) or (b) any voting trust, proxy, or other agreement or understanding with respect to the voting of any Equity Interest of the Target, in each case, other than this Agreement or the Organizational Documents of the Target.

3.3 Consents; No Conflicts.

- (a) Except as set forth in <u>Disclosure Schedule 3.3(a)</u>, neither (i) the execution and delivery by the Seller of this Agreement, any Transaction Document to which the Seller is a party, or any Transaction Document or other instrument required hereby or thereby to be executed and delivered by it at the Closing nor (ii) the performance by the Seller of its obligations hereunder or thereunder will require any consent, approval, authorization or waiver of, notice to, or filing with, any Governmental Authority or any other Person.
- (b) Except as set forth in <u>Disclosure Schedule 3.3(b)</u>, the execution and delivery of this Agreement and the Escrow Agreement and, as of the Closing, any other Transaction Document to which the Seller is or will be a party by the Seller, and the performance by the Seller of its obligations hereunder or thereunder will not conflict with, violate, or breach the terms of, cause a default (with or without notice or lapse of time or both) under, or give rise to a right of termination, cancellation, or acceleration of any obligation or to the loss of a benefit under, or result in creation of any Lien on any of the Interests under: (i) the Organizational Documents of the Seller; (ii) except with respect to any Required Third-Party Consents, any Contract or other instrument to which the Seller is a party or by which it or any of its properties or assets are bound; or (iii) assuming the Authorizations or Notifications referred to in <u>Section 4.3</u> are duly and timely obtained, any Law, Regulation or Order applicable to the Seller or any of its properties or assets, except, in the case of <u>clauses (ii)</u> and (<u>iii)</u>, such matters as would not, individually or in the aggregate, reasonably be expected to prevent or materially delay, make illegal or otherwise interfere with the ability of the Seller to consummate the transactions contemplated hereby.
- 3.4 <u>Litigation</u>. Except as set forth in <u>Disclosure Schedule 3.4</u>, there is no Proceeding pending or, to the Knowledge of the Seller, threatened against or affecting the Seller or any of its Affiliates that, individually or in the aggregate, questions or challenges the validity of this Agreement or that would reasonably be expected to prevent, materially delay, make illegal or otherwise materially interfere with the ability of the Seller to consummate the transactions contemplated hereby.
- 3.5 <u>Broker's or Finder's Fees</u>. Except as set forth in <u>Disclosure Schedule 3.5</u>, no investment banker, broker, finder, agent or other Person is entitled to any brokerage or finder's fee or similar commission in respect of the transactions contemplated by this Agreement or any Transaction Document based in any way on agreements, arrangements or understandings made by or on behalf of any Seller Party.

ARTICLE IV REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE TARGET COMPANIES

The Seller represents and warrants to the Buyer as of the Execution Date and the Closing Date as follows, except as set forth in the Disclosure Schedules:

- 4.1 Organization, Good Standing, and Authority; Capitalization of the Target Companies.
- (a) Each of the Target Companies is a legal entity duly formed, validly existing and in good standing under the Laws of the jurisdiction in which it is formed or organized, as applicable, and has all requisite limited liability company power and authority to carry on its business as it is now being conducted and to own, operate and lease the assets it now owns, operates, or holds under lease. Each of the Target Companies is duly qualified to transact business and is in good standing in the each of the jurisdictions in which it has assets or conducts activities that would require it to be so qualified or in good standing, except where failure to be so qualified or in good standing would not, individually or in the aggregate, reasonably be expected to have a Target Material Adverse Effect.
- (b) The Equity Interests set forth in <u>Disclosure Schedule 4.1(b)</u>, including the Interests, constitute all of the issued and outstanding Equity Interests of the Target. All of the outstanding Equity Interests of the Target have been duly authorized and validly issued (in accordance with the Organizational Documents of the Target) and are fully paid (to the extent required under the Organizational Documents of the Target), nonassessable (except as such non-assessability may be affected by applicable Law) and not subject to preemptive rights or other preferential rights of subscription or purchase (except those waived or fully complied with at or prior to the Closing). All of such Equity Interests were issued in compliance with applicable Law and were not issued in violation of the Organizational Documents of the Target. The Target has not issued, and is not required to issue, any Class B Units.
- (c) Each of the Target Companies is set forth in <u>Disclosure Schedule 4.1(c)</u>. All of the Equity Interests of each of the Target Subsidiaries are held of record, directly or indirectly, by the Target. Except as set forth in <u>Disclosure Schedule 4.1(c)</u>, all of the outstanding Equity Interests in each of the Target Subsidiaries have been duly authorized and validly issued (in accordance with the Organizational Documents of such Target Company) and are fully paid (to the extent required under the Organizational Documents of such Target Company), nonassessable (except as such non-assessability may be affected by applicable Law) and not subject to preemptive rights or other preferential rights of subscription or purchase (except those waived or fully complied with at or prior to the Closing), and are owned, directly or indirectly, by the Target free and clear of all Liens other than Permitted Equity Encumbrances. All of such Equity Interests were issued in compliance with applicable Law and were not issued in violation of the Organizational Documents of the applicable Target Company. Except for the Target's ownership of Equity Interests of its Subsidiaries and as set forth in <u>Disclosure Schedule 4.1(c)</u>, no Target Company owns, directly or indirectly, any Equity Interests of any Person.

- (d) The Seller has made available to the Buyer true, correct and complete copies of the Organizational Documents of each of the Target Companies, each as in effect on the Execution Date.
- 4.2 No Conflicts. Except as set forth in <u>Disclosure Schedule 4.2</u>, the execution and delivery of this Agreement and the Escrow Agreement and, as of the Closing, any other Transaction Document to which the Seller is or will be a party by the Seller, and the performance by the Seller of its obligations hereunder or thereunder will not conflict with, violate, or breach the terms of, cause a default (with or without notice or lapse of time or both) under, or give rise to a right of termination, cancellation, or acceleration of any obligation or to the loss of a benefit under, or result in creation of any Lien on any of the Interests or the Assets under: (a) the Organizational Documents of the Target Companies; (b) except with respect to any Required Third-Party Consents, any Contract or other instrument to which a Target Company is a party or by which or to which any of the Assets are bound or subject; or (c) assuming the Authorizations or Notifications referred to in <u>Section 4.3</u> are duly and timely obtained or made, any Law, Regulation or Order applicable to a Target Company or any of the Assets.
- 4.3 <u>Consents and Authorizations</u>. Except for (a) Post-Closing Notifications, (b) any Notifications set forth in <u>Disclosure Schedule 4.3(b)</u> (the "*Required Notifications*"), (c) any Third-Party Consent set forth in <u>Disclosure Schedule 4.3(c)</u> (a "*Required Third-Party Consent*"), (d) any Authorizations set forth in <u>Disclosure Schedule 4.3(d)</u>, and (e) Authorizations, Notifications or Third-Party Consents previously obtained or made, no Authorization, Notification or Third-Party Consent is necessary for the Seller to execute, deliver and perform its obligations under this Agreement and the Transaction Documents to which the Seller is or shall be a party.
- 4.4 <u>Permits</u>. Other than with respect to Permits issued pursuant to or required under Environmental Laws which are the subject of <u>Section 4.13</u> and except as set forth in <u>Disclosure Schedule 4.4</u>: (a) the Target Companies possess all material Permits as are necessary to carry on the Business as currently conducted and (b) all such Permits are in full force and effect and no Proceeding is pending or, to the Knowledge of the Seller, threatened to suspend, revoke, or terminate any such Permit or declare any such Permit invalid. No Seller Party has received any written notice alleging any material violation under any Permit held by the Target Companies and, to the Knowledge of the Seller, there are no investigations or reviews pending or threatened by any Governmental Authority relating to any alleged violation of the terms of any Permit arising out of the Business other than, in each case, those that have been resolved. The consummation of the transactions contemplated by this Agreement and the Transaction Documents will not cancel, suspend, terminate or otherwise require modification of any material Permit of the Target Companies.

4.5 Taxes.

- (a) All income and other Tax Returns required to be filed by or with respect to any of the Target Companies have been duly and timely filed, all such Tax Returns are true, correct and complete in all material respects, and all Taxes required to be paid by or with respect to any of the Target Companies have been timely paid in full (whether or not shown on any Tax Return).
- (b) Each of the Target Companies has withheld or collected and paid over to the appropriate Governmental Authorities all Taxes required by Law to be withheld or collected and each of the Target Companies has properly received and maintained any and all certificates, forms and other documents required by Law for any exemption from withholding and remitting any Taxes.

- (c) The unpaid Taxes of the Target Companies (i) did not, as of December 31, 2019, materially exceed the accrued Taxes payable (rather than any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the Year End Financial Statements (rather than in any notes thereto) and (ii) do not materially exceed that accrual as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of the Target Companies in filing their Tax Returns. Any reserve for deferred Taxes reflected on the Year End Financial Statements is adequate to cover any deferred Tax Liability of the Target Companies through the date of the Year End Financial Statements.
- (d) Other than Permitted Encumbrances, (i) there are no Liens for unpaid Taxes on any of the Assets and (ii) no Claim for unpaid Taxes has been made by any Governmental Authority that could give rise to any such Lien.
- (e) There are no Claims or audits, examinations or other Proceedings commenced, ongoing or pending or, to the Knowledge of the Seller, threatened with respect to (i) Taxes of or with respect to any Target Company or for which any Target Company may be liable or (ii) any Tax Return filed by or with respect to any Target Company. None of the Target Companies has received any notice from any Governmental Authority indicating an intent to open any audit, examination or other Proceeding with respect to Taxes that remains in effect under applicable Tax Law. No Claim or deficiency against any Target Company for the assessment or collection of any Taxes has been asserted or proposed against or with respect to any Target Company, which Claim or deficiency has not been settled with all amounts determined to have been due and payable having been timely paid in full.
- (f) No Claim has been made by any Governmental Authority in any jurisdiction where any Target Company does not file Tax Returns that such Target Company is, or may be, subject to Tax by that jurisdiction.
- (g) None of the Target Companies will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any Tax period (or portion thereof) ending after the Closing Date as a result of any: (i) change in method of accounting pursuant to Section 481(a) or Section 482 of the Code (or any corresponding or similar provision of state, local or foreign Tax Law); (ii) installment sale or open transaction made or entered into on or prior to the Closing Date; (iii) prepaid amount received or deferred revenue accrued on or prior to the Closing Date; (iv) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign Tax Law) executed on or prior to the Closing Date; (v) "subpart F income" (as defined in Section 952(a) of the Code), "global intangible low-taxed income" (as defined in Section 951A(b) of the Code) or amount determined under Section 956 of the Code; or (vi) use of an improper method of accounting for a Tax period ending on or prior to the Closing Date.
- (h) The Seller's regarded parent, Riviera Resources, Inc., is not a "foreign person" within the meaning of Section 1445 of the Code or a "disregarded entity" as defined in Treasury Regulations Section 1.1445-2(b)(2)(iii).

- (i) None of the Target Companies is a party to, bound by or has any obligation or potential Liability under any Tax allocation, sharing or indemnity Contract or other agreement or arrangement, or under any other similar Contract or other agreement or arrangement relating to Taxes.
- (j) None of the Target Companies has any permanent establishment (within the meaning of any applicable Tax treaty or convention) or office or fixed place of business in a country other than the United States.
- (k) None of the Target Companies has been a "distributing corporation" or a "controlled corporation" in connection with a distribution described, or purported to be described, in Section 355 or Section 361 of the Code in the last three years.
- (l) Except as set forth in <u>Disclosure Schedule 4.5(l)</u>, none of the Target Companies (i) is, or has within the last seven (7) years, been, a member of an affiliated, aggregated, combined, consolidated or unitary group or (ii) has, or within the last seven (7) years, has had, any Liability for the Taxes of any Person under Treasury Regulations Section 1.1502-6 (or any similar or corresponding provision of foreign, state or local Law), as a transferee or successor, by Contract or other agreement or arrangement, by operation of Law or otherwise.
- (m) None of the Target Companies is, or has within the last seven (7) years, been, a partner in a partnership or joint venture, a party to a Tax partnership agreement or other Contract, agreement or arrangement that is treated, or required to be treated, as an arrangement requiring a partnership income Tax Return to be filed under Subchapter K of Chapter 1 of Subtitle A of the Code, or an owner of an interest in an entity treated as a partnership for Tax purposes.
- (n) None of the Target Companies has within the last seven (7) years participated in or been a party to, or has any Liability for the payment of any Tax resulting from a Person's participation, in any "listed transaction" within the meaning of Treasury Regulations Section 1.6011-4 (or any similar or corresponding provision of state, local or foreign Law).
- (o) The Target and each of the Target Subsidiaries has at all times within the last seven (7) years, including through the time immediately prior to the Closing, been properly treated and classified for U.S. federal income tax purposes (and, where applicable, state, local and foreign Tax purposes) as a disregarded entity, and no election has been or will be filed or made to change such treatment and classification for U.S. federal income tax purposes (or applicable state, local or foreign Tax purposes).
- (p) None of the Assets is (i) "tax-exempt use property" within the meaning of Section 168(h)(1) of the Code or Section 470(c)(2) of the Code, (ii) "tax-exempt bond financed property" within the meaning of Section 168(g)(5) of the Code, (iii) "limited use property" within the meaning of Revenue Procedure 2001-28, 2001-1 C.B. 1156, (iv) described in Section 168(g)(1)(A) of the Code, (v) subject to a lease described in Section 7701(h) of the Code or (vi) subject to any provision of state, local or foreign Law comparable to any of the foregoing.
- (q) None of the Target Companies is currently entitled to any material Tax incentive, deferral, holiday or abatement arrangement with any Governmental Authority or would be subject to any material recapture, clawback, termination or similar adverse consequence with respect to any Tax incentive, holiday, credits or other Tax reduction, deferral or abatement arrangement as a result of the transactions contemplated by this Agreement.

- (r) None of the Target Companies has (i) entered into any agreement or waiver extending any statute of limitations with respect to Taxes that remains in effect, (ii) agreed to any extension of time with respect to a Tax assessment or deficiency that remains in effect, (iii) entered into any closing agreement that remains in effect under applicable Tax Law, (iv) requested or is the subject of or expressly bound by any private letter ruling, technical advice memorandum or similar ruling or memorandum with any Governmental Authority that remains in effect with respect to any Taxes, nor is any such request outstanding, or (v) entered into any Contract with any Governmental Authority with respect to Taxes that (A) requires the Target Company to take any action or to refrain from taking any action or (B) would be terminated or adversely affected as a result of the transactions contemplated by this Agreement. No power of attorney granted by or with respect to any Target Company with respect to any Taxes is in effect that will not be revoked or cancelled at or prior to the Closing.
- 4.6 <u>Compliance with Laws</u>. Except as set forth in <u>Disclosure Schedule 4.6</u>: (a) each of the Target Companies is, and, for the last three (3) years has been, in compliance in all material respects with all applicable Laws and Regulations; and (b) no Seller Party has received written notice from any Governmental Authority that it is not in compliance with any applicable Law and, to the Knowledge of the Seller, there are no investigations or reviews pending or threatened by any Governmental Authority relating to any alleged violation of Law other than, in each case, those that have been resolved.
- 4.7 <u>Material Contracts</u>. <u>Disclosure Schedule 4.7(a)</u> contains a complete list, as of the Execution Date, of all the Material Contracts, and true and complete copies (including all material amendments thereto) of all such Material Contracts have been made available to the Buyer. Other than with respect to the matters set forth in <u>Disclosure Schedule 4.7(b)</u>, all such Material Contracts are in full force and effect on the Execution Date, and no Target Company is in material breach under any Material Contract nor has any Seller Party received, as of the Execution Date, any written notice of material breach or any event that with notice or lapse of time, or both, would constitute a material breach under a Material Contract by a Target Company if left unresolved, and, to the Knowledge of the Seller, no other Person that is party to a Material Contract is in material breach under any Material Contract. To the Knowledge of the Seller, no event has occurred that (with or without lapse of time) would reasonably be expected to result in a breach or violation of, or a default under, or give rise to the right of another party to terminate, the terms of any Material Contract. The Target Companies have performed all of their respective obligations, in all material respects, under the Material Contracts to which each of them is party.
- 4.8 <u>Intellectual Property</u>. Except as would not, individually or in the aggregate, reasonably be expected to have a Target Material Adverse Effect, the Target Companies have ownership of, or valid licenses to use, all trademarks, trade names, patents, service marks, brand names, computer programs, databases, industrial designs, copyrights, know-how, processes and other related intellectual property required to conduct their respective businesses as now being conducted ("*Intellectual Property*"). Except as set forth in <u>Disclosure Schedule 4.8</u>, the Target Companies' use of such Intellectual Property does not infringe on the rights of any Person, and neither the Seller nor any Target Company has received any written notice from any Person of any such alleged infringement by a Target Company. To the Knowledge of the Seller, no Person is infringing on the Target Companies' rights with respect to the Intellectual Property. Except as set forth in <u>Disclosure Schedule 4.8</u>, there are no royalty agreements between a member of the Target Companies, on the one hand, and any Third Person, on the other hand, relating to Intellectual Property.

4.9 <u>Broker's or Finder's Fees</u>. Except as set forth in <u>Disclosure Schedule 4.9</u>, no investment banker, broker, finder, agent or other Person is entitled to any brokerage or finder's fee or similar commission in respect of the transactions contemplated by this Agreement or any Transaction Document based in any way on agreements, arrangements, or understandings made by or on behalf of any Target Company.

4.10 Employee Benefits.

- (a) <u>Disclosure Schedule 4.10(a)</u> sets forth a list of all material Business Benefit Plans and designates whether such Business Benefit Plan constitutes a Target Company Benefit Plan or Seller Benefit Plan. With respect to each material Business Benefit Plan, the Seller has made available to the Buyer, where applicable, true and complete copies of: (i) the current plan document and all applicable amendments for each such plan or, in the case of an unwritten Business Benefit Plan, a written description of such plan; and (ii) all material, non-routine correspondence during the last three (3) years with all Governmental Authorities for which a Liability remains, in each case, the delivery of which to the Buyer would not violate applicable Law.
- (b) None of the Target Companies contributes to or is required to contribute to (or within the preceding six (6) years, has contributed to or been required to contribute to) a Multiemployer Plan. None of the Target Companies has any Liability (including on account of an ERISA Affiliate) with respect to any plan that is subject to Title IV or Section 302 of ERISA or Section 412 of the Code. No Business Benefit Plan is a multiple employer welfare arrangement (within the meaning of Section 3(40)(A) of ERISA). No Business Benefit Plan is funded through a trust that is intended to be exempt from federal income taxation pursuant to Section 501(c)(9) of the Code.
- (c) (i) Each Business Benefit Plan has been maintained, operated, funded and administered in compliance in all material respects with its terms and with all applicable Laws, including ERISA and the Code; (ii) each Business Benefit Plan intended to be qualified under Section 401(a) of the Code has received a favorable opinion, advisory or determination letter from the IRS, and, to the Knowledge of the Seller, nothing has occurred since the date of any such opinion or determination letter that would reasonably be expected to adversely affect such qualification; (iii) no Proceeding is pending or, to the Knowledge of the Seller, threatened in writing with respect to any Target Company Benefit Plan; and (iv) except as set forth in Disclosure Schedule 4.10(c), no material Target Company Benefit Plan has been established in the last twelve (12) months for which a liability remains outstanding.
- (d) No Business Benefit Plan provides medical benefits beyond termination of service or retirement to any Available Employee other than coverage mandated by Section 4980B(f) of the Code and Section 601-608 of ERISA (or similar state Law) or provided in connection with a Business Benefit Plan providing severance benefits.

- (e) Other than as set forth in <u>Disclosure Schedule 4.10(e)</u>, the execution and performance of this Agreement will not (i) constitute a stated triggering event under any Business Benefit Plan that will result in a material payment (whether of severance pay or otherwise) becoming due to any current or former officer, employee, manager, director or individual consultant (or dependents of such persons) of the Seller or any of the Target Companies or (ii) accelerate the time of a material payment or vesting of a material benefit or materially increase the amount of compensation due to any current or former officer, employee, manager, director or individual consultant (or dependents of such persons) of any of the Target Companies.
- (f) Except as set forth in <u>Disclosure Schedule 4.10(f)</u>, there is no Contract, arrangement or plan that would reasonably be expected to result, separately or in the aggregate, in the payment of any "excess parachute payment" within the meaning of Code Section 280G (or any corresponding provision of state, local or foreign Law) as a result of the transactions contemplated by this Agreement. For purposes of determining whether any such Contract, arrangement or plan would result in the payment of any "excess parachute payment," no payments or benefits payable pursuant to agreements entered into between the Buyer or any of its Affiliates (or at its or their direction) and any "disqualified individual" (within the meaning of Code Section 280G) shall be taken into account.
- (g) The Target has not issued, and is not required to issue, any Class B Units under any Target Company Benefit Plan or otherwise. Other than the Target Equity Awards, the Target Companies have not issued any other Equity Interest Equivalents under any Target Company Benefit Plan.
- 4.11 Employee Matters. Other than with respect to the matters set forth in Disclosure Schedule 4.11:
- (a) The Seller has made available to the Buyer a schedule of all Available Employees, setting forth for each individual the following information: (i) employer(s); (ii) work location; (iii) job title; (iv) hire date; and (v) leave status (the "Available Employee List").
- (b) As of the Execution Date, (i) none of the Available Employees are subject to, and no Target Company or other Seller Party (with respect to the Available Employees) is, or has been for the last three (3) years, a party to any collective bargaining agreement or other Contract with a union, works council or similar labor organizations, (ii) no petition for certification or union election is pending or, to the Knowledge of the Seller, threatened with respect to the Available Employees, and (iii) there are no strikes, slowdowns, work stoppages or other material labor disputes pending or, to the Knowledge of the Seller, threatened between any Target Company or other Seller Party (with respect to the Available Employees), on the one hand, and its respective employees, on the other hand.
- (c) Each of the Target Companies and other Seller Parties (with respect to the Available Employees) is in material compliance with all applicable Laws relating to employment and employment practices and terms and conditions of employment, including all Law relating to immigration, workplace safety and health, and wages, hours and overtime compensation (including, without limitation, the classification of employees as exempt or non-exempt under the Fair Labor Standards Act), labor relations, equal employment opportunities, fair employment practices, employment discrimination, harassment, retaliation, reasonable accommodation, disability rights or benefits, child labor, hiring, promotion and termination of employees, working

conditions, meal and break periods, privacy, health and safety, workers' compensation, leaves of absence, paid sick leave and unemployment insurance. All individuals characterized and treated by the Seller Parties as consultants or independent contractors of the Business are properly treated as independent contractors under all applicable Laws, except where such noncompliance would not result in any material liability to the Target Companies.

(d) There are no, and during the three (3) year period preceding the date hereof there have been no, material Proceedings pending or, to the Knowledge of the Seller, threatened, relating to any Available Employee or any Business Benefit Plan.

4.12 Financial Statements; Absence of Undisclosed Liabilities.

- (a) The Seller has delivered to the Buyer true and complete copies of (i) the consolidated audited balance sheet of the Target as of December 31, 2019 and 2018 (the "*Balance Sheets*") and related consolidated audited income statements, statements of cash flows, and statements of changes in members' equity of the Target for the years then ended (collectively, the "*Year End Financial Statements*") and (ii) the unaudited balance sheet of the Target as of June 30, 2020 (the "*Balance Sheet Date*") and related unaudited income statements, statements of cash flows, and statements of changes in members' equity of the Target for the six-month period ended June 30, 2020, together with the notes thereto (the "*Interim Financial Statements*" together with the Year End Financial Statements, the "*Financial Statements*") and such Financial Statements are set forth in Disclosure Schedule 4.12(a).
- (b) Each of the balance sheets included in the Financial Statements (including any related notes and schedules) fairly presents in all material respects the financial position of the Target Companies, as of the date thereof, and each of the income statements, statements of cash flows and statements of changes in members' equity included in the Financial Statements (including any related notes and schedules) fairly presents in all material respects the results of operations, cash flows and changes in members' equity, as the case may be, of the Target Companies for the periods set forth therein, in each case, in accordance with GAAP, subject, in the case of the Interim Financial Statements, to normal year-end adjustments and the absence of notes or other textual disclosures required under GAAP that are not, individually or in the aggregate, material.
- (c) There are no Liabilities of the Target Companies that are not reflected or reserved against in the Interim Financial Statements, other than, subject to the year-end adjustments provided for in Section 4.12(b), Liabilities not required to be presented in unaudited interim financial statements prepared in conformity with GAAP, and, since the Balance Sheet Date, the Target Companies have not incurred any Liability that would be required to be reflected or reserved against on a consolidated balance sheet of the Target Companies prepared in accordance with GAAP applied on a basis consistent with the Balance Sheets, in each case, other than Liabilities (i) not required to be presented in unaudited interim financial statements prepared in conformity with GAAP, (ii) that are current liabilities incurred in the Ordinary Course of Business of the Target Companies since the Balance Sheet Date, (iii) not in excess of \$100,000, (iv) arising after the date of this Agreement in connection with the transactions contemplated by this Agreement, (v) for expenses related to the transactions contemplated by the Transaction Documents or (vi) reflected in the calculation of Measurement Time Working Capital and included in the Post-Closing Statement.

- (d) Except as set forth in <u>Disclosure Schedule 4.12(d)</u>, none of the Target Companies has any outstanding Indebtedness, including letters of credit, guarantees, surety bonds and performance bonds, whether or not matured.
- (e) The Target Companies maintain a system of internal accounting controls adequate to provide reasonable assurance that transactions are recorded as necessary to permit preparation of the Financial Statements and to maintain accountability for the Assets.
- (f) The accounts receivable of the Target Companies represent valid obligations arising from the transactions actually made or services actually performed by the Target Companies in the Ordinary Course of Business.

4.13 Environmental Matters. Except as set forth in <u>Disclosure Schedule 4.13</u>:

- (a) each Target Company has been and is in compliance in all material respects with all Environmental Laws in the jurisdictions where each Target Company operates;
- (b) each Target Company has obtained and maintains in full force and effect, including timely filing of any renewals, and is in compliance in all material respects with all Permits, licenses and other authorizations required for the operation of each Target Company's business as currently operated under applicable Environmental Laws ("Environmental Permits"), and all such Environmental Permits are valid and in good standing; no Seller Party has received any written notice alleging any material violation under any Environmental Permit and, to the Knowledge of the Seller, there are no investigations or reviews pending or threatened by any Governmental Authority relating to any alleged violation of the terms of any Environmental Permit arising out of the Business other than, in each case, those that have been resolved. The consummation of the transactions contemplated by the Agreement and the Transaction Documents will not, to the Knowledge of the Seller, cancel, suspend, terminate or otherwise require modification of any Environmental Permit;
- (c) there has been no Release of any Hazardous Material at, on, under, to or from any of the property owned, operated or leased by any Target Company or for which any Target Company has assumed contractual liability, in each case in a manner that would reasonably be expected to give rise to any material Loss for any Target Company under any Environmental Law;
- (d) there are no Proceedings pending or, to the Knowledge of the Seller, threatened in writing against any Target Company alleging noncompliance with or liability under any Environmental Law;
- (e) no Target Company is subject to any Orders, Claims, Liens or Proceedings by any Governmental Authority or third party with respect to (i) any Environmental Law, including without limitation those related to allegations of economic loss, personal injury, illness or damage to real or personal property or the Environment, (ii) Remedial Actions or (iii) any Release or threatened Release of, or exposure to, a Hazardous Material ("Environmental Claims") and no such Environmental Claims are threatened in writing;

- (f) the Seller has made available to the Buyer all material environmental site assessments and reports prepared by third-party engineering, consulting or similar firms for or with respect to the operations or properties of the Target Companies that are in the possession or control of the Seller or the Target Companies; and
- (g) to the Knowledge of the Seller, none of any other properties used by the Target Companies in their Ordinary Course of Business or the properties to which Hazardous Materials generated or transported by the Target Companies have been transported or disposed, is (i) listed on the CERCLA National Priorities List, subject to CERCLA response, or any other similar list of sites maintained by any Governmental Authority requiring environmental investigation or response action or (ii) is the subject of any remediation, removal, cleanup, investigation, response action, Claim, Order, settlement, judgment or enforcement action regarding any contamination by or Release or threatened Release of Hazardous Materials, in each case that would reasonably be expected to result in material Liability to the Target Companies or materially interfere with the operations of the Target Companies.

4.14 Litigation.

- (a) Except as set forth in <u>Disclosure Schedule 4.14(a)</u>, (i) there are no Proceedings pending or, to the Knowledge of the Seller, threatened in writing against or involving any of (A) the Target Companies, the Business or the Assets, or (B) the Seller Parties with respect to the Target Companies, the Business or the Assets, or the Seller Parties' business activities on behalf of the Target Companies, and there have been no such Proceedings within the last two (2) years that resulted in a settlement or judgment, and (ii) there are no bankruptcy, insolvency, reorganization or arrangement proceedings pending, being contemplated by or, to the Knowledge of the Seller, threatened against the Target Companies.
- (b) Except as set forth in <u>Disclosure Schedule 4.14(b)</u>, no Target Company is subject to any outstanding Order (other than routine regulatory orders) or any executory compliance or settlement agreement, conciliation agreement, memorandum of understanding or letter of commitment with a Third Person.
- (c) Except as set forth in <u>Disclosure Schedule 4.14(c)</u>, (i) there are no material existing disputes with respect to the Target Companies, the Business or the Assets and (ii) no Seller Party has received any written notice alleging any material violation under any Permit held by the Target Companies and, to the Knowledge of the Seller, there are no investigations or reviews pending or threatened by any Governmental Authority relating to any alleged violation of the terms of any Permit arising out of the Business other than, in each case, those that have been resolved.

4.15 Property.

(a) <u>Disclosure Schedule 4.15(a)(i)</u> contains a true and complete list of all of the real property owned in fee by the Target Companies (the "Owned Real Property"). Except as set forth in <u>Disclosure Schedule 4.15(a)(i)</u>, each Target Company owns and has good and valid title to all of its Owned Real Property, free and clear of all Liens (except in all cases for Permitted Encumbrances). <u>Disclosure Schedule 4.15(a)(ii)</u> contains a true and complete list of all leases or subleases of real property by which a Target Company holds a leasehold interest (the "Leased Real").

Property"). Except as set forth in <u>Disclosure Schedule 4.15(a)(ii)</u>, each Target Company has a good and valid leasehold interest in all of its Leased Real Property, free and clear of all Liens (except in all cases for Permitted Encumbrances). Except as set forth in <u>Disclosure Schedule 4.15(a)(ii)</u>, all leases under which any Target Company leases any real property (the "**Real Property Leases**") are valid and effective against such Target Company and, to the Knowledge of the Seller, the counterparties thereto, in accordance with their respective terms and there is not, under any of the Real Property Leases, any existing material default (and no event has occurred which, with or without notice or lapse of time or both, would constitute a material default) or material breach by any Target Company and, to the Knowledge of the Seller, the counterparties thereto. The Seller has made available to the Buyer true and complete copies of the Real Property Leases.

- (b) The Target Companies have such consents, easements, rights-of-way, Permits or land-related licenses and surface use agreements, land use agreements and similar land-related agreements from each Person (collectively, "Rights-of-Way") as are sufficient to conduct their businesses in all material respects as currently conducted, except such Rights-of-Way that, if not obtained, would not materially interfere with the conduct of the Business as conducted as of the date hereof. Each of the Target Companies has fulfilled and performed all its obligations with respect to such Rights-of-Way which are required to be fulfilled or performed as of the date of this Agreement (subject to all applicable waivers, modifications, grace periods and extensions). There is not, under any of the Rights-of-Way, any existing default (and no event has occurred which, with or without notice or lapse of time or both, would constitute a default) or breach by any Target Company and, to the Knowledge of the Seller, the counterparties thereto. The Seller has made available to the Buyer true and complete copies of the Rights-of-Way.
- (c) There are no assessments against the Rights-of-Way held by the Target Companies for public improvements, and there is no pending and, to the Knowledge of the Seller, there is no threatened, condemnation of any Real Property Assets by any Governmental Authority that would materially interfere with the conduct of the Business as conduced as of the date hereof.
- (d) The Owned Real Property, the Leased Real Property and the Rights-of-Way (collectively, the "*Real Property Assets*") constitute all of the real property used for the conduct of the Business, in all material respects.
- (e) The Target Companies have good and valid title to all of the Personal Property Assets, in each case, free and clear of all Liens, except for Permitted Encumbrances. True and accurate descriptions of all material Personal Property Assets have been made available to the Buyer. The Personal Property Assets are, taken as a whole, (A) sufficient in all material respects for the ownership of the Business, normal wear and tear excepted, and (B) in a state of repair adequate in all material respects for normal operations in accordance with standard industry practice in the areas in which they are operated. Except as set forth in <u>Disclosure Schedule 4.15(e)</u>, all leases under which any Target Company leases personal property involving aggregate payments pursuant to such lease in excess of \$25,000 in any future calendar year (the "Personal Property Leases") are valid and effective against such Target Company and, to the Knowledge of the Seller, the counterparties thereto, in accordance with their respective terms and there is not, under any of the Personal Property Leases, any existing material default (and no event has occurred which, with or without notice or lapse of time or both, would constitute a material default) or material breach by any Target Company and, to the Knowledge of the Seller, the counterparties thereto. The Seller has made available to the Buyer true and complete copies of the Personal Property Leases.

- (f) Except as set forth in <u>Disclosure Schedule 4.15(f)</u>, the Assets constitute all the material assets necessary to conduct the Business as conducted in the Ordinary Course of Business. The Assets have been constructed, owned, maintained and, since the applicable in-service date in respect thereof, operated in a good and workmanlike manner and in all material respect in accordance with customary practices in the oil and gas and salt water disposal industries and all applicable Laws and are, in all material respects, in good repair, working order and operating condition and are adequate in all material respect for their present or intended uses by the Target Companies, ordinary wear and tear excepted.
- 4.16 <u>Affiliate Transactions</u>. Except as set forth in <u>Disclosure Schedule 4.16</u> and the Organizational Documents of the Target, there are no Contracts between (i) a Target Company or any of its directors, managers, officers, employees or consultants, or any members of their immediate families, on the one hand, and (ii) the Seller or its Affiliates (which, for the avoidance of doubt, does not include the Target Companies for purposes of this <u>Section 4.16</u>) or any of their respective directors, managers, officers, employees or consultants, or any members of their immediate families, on the other hand. None of Seller or its Affiliates (which, for the avoidance of doubt, does not include the Target Companies for purposes of this <u>Section 4.16</u>) or any of their respective directors, managers, officers, employees or consultants or any members of their immediate families owns any interest in any Assets.
- 4.17 <u>Absence of Certain Changes</u>. Except as set forth in <u>Disclosure Schedule 4.17</u>, since the Balance Sheet Date, except for this Agreement and the transactions contemplated by this Agreement and the other Transaction Documents, (x) the Target Companies have conducted the Business in the ordinary course in substantially the same manner in which it has been previously conducted, (y) there has not occurred any change, effect, event, circumstance, development, condition or occurrence that, individually or in the aggregate, has had or would reasonably be expected to have a Target Material Adverse Effect, and (z) without limiting the generality of <u>clause (x</u>), no Target Company has suffered any Casualty Loss in excess of \$100,000 or taken any action described in <u>Section 6.1(a)</u> that, if taken after the Execution Date and prior to the Closing Date without the prior written consent of the Buyer, would breach or violate such provision.
- 4.18 <u>Insurance</u>. <u>Disclosure Schedule 4.18</u> sets forth a true and complete list, including the name of the insurer, the risks insured and related limits, of the insurance policies maintained by the Seller or any of its Affiliates insuring the Assets or the Target Companies. All such policies are in full force and effect. All premiums payable under such policies (including with respect thereto covering all periods up to and including the Closing Date) have been paid in a timely manner and the Seller Parties have complied with the material terms and material conditions of all such policies. There is no material Claim outstanding under any such insurance policy and no event has occurred, and no circumstance or condition exists, that has given rise to or serves as the basis for or (with or without notice or lapse of time) could reasonably be expected to give rise to or serve as the basis for any such Claim under any such policy. None of the Seller Parties has received any written notice from any insurer or reinsurer of any reservation rights with respect to pending or paid claims.

- 4.19 <u>Bank Accounts</u>. <u>Disclosure Schedule 4.19</u> sets forth a true and complete list of (a) (i) all deposit, demand, time, savings, passbook, security or similar accounts that the Target Companies maintain with any bank or financial institution, (ii) the names and addresses of the financial institutions maintaining each such account, (iii) the purpose for which such account is established and (iv) the authorized signatories on each account and (b) the names of all Persons, if any, holding powers of attorney from a Target Company and a summary statement of the terms thereof.
- 4.20 <u>Books and Records</u>. All books and records of the Target Companies and the Business have been maintained in accordance with applicable Law in all material respects and in the Ordinary Course of Business of the Target Companies, and contain true and complete records of all meetings and other material actions taken by the Target Companies. Such books and records have been made available to the Buyer.
- 4.21 <u>Insolvency</u>. No Target Company (a) is insolvent, (b) is in receivership or dissolution, (c) has made any assignment for the benefit of creditors, (d) has admitted in writing its inability to pay its debts as they mature, (e) has been adjudicated bankrupt or (f) has filed a petition in voluntary bankruptcy, a petition or answer seeking reorganization, or an arrangement with creditors under the federal bankruptcy Laws or any other similar Laws, nor has any such petition been filed against such Target Company.
- 4.22 <u>Royalties</u>. All rentals, royalties, and other payments due in connection with ownership and operation of the Assets, including any payments due in connection with any division order or division of interest obligations, have been timely and properly paid, except those amounts, if any, in suspense.
- 4.23 Mineral Owner Accommodation Requests. No Seller Party has received any notices, requests, claims or demands (and to the Knowledge of the Seller, there are no threatened notices, requests, claims or demands) by the owners of the mineral interests underlying the Real Property Assets associated with requests for reasonable accommodations in the operation of the Assets to permit such owners to utilize their rights, as mineral owners, to explore for, drill, complete, recomplete or other operations located on or around the Real Property Assets, other than accommodation requests that would not materially interfere with or detract from the operation, value or use, in any material respect of the Assets by the Buyer.
- 4.24 <u>Throughput Data</u>. <u>Disclosure Schedule 4.24</u> sets forth summary delivery and throughput data and other operating data reflected therein with respect to each Designated Agreement (the "*System Data*") for the three (3)-month period ended August 18, 2020, which System Data is true and correct in all material respects.
- 4.25 <u>Hydrocarbon Imbalances</u>; <u>Future Delivery of Hydrocarbons</u>. Except as set forth in <u>Disclosure Schedule 4.25</u>, none of the Target Companies (a) has any imbalances of Hydrocarbons pertaining to the operation of the Target Company Systems or (b) is obligated, by virtue of any Hydrocarbon imbalance, prepayment arrangement under any Contract for the sale of Hydrocarbons, forward sale of production or any other obligation, to deliver Hydrocarbons at some future time without receiving full payment therefor, other than in a manner consistent with the normal cycle of billing.
- 4.26 <u>Regulatory Status</u>. Except as set forth in <u>Disclosure Schedule 4.26</u>, no portion of the Target Company Systems is subject to the jurisdiction of the Federal Energy Regulatory Commission under the Natural Gas Act of 1938, as amended, the Natural Gas Policy Act of 1978 or the Interstate Commerce Act. None of the Target Companies is subject to Regulation as a public utility company or a public service company or any similar designation(s) by any state public service commission.

- 4.27 <u>Target Company Business</u>. None of the Target Companies has engaged in any material respect in any business other than the Business, in each case, during the three (3) year period preceding the date hereof.
- 4.28 <u>FCC Licenses</u>. No Target Company nor other Seller Party (on behalf of any Target Company) holds any FCC Licenses, and no FCC Licenses are required for the operation of the Target Company Systems as conducted on the date hereof.

4.29 SWD Wells.

- (a) <u>Disclosure Schedule 4.29(a)</u>, sets forth, for each SWD Well, the maximum authorized discharge capacity stated in its respective Permit and the minimum capacity such SWD Well is capable of discharging its contents based on the current configurations of such SWD Well.
- (b) Except as set forth in <u>Disclosure Schedule 4.29(b)</u>, (i) the wellbore, casing of the well, tanks, pipelines, electrical equipment and pumps of each SWD Well are structurally sound, in good operating condition (subject to normal wear and tear), and are otherwise of the quality usable in the Ordinary Course of Business, (ii) to the Knowledge of the Seller, each SWD Well is, and during the time the applicable Target Company has owned the SWD Wells has been, used only for the disposal of materials characterized pursuant to applicable Environmental Law as non-hazardous oilfield waste or saltwater and (iii) no Target Company has received any unresolved written notice that any of the SWD Wells is required under applicable Law to be currently plugged and abandoned.
- 4.30 No Other Representations and Warranties. In making the decision to enter into this Agreement and consummate the transactions contemplated hereby, the Seller has relied solely upon the representations and warranties of the Buyer set forth in Article V. Except for the representations and warranties contained in Article V, none of the Buyer or any of its Representatives, nor any other Person has made or is making, and the Seller has not relied upon, any other representation or warranty of any kind or nature whatsoever, oral or written, express or implied, with respect to the Buyer, its Affiliates, this Agreement, the Transaction Documents or the transactions contemplated hereby or thereby. Except for the representations and warranties contained in Article V, the Seller disclaims, on its own behalf and on behalf of its Affiliates, any other representations or warranties of the Buyer, whether made by the Buyer or any of its Representatives or any other Person, with respect to the Buyer, this Agreement, the Transaction Documents or the transactions contemplated hereby or thereby or any reliance thereon.

ARTICLE V REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE BUYER

Subject to the limitation set forth in <u>Article X</u>, the Buyer represents and warrants to the Seller as of the Execution Date and the Closing Date as follows:

5.1 Organization, Good Standing, and Authority of the Buyer.

- (a) The Buyer is a limited liability company duly organized or formed, as applicable, validly existing, and in good standing under the Laws of its jurisdiction of organization or formation, with all requisite power and authority to own, lease, and operate its properties and to carry on its business as now being conducted.
- (b) The Buyer has the full right, power and authority to enter into and perform its obligations under this Agreement and the Transaction Documents to which it is or shall be a party and to purchase the Interests from the Seller at the Closing. The execution, delivery, and performance of this Agreement and the Escrow Agreement have been, and the other Transaction Documents to which the Buyer will be a party will be, prior to the Closing, duly authorized by all requisite limited liability company action on the part of the Buyer. This Agreement and the Escrow Agreement have been, and the other Transaction Documents to which the Buyer will be a party will be, prior to the Closing, duly executed and delivered by the Buyer, and (assuming due authorization, execution and delivery hereof by the other parties hereto and thereto) such Transaction Documents constitute (in the case of this Agreement and the Escrow Agreement) or will constitute at the Closing (in the case of such other Transaction Documents), legal, valid and binding obligations of the Buyer, enforceable against the Buyer in accordance with their terms, subject, as to enforceability, to bankruptcy, insolvency, reorganization, moratorium and other laws of general applicability relating to or affecting creditors' rights and to general principles of equity (regardless of whether such enforceability is considered in a Proceeding in equity or at Law).

5.2 No Conflicts.

- (a) Neither (i) the execution and delivery by the Buyer of this Agreement, any Transaction Documents to which the Buyer is a party, or any Transaction Document or other instrument required hereby or thereby to be executed and delivered by the Buyer at the Closing nor (ii) the performance by the Buyer of its obligations hereunder or thereunder will require any consent, approval, authorization or waiver of, notice to, or filing with, any Governmental Authority or any other Person.
- (b) The execution and delivery of this Agreement and the Escrow Agreement and, as of the Closing, any other Transaction Document to which the Buyer is a party by the Buyer, and the performance by the Buyer of its obligations hereunder or thereunder will not conflict with, violate, or breach the terms of, cause a default (with or without notice or lapse of time or both) under, or give rise to a right of termination, cancellation, or acceleration of any obligation or to the loss of a benefit under, or result in creation of any Lien on any of the Interests under: (i) the Organizational Documents of the Buyer, (ii) any material Contract or other instrument to which the Buyer is a party or by which it or any of its properties or assets are bound, or (iii) assuming the Authorizations or Notifications referred to in Section 4.3 are duly and timely obtained, any Law, Regulation or Order applicable to the Buyer, except, in the case of clauses (ii) and (iii), such matters as would not, individually or in the aggregate, reasonably be expected to prevent, delay, make illegal or otherwise interfere with the ability of the Buyer to consummate the transactions contemplated hereby.

- 5.3 <u>Litigation</u>. There is no Proceeding pending or, to the Knowledge of the Buyer, threatened against or affecting the Buyer or any of its Affiliates that, individually or in the aggregate, could reasonably be expected to prevent, delay, make illegal or otherwise interfere with the ability of the Buyer to consummate the transactions contemplated hereby.
- 5.4 <u>Bankruptcy</u>. There are no bankruptcy, insolvency, reorganization or receivership Proceedings pending against, being contemplated by, or, to the Knowledge of the Buyer, threatened against the Buyer.
- 5.5 <u>Broker's or Finder's Fees</u>. No investment banker, broker, finder, agent or other Person is entitled to any brokerage or finder's fee or similar commission in respect of the transactions contemplated by this Agreement or any Transaction Document based in any way on agreements, arrangements or understandings made by or on behalf of the Buyer or any of its respective Affiliates, in each case, that will be the obligation of the Seller or its Affiliates.

5.6 Investment Intent.

- (a) The Buyer is acquiring the Interests for its own account for investment and not with a view to, or for sale in connection with, any distribution thereof, nor with any present intention of distributing or selling the same, and the Buyer has no present or contemplated agreement, undertaking, arrangement, obligation, indebtedness or commitment providing for the disposition thereof.
- (b) The Buyer is an "accredited investor" within the meaning of Rule 501(a) under Regulation D of the Securities Act and has such knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of its investment in the Interests and the Target, and the Buyer is capable of bearing the economic risks of such investment and is able to bear the complete loss of its investment in the Interests and the Target.
- 5.7 <u>Sufficiency of Funds</u>. The Buyer will have at Closing, and at all times thereafter until the satisfaction of its obligations under this Agreement will have, sufficient cash, available lines of credit or other sources of immediately available funds on hand to enable the Buyer to consummate the transactions contemplated by this Agreement and the other Transaction Documents and to satisfy its obligations hereunder. The Buyer expressly acknowledges that the failure to have sufficient funds shall in no event be a condition to the performance of its obligations under this Agreement. No part of the funds paid by the Buyer pursuant to this Agreement has been, or is, directly or indirectly derived from, or related to, any activity that is reasonably expected to contravene any applicable Anti-Money Laundering Legislation.
- 5.8 <u>Solvency</u>. Assuming the representations and warranties of the Seller contained in this Agreement are true in all material respects, at and immediately after the Closing, and after giving effect to the transactions contemplated hereby, the Buyer will be Solvent.

5.9 <u>Independent Evaluation</u>. The Buyer is sophisticated in the evaluation, purchase, ownership, development, investment in and operation of oil and gas pipeline assets, including assets substantially similar to the Assets, and the Buyer is capable of evaluating the merits and risks of acquiring the Interests and bearing the economic risks of such investment. The Buyer has conducted its own independent investigation of the Target Companies, the Interests and the Assets, which investigation was done by the Buyer and its own advisors.

5.10 Business Investigation; Disclaimer Regarding Projections.

- (a) The Buyer has conducted such investigation of the business of the Target Companies as they have deemed necessary in order to make an informed decision concerning the transactions contemplated hereby. The Buyer acknowledges that it has had an opportunity to ask questions of the officers and management of the Seller and the Target Companies and has been given access to and have had the opportunity to visit and examine the Assets and the properties of the Target Companies. For the purpose of conducting these investigations, the Buyer has employed the services of its own Representatives and has reviewed such information, and obtained such advice, as the Buyer has deemed necessary or advisable to evaluate the Interests, the Target Companies and the transactions contemplated by this Agreement and the Transaction Documents. In all matters affecting the condition of the Assets, the properties of the Target Companies and the contents of the documents, records, reports or other materials in connection with the transactions contemplated hereby, the Buyer is relying upon the advice and opinion offered by its own Representatives. Notwithstanding any other provision of this Agreement to the contrary, except with respect to the representations and warranties of the Seller set forth in Article III and Article IV, neither the Seller, nor the Target Companies, nor any of their respective Representatives shall have or be subject to any liability to the Buyer or any other Person resulting from the distribution to the Buyer, or the Buyer's use of, any such information, including any information, documents or material provided or made available to the Buyer and its Representatives in any "data rooms" (virtual or otherwise), management presentations or in any other form in expectation of the transactions contemplated hereby.
- (b) In connection with the Buyer's investigation of the Business, the Buyer has received from the Seller, the Target Companies and their respective Representatives certain projections, forecasts and other forward looking information, including projected financial statements, cash flow items, business plans and other data related to the Target Companies, the Business and the prospects of the Target Companies. The Buyer acknowledges that (i) there are uncertainties inherent in attempting to make such projections, forecasts and plans and, accordingly, are not relying on them, (ii) the Buyer is taking full responsibility for making its own evaluation of the adequacy and accuracy of all projections, forecasts and plans so furnished to them, and (iii) except with respect to the representations and warranties of the Seller set forth in Article III and Article IV, the Buyer shall not have any claim against any Person with respect to any of the foregoing. Accordingly, notwithstanding any other provision of this Agreement to the contrary, the Buyer acknowledges that neither the Seller, nor the Target Companies, nor any of their respective Representatives has made any representation or warranty with respect to such projections, forecasts and other forward looking information.

5.11 Debt Financing.

- (a) The Buyer has delivered to the Seller a true, accurate and complete copy of the Debt Commitment Letter as of the date hereof.
- (b) The Buyer will have, upon receipt of, the funds in accordance with the Debt Commitment Letter and the Equity Commitment Letter, sufficient available funds on hand for it to consummate the transactions contemplated by this Agreement and perform its obligations under this Agreement and the Transaction Documents, including to pay any fees and expenses allocated to the Buyer pursuant to this Agreement to be paid on the Closing Date. The Buyer also has delivered to the Seller a complete and correct copy of the executed fee letter on the date hereof that relates to the Buyer's Debt Financing as redacted in a customary manner (i.e., redacted as to pricing, economic market flex and other provisions; provided, that the redacted pricing, market flex and other economic provisions set forth therein will not affect availability or conditionality of the Debt Financing at Closing) (as amended, modified, supplemented, replaced or extended from time to time after the date of this Agreement in accordance with Section 6.21(a), the "Fee Letter"). The Debt Commitment Letter has not been amended or modified prior to the date of this Agreement, and no such amendment or modification that could be reasonably be expected to impact the conditionality or aggregate amount of the Debt Financing is contemplated. Subject to the qualification that such enforceability may be limited by bankruptcy, insolvency, reorganization or other laws of general applicability relating to or affecting rights of creditors and that equitable remedies, including specified performance, are discretionary and may not be ordered, the Debt Commitment Letter is valid, binding and in full force and effect as of the date hereof, no event has occurred which, with or without notice, lapse of time or both, would either constitute or reasonably be expected to constitute a breach or default under the Debt Commitment Letter on the part of the Buyer or, to the Knowledge of the Buyer, the other parties thereto. As of the date hereof, the Buyer has paid (or caused to be paid) all fees or other amounts due and owing as of the date hereof in connection with the Debt Commitment Letter pursuant to the terms thereof. As of the date hereof, no termination of the Debt Commitment Letter is contemplated by the Buyer or, to the Knowledge of the Buyer, any other party thereto. The Buyer has not entered into any side letters or other contracts, understandings, agreements or arrangements related to the conditionality and availability of the funding under the Debt Financing, other than as expressly set forth in the Debt Commitment Letter delivered to the Seller.

5.12 Equity Financing.

- (a) The Buyer has delivered to the Seller a true, accurate and complete copy of the Equity Commitment Letter as of the date hereof.
- (b) The Buyer will have, upon receipt of, the funds in accordance with the Debt Commitment Letter and the Equity Commitment Letter, sufficient available funds on hand for it to consummate the transactions contemplated by this Agreement and perform its obligations under this Agreement and the Transaction Documents, including to pay any fees and expenses allocated to the Buyer pursuant to this Agreement to be paid on the Closing Date. The Equity Commitment Letter has not been amended or modified prior to the date of this Agreement. Subject to the qualification that such enforceability may be limited by bankruptcy, insolvency, reorganization or other laws of general applicability relating to or affecting rights of creditors and that equitable remedies, including specified performance, are discretionary and may not be ordered, the Equity Commitment Letter is valid, binding and in full force and effect as of the date hereof, no event has occurred which, with or without notice, lapse of time or both, would either constitute or reasonably be expected to constitute a breach or default under the Equity Commitment Letter on the part of the Buyer or, to the Knowledge of the Buyer, the other

parties thereto. As of the date hereof, no termination of any of the Equity Commitment Letter is contemplated by the Buyer or, to the Knowledge of the Buyer, any other party thereto. The Buyer has not entered into any side letters or other contracts, understandings, agreements or arrangements related to the conditionality and availability of the funding under the Equity Financing, other than as expressly set forth in the Equity Commitment Letter delivered to the Seller.

5.13 No Other Representations and Warranties. In making the decision to enter into this Agreement and consummate the transactions contemplated hereby, the Buyer has relied solely upon the representations and warranties of the Seller set forth in Article III and Article IV. Neither the Seller, nor the Target Companies, nor any of their respective Representatives, nor any other Person has made or is making any representations or warranties to the Buyer or any other Person regarding the probable success or profitability of the Target Companies or the Interests (whether before or after the Closing).

ARTICLE VI COVENANTS

6.1 Conduct of Business.

- (a) Except (i) as expressly required by this Agreement, (ii) as set forth in <u>Disclosure Schedule 6.1(a)</u>, (iii) as required by applicable Law, (iv) as provided for in the Organizational Documents of the Target Companies as in effect on the date of this Agreement, or (v) as consented to by the Buyer (which consent shall not be unreasonably withheld, delayed or conditioned), during the period from the Execution Date until the Closing, the Seller shall, and shall cause each of the Target Companies to (A) conduct its business in the Ordinary Course of Business; (B) use commercially reasonably efforts to preserve intact the Business and such Target Company's relationship with employees, material customers, material suppliers and creditors; (C) use commercially reasonable efforts consistent with past practice to maintain and to keep its tangible assets in good repair and condition, ordinary wear and tear excepted and (D) use commercially reasonable efforts to keep in full force and effect insurance applicable to its assets and operations comparable in amount and scope of coverage to that is currently maintained. Without limiting the generality of the foregoing, except (A) as expressly required by this Agreement, (B) as set forth in <u>Disclosure Schedule 6.1(a)</u>, (C) as required by applicable Law, or (D) as consented to in writing by the Buyer, during the period from the Execution Date to the Closing Date, the Seller shall not, and shall cause each of the Target Companies not to, and with respect to subclause (xi) below, the Seller shall not (and shall cause the other Seller Parties) not to:
 - (i) (A) issue, sell, grant, dispose of, accelerate the vesting of or modify, as applicable, any of its Equity Interests, (B) redeem, purchase or otherwise acquire any of its Equity Interests, (C) split, combine or reclassify any of the Equity Interests or (D) declare, set aside or pay any dividend or other distribution, other than wholly in cash, in respect of the Interests;
 - (ii) incur, refinance, assume or guarantee any Indebtedness, or issue or sell any debt securities or options, warrants, calls or other rights to acquire any debt securities of any Target Company, other than (A) borrowings under the Existing Credit Facility, or any replacements thereof; *provided* that the aggregate Indebtedness of the Target Companies does not exceed \$80,000,000.00 in the aggregate, (B) borrowings from any Target Company by any other Target Company and (C) repayments of borrowings from any Target Company by any other Target Company;

- (iii) sell, transfer, lease or otherwise dispose of any of its Assets (including any inventory reclassified from products, plant and equipment to current assets in the Interim Financial Statements) with a fair market value in excess of \$100,000, individually or \$250,000 in the aggregate, except, (A) pursuant to Contracts in force as of the Execution Date and listed in <u>Disclosure Schedule 6.1(a)</u>, (B) dispositions of obsolete or worthless equipment, (C) dispositions of crude oil, natural gas, condensate, natural gas liquids and other produced Hydrocarbons and minerals in the Ordinary Course of Business or (D) sales, transfers, leases, or other disposals to any Target Company;
- (iv) make any capital expenditure or capital expenditures, except for (A) any capital expenditures set forth on the CapEx Budget, (B) unbudgeted emergency operations, repairs, replacements or expenditures and (C) any capital expenditures that, in the aggregate, do not exceed 110% of the total amount of capital expenditures set forth on the CapEx Budget;
- (v) directly or indirectly acquire by merging or consolidating with, or by purchasing any Equity Interest in, or by any other manner, any Person or division, business or Equity Interest of any Person;
- (vi) (A) enter into any Contract or agreement that would be a Material Contract if in existence as of the Execution Date or (B) amend or terminate any Material Contract (unless due to a material breach by a counterparty or expiration in accordance with the terms of such Material Contract); *provided*, *however*, that this Section 6.1(a)(vi) shall not prohibit entering into any Material Sales Contract or Material Gathering Contract that is cancellable without payment or penalty upon notice of 30 days or less; *provided*, *further*, that with respect to any Material Sales Contract with a term of four (4) days or longer, the Seller shall first reasonably consult with the Buyer;
- (vii) in each case, solely to the extent related to any Target Company or the Assets and, in the case of any consolidated or combined return of which any Target Company is not the parent, solely to the extent any such action would have the effect of increasing any Tax of any Target Company to with respect to the Assets following the Closing, (A) change its Tax accounting period, any material method of Tax accounting or any methods of reporting income or deduction for Tax or accounting practice or policy, (B) make, change, or revoke any material Tax election other than in the Ordinary Course of Business, (C) prepare any material Tax Return in a manner which is not consistent with the past practice of such Target Company with respect to the treatment of items on such Tax Returns, (D) file any amendment to a Tax Return, (E) incur any material Liability for Taxes other than in the Ordinary Course of Business, (F) enter into any closing agreement with respect to Taxes or settle or compromise any Claim or audit, examination or other Proceeding with respect to Taxes, (G) consent or agree to any extension or waiver of the limitation period applicable to any Claim or assessment in respect of Taxes with any Governmental Authority or (H) surrender any right to claim a refund of material Taxes;

- (viii) amend a Target Company's Organizational Documents;
- (ix) adopt a plan or agreement of complete or partial liquidation, dissolution, restructuring, recapitalization, merger, consolidation or other reorganization (other than transactions exclusively between Target Companies);
 - (x) change in any material respect its accounting methods, practices or principles except as required by GAAP or applicable Law;
- (xi) except as allowed under the terms of any Business Benefit Plan, (A) employ any additional Persons as common law employees in any Available Employee capacity, except as reasonably necessary to replace any open Available Employee position below executive officer in the Ordinary Course of Business, (B) grant or promise any increase in salary, wages, benefits, severance, bonuses or other compensation payable or to become payable to any Available Employee, (C) engage any additional individual independent contractors, consultants or agents pursuant to any Contract for which the Target Companies will have any continuing obligation after the Closing, (D) amend (other than to the minimum extent required to conform with applicable Law), terminate, accelerate rights under or become obligated to make payments under or with respect to (1) any Business Benefit Plan (or any arrangement that would constitute a Business Benefit Plan, if adopted) or (2) any loan with any of its officers, directors or managers, or (E) enter into, establish or adopt any additional Business Benefit Plan that is not otherwise in effect as of the date hereof;
- (xii) enter into or assume, or in the case of the Seller, cause or permit any Target Company to enter into or assume, any Contract for employment or similar agreement or arrangement with or to employ or receive services from any Person;
- (xiii) waive any claims or rights under any Contracts or otherwise pertaining to the Business, other than claims or rights that are immaterial in amount and consequence to the Business;
- (xiv) delay or postpone any payment of accounts payable or other payables or expenses, or accelerate the collection of accounts receivable or cash collections of any type;
 - (xv) form any Subsidiary of a Target Company;
- (xvi) mortgage, pledge or subject any Lien (other than a Permitted Encumbrance) any of the Assets or any Equity Interests in a Target Company;
- (xvii) institute, settle or compromise any pending or threatened Claim or Proceeding, other than Claims not in excess of \$25,000 individually or \$100,000 in the aggregate; or
 - (xviii) agree, in writing or otherwise, to take any of the foregoing actions.
- (b) In the event of an emergency, the Seller or the applicable Target Company may take such commercially reasonable action as a prudent operator would take, and any such actions shall not be deemed to be a breach of the provisions of <u>Section 6.1(a)</u>, and shall notify the Buyer of such action promptly thereafter.

6.2 Access to Information; Confidentiality.

- (a) From the Execution Date through the Closing Date, upon reasonable advance written notice from the Buyer to the Seller and subject to applicable Laws relating to the exchange of information, the Seller shall, and shall cause each of the Target Companies to (i) afford to the Buyer and its Representatives reasonable access during the Target Companies' normal business hours (and, with respect to books and records, the right to copy) to all of the Target Companies' offices, properties, books and records, Contracts, records and correspondence (in each case, whether in physical or electronic form), officers, employees, accounting firms, counsel, financial advisors and other Representatives, (ii) furnish the Buyer and its Representatives with such additional information concerning the Target Companies' businesses, properties and personnel as may be requested, and (iii) allow the Buyer and its Representatives to conduct site visits of the Real Property Assets during normal businesses hours; provided, however, that the Seller shall have the right to have a Representative of the Seller of its Affiliates present for any communication with the Target Companies' Representatives or any site visits. None of the Buyer's access hereunder shall include any operation of equipment or any Phase II environmental site assessment or invasive or subsurface investigation, testing, analysis or sampling, including any of the environmental media or property, in each case, without the prior written consent of the Seller.
- (b) This Section 6.2 shall not require the Seller to permit any access, or to disclose any information, (i) which the Seller reasonably believes it or the Target Companies are prohibited from providing to the Buyer by reason of applicable Law, which constitutes or allows access to information protected by attorney-client privilege, or which the Seller or the Target Companies are required to keep confidential or prevent access to by reason of any Contract with or duty to any Third Person, (ii) relating to pricing or other matters that are highly sensitive if the exchange of such documents (or portions thereof) or information, as determined by Seller Counsel, would reasonably be expected to result in regulatory concerns for the Seller or any of its Affiliates, (iii) relating to any potential sale of any of the Target Companies or the Business to any other Person, or (iv) if the Seller or its Affiliates, on the one hand, and the Buyer or any of its Subsidiaries, on the other hand, are adverse parties in a litigation, such information being reasonably pertinent thereto; *provided*, *however*, that the Seller shall make appropriate substitute disclosure arrangements, if available, under circumstances in which the restrictions of the foregoing provisions apply (other than with respect to the restrictions in clause (iii) above).
- (c) Any inspection or investigation conducted by the Buyer or its Representatives prior to the Closing will be conducted in accordance with applicable Laws, including any applicable Environmental Laws, and in such manner as not to interfere unreasonably with the business or operations of the Seller or the Target Companies. The Seller and its Affiliates make no representation or warranty as to the accuracy of any information (if any) provided pursuant to this Section 6.2, and the Buyer may not rely on the accuracy of any such information, in each case other than as expressly set forth in the representations and warranties contained in Article III, Article IV and the Transaction Documents.

(d) If the Buyer exercises its rights of access under this Section 6.2, or conducts examinations or inspections under this Section 6.2 or otherwise, then (i) such access, examination and inspection will be at the Buyer's sole risk, cost and expense and the Buyer waives and releases, on behalf of itself and each Person undertaking any such examination or inspection on the Buyer's behalf, all Losses and other Claims against the Seller, the Target Companies, their Affiliates and their respective Representatives (collectively the "Inspection Indemnitees") arising in any way therefrom or in any way related thereto and (ii) except to the extent of an Inspection Indemnitee's gross negligence or willful misconduct, the Buyer will indemnify, defend and hold harmless the Inspection Indemnitees from and against any and all Losses and other Claims of any kind or character attributable to any personal injury, death or physical or other property damage, or violation of the Seller's, the Target Companies' or any third-party operator's rules, regulations or operating policies of which the Buyer was previously informed in writing, arising from such examinations or inspections; provided, however, the Parties agree that in no event shall any Inspection Indemnitee be entitled to indemnification by the Buyer for any Losses to the extent arising out of any preexisting environmental contamination or noncompliance with Environmental Law, except to the extent such contamination or noncompliance is exacerbated by Buyer. THE FOREGOING RELEASE AND INDEMNIFICATION WILL APPLY WHETHER OR NOT SUCH DAMAGES OR OTHER CLAIMS ARISE OUT OF (A) NEGLIGENCE (INCLUDING SOLE NEGLIGENCE, SIMPLE NEGLIGENCE, CONCURRENT NEGLIGENCE, ACTIVE OR PASSIVE NEGLIGENCE, BUT EXCLUDING GROSS NEGLIGENCE OR WILLFUL MISCONDUCT) OF THE SELLER OR ITS AFFILIATES OR (B) STRICT LIABILITY. Notwithstanding any provision to the contrary contained in this Agreement, the provisions of this Section 6.2(d) will survive the termination of this Agreement pursuant to Article IX and the Closing.

(e) Except for disclosures permitted by the terms of the Confidentiality Agreement, the Buyer and its Representatives will hold all information received from the Seller pursuant to this Section 6.2 in confidence in accordance with the terms of the Confidentiality Agreement.

6.3 <u>Contact with Customers, Suppliers and Other Business Relations</u>. Prior to the Closing, the Buyer and its Representatives shall not (i) contact or communicate with the employees (except the executive officers and other employees (A) as authorized by the executive officers, (B) as requested by the Buyer, to the extent such requests are commercially reasonable and would not interfere with the operations of the Business, (C) as or permitted pursuant to this Agreement) or the Third Person customers, suppliers and other Third Person business relations of the Target Companies (except for the entities set forth on <u>Schedule 6.3</u>) in connection with, or relating in any way to, the transactions contemplated hereby, without the prior written consent of the Seller (which consent may be conditioned, including the conditions set forth on <u>Schedule 6.3</u> with respect to the applicable entities set forth thereon), (ii) persuade or seek to persuade any actual or prospective Third Person customer or supplier of the Seller or the Target Companies to cease to do business or to reduce the amount of business which that Person has customarily done or contemplates to continue doing with the Seller or the Target Companies, as applicable, or (iii) interfere or attempt to interfere with the business relationships (whether formed prior to or after the Execution Date) between the Seller or the Target Companies and any Third Person customer, supplier or independent contractor of the Seller or the Target Companies. For the avoidance of doubt, nothing contained herein shall prevent the Buyer and its Representatives from contacting or communicating with any Person in the Ordinary Course of Business of the Buyer and its Affiliates for matters unrelated to the transactions contemplated by this Agreement or from conducting general market diligence, in each case, subject to the terms and conditions of any applicable confidentiality agreement (including the Confidentiality Agreement).

6.4 <u>Non-Solicitation</u>. From the day after the Closing Date through the date that is twelve (12) months after the Closing Date, a Buyer Party may solicit and hire any employee of a Seller Party so long as such Buyer Party pays the Seller a sum equal to the actual severance paid or to be paid by the Seller in connection with the Seller's termination of such person's employment relationship with any Seller Party within two (2) days of the date upon which such individual commences employment with the Buyer or any Affiliate of the Buyer.

6.5 Insurance. From and after the Closing Date, the Target Companies and the Assets shall cease to be insured by the insurance policies held by the Seller or any of its Affiliates (excluding, after the Closing, the Target Companies), or by any of their self-insured programs, and neither the Buyer nor its Affiliates (including, after the Closing, the Target Companies) shall have any access, right, title or interest to or in any such insurance policies (including to all claims and rights to make claims and all rights to proceeds) to cover the Assets or any Liability arising from the business or operations of the Target Companies; provided that, with respect to events or circumstances relating to the Assets or Target Companies that occurred or existed prior to the Closing Date that are covered by any occurrence-based insurance policies held by the Seller or any of its Affiliates with any third-party insurers (the "Seller Insurance"), the Buyer or the Target Companies may, after the Closing Date and subject to prior consultation with the Seller, make claims under the Seller Insurance to the extent such coverage and limits are available to the Target Companies under the Seller Insurance. The Seller shall have the right (but not the obligation) to monitor or associate with such claims. By making any claims under the Seller Insurance, the Buyer agrees to promptly (and in any event within ten (10) Business Days) reimburse the Seller Parties for all documented costs and expenses (including any increased premiums, claim handling fees, allocated claims expenses, losses, liabilities or fees) incurred by any Seller Party as a result of such claims, and the Buyer shall exclusively bear the amount of any "deductible" or net retention associated with any claims under the Seller Insurance. In order to support this Section 6.5, the Seller shall, at the Buyer's sole expense, cooperate with the Buyer to make claims and amend the Employment Practices Liability and Directors & Officer's Liability insurance policies specifically related to extended reporting or run off endorsements and policy changes triggered by the Closing. The applicable Seller Party shall retain all rights to control its insurance policies or programs, including the right to exhaust, settle, release, commute, buy back or otherwise resolve disputes with respect to any of its insurance policies and programs. The Seller and its Affiliates may, to be effective as of the Closing, amend any insurance policies in the manner it deems appropriate to give effect to this Section 6.5; provided that no such amendment or modification shall materially impair the Buyer's or the Target Companies' rights under this Section 6.5 with respect to the Seller Insurance. From and after the Closing, the Buyer shall be responsible for securing all insurance it considers appropriate for its operation of the Assets.

6.6 Efforts. Subject to the terms and conditions of this Agreement, each of the Seller and the Buyer shall cooperate with the other and use (and shall use their commercially reasonable efforts to cause the Target Companies and the Buyer's Subsidiaries, respectively, to use) its commercially reasonable efforts to (a) take, or cause to be taken, all actions, and do, or cause to be done, all things, necessary, proper or advisable to cause the conditions to the Closing to be satisfied as promptly as practicable (and in any event no later than the Outside Date) and to consummate and make effective, in the most expeditious manner practicable, the transactions contemplated hereby, including preparing and filing promptly and fully all documentation to effect all necessary filings, Notifications, notices, petitions,

statements, waivers, submissions of information, applications and other documents, (b) obtain promptly (and in any event no later than the Outside Date) and maintain all approvals, consents, clearances, expirations or terminations of waiting periods, registrations, Permits, Authorizations and other confirmations from any Governmental Authority or Third Person necessary, proper or advisable to consummate the transactions contemplated hereby, (c) defend any Proceedings challenging this Agreement or the consummation of the transactions contemplated hereby or seek to have lifted or rescinded any injunction or restraining order or other Order adversely affecting the ability of the Parties to consummate the transactions contemplated hereby; *provided, however*, that the Parties acknowledge and agree that the obligations of the Parties with respect thereto shall not extend beyond the Outside Date, and (d) obtain all necessary consents, approvals or waivers from third parties.

6.7 Tax Matters.

(a) Tax Returns.

- (i) The Seller shall, at the sole expense of the Seller, cause to be prepared and, unless otherwise required by applicable Law, timely file all Tax Returns required to be filed by or with respect to any of the Target Companies for all Tax periods ending on or before the Closing Date (each, a "*Pre-Closing Tax Period*") that are due on or after the Closing Date, including, for the avoidance of doubt, all Tax Returns of any combined group of which each of (i) a Target Company and (ii) Seller or an Affiliate of Seller (other than a Target Company) is or was a member on or prior to the Closing Date for Texas franchise Tax purposes (including in each such Tax Return any and all items and activities of the Target Companies through and including the Closing Date), and Seller (or such Affiliate of Seller treated as the reporting entity for Texas franchise Tax purposes) shall pay all Texas franchise Taxes owed with respect to such Tax Returns; *provided*, *however*, that Buyer shall take no action on the Closing Date after the Closing that would have the effect of increasing the amount of Texas gross receipts incurred by any Target Company on the Closing Date.
- (ii) The Buyer shall timely file or cause to be timely filed when due (taking into account all extensions properly obtained) all Tax Returns of the Target Companies for all Straddle Periods.
- (iii) All Tax Returns described in Sections 6.7(a)(i) and 6.7(a)(ii) shall be prepared on a basis consistent with past practice of the applicable Target Company, and without a change of any election or any accounting method. Not later than thirty (30) days prior to the due date for filing any such Tax Return, the Seller or Buyer (as applicable) shall deliver a draft of such Tax Return (together with schedules, statements and supporting documentation) to the Buyer or Seller (as applicable) for its review and reasonable comment. The filing Party shall cause such Tax Return (as revised to incorporate the other Party's reasonable comments made in writing to the filing Party before the due date for such Tax Return) to be timely filed.
- (iv) Except as otherwise required by Law, without the prior written consent of the Seller (which consent shall not be unreasonably withheld, conditioned or delayed), neither the Buyer nor any Target Company, nor any Affiliate thereof, shall, with respect to any Target Company for any Pre-Closing Tax Period or any Straddle Period: (A) make any election under the Code or any other Tax Law (other than an election required to be made under Sections 6226(a) or

754 of the Code pursuant to Section 6.7(a)(vi) or otherwise expressly required pursuant to this Agreement); or (B) file any amended Tax Return, initiate any voluntary disclosure agreement or program, or similar disclosure process, or agree to any adjustment of any Tax item with a Governmental Authority that relates to a Pre-Closing Tax Period Tax Return or Straddle Period Tax Return of a Target Company, if in any such case, such action would have the effect of increasing the Seller's or any of its direct or indirect owners' liability for any Taxes. For the avoidance of doubt, the parties agree that, in the absence of a final determination to the contrary, the Synthetic Equity Interest Agreement shall not be treated as having caused any Target Company to constitute a partnership for any Tax purpose or caused any Target Company to have been party to any Tax partnership or similar arrangement; *provided*, however, that in the event there is a material change under the Synthetic Equity Interest Agreement to the economic entitlements of the parties thereto after the Closing that causes the Buyer to conclude that it is required by Law to treat the Synthetic Equity Interest Agreement as having caused a Target Company to constitute a partnership or to be party to a Tax partnership or similar arrangement for a Straddle Period or a Tax period following the Closing, the Buyer shall be entitled to act consistently with such determination; *provided further*, however, that the Buyer shall consult with the Seller regarding such determination and take the Seller's input into account in good faith before taking any action consistent with such determination.

(v) If any Target Company receives a Tax refund or applies a credit against Taxes which refund or credit arises from or is attributable to the Sales Tax Contest, such refund or the amount of such credit (less any reasonable costs incurred by the Buyer or any Target Company to obtain such refund or credit) shall be paid to the Seller; *provided*, that the Buyer or the applicable Target Company shall be entitled to either withhold payment of such refund to the Seller until the Seller obtains a written release of the Buyer from any obligations to Ernst & Young with respect to the contingent fee engagement related to the Sales Tax Contest or, without duplication, pay such contingent fee out of the proceeds of the Sales Tax Contest; and *provided*, further, that the Buyer shall be entitled to offset against the amount of such refund any Liabilities for Taxes related to the Sales Tax Contest that are imposed on any Target Company or for which Buyer or any Target Company are held liable (directly or indirectly). For the avoidance of doubt, notwithstanding anything in this Agreement to the contrary, the Seller shall have the sole right to control the Sales Tax Contest and shall keep the Buyer reasonably informed with respect to same.

(vi) To the extent any Target Company is, pursuant to either (1) a final determination or (2) a preliminary finding by a Governmental Authority that requires a Target Company to appoint a "partnership representative" under Section 6223 of the Code, determined to have constituted a partnership for Tax purposes or determined to have been party to any Tax partnership or similar arrangement, in each case, during any Pre-Closing Tax Period, with respect to any adjustment of any Tax item, or any audit, examination or other Proceeding related to any Target Company (or such Tax partnership or similar arrangement), the Seller shall, to the extent permitted by Law, take, and cause its Affiliates to take, such actions as are needed to cause the applicable Target Company (or such Tax partnership or similar arrangement) to (A) cause the Buyer to be designated as the "partnership representative" of the applicable Target Company (or such Tax partnership or similar arrangement), and to cause Buyer's appointee to be designated as the "designated individual" to act on behalf of Buyer in its capacity as partnership representative of the applicable Target Company (or such Tax partnership or similar arrangement), in each case,

by executing IRS Form 8979 or such successor or similar form or authorization prescribed for such purpose (and any corresponding or similar form required to approve such similar designations under applicable state or local Tax Law), (B) make a timely "push out" election under Section 6226(a) of the Code (and any corresponding or similar provision of state or local Tax Law) for such Pre-Closing Tax Period, and (C) file an actual or protective election (as the case may be) under Section 754 of the Code with the next Tax Return eligible to be filed. For the avoidance of doubt, nothing about this Section 6.7(a)(vi) shall be interpreted as modifying any other requirement or obligation under this agreement, including, without limitation, the obligations in Section 6.7(a)(iv).

- (b) <u>Cooperation</u>. In connection with the preparation of Tax Returns, audit examinations and any administrative or judicial Proceedings relating to the Tax Liabilities imposed on or with respect to any Target Company, the Parties shall cooperate fully with each other, including the furnishing or making available during normal business hours of records, information, personnel (as reasonably required), books of account, or other materials reasonably relevant or helpful for the preparation of such Tax Returns, the conduct of audit examinations or any administrative or judicial Proceedings or the defense of Claims by Governmental Authorities as to the imposition of Taxes. The Parties agree to: (i) retain all books and records with respect to Tax matters pertinent to the Target Companies relating to any Pre-Closing Tax Period or any Straddle Period until the earlier of (A) sixty (60) days after the expiration of the applicable statute of limitations and any extension thereof for the respective taxable periods and (B) the wind-down, dissolution or liquidation of either Party, and to abide by all record retention agreements entered into with any Governmental Authority; and (ii) give the other Party reasonable written notice prior to transferring, destroying or discarding any such books and records (including in the event of the wind-down, dissolution or liquidation of a Party) and, if such other Party so requests, allow such other Party to take possession of such books and records.
- (c) <u>Transfer Taxes</u>. Neither Buyer nor Seller anticipates that the purchase and sale of the Interests described hereunder will result in any Liabilities for Transfer Taxes. Notwithstanding the immediately preceding sentence or anything in this Agreement to the contrary, all Transfer Taxes shall be borne by the Buyer. The Buyer and the Seller shall, upon written request from the Buyer, use their commercially reasonable efforts to mitigate, reduce or eliminate any such Transfer Tax that could be imposed.
- (d) <u>Tax Sharing Agreements</u>. Seller shall cause any and all Tax allocation, sharing and indemnity Contracts and any other similar Contract or other agreement or arrangement relating to Taxes (whether written or unwritten) solely between any of the Target Companies, on the one hand, and the Seller or any of its Affiliates (other than the Target Companies), on the other hand, to be terminated at or prior to the Closing. After such termination, none of the Target Companies shall have any further rights or liabilities thereunder. At or prior to Closing, Seller and each of its Affiliates that are parties to the LINN Tax Matters Agreement shall enter into a waiver and release agreement with respect to the LINN Tax Matters Agreement (the "LINN Tax Matters Termination Agreement") pursuant to which Seller and each such Seller Affiliate will waive all rights against the Target Companies which Seller or such Seller Affiliate may have under the LINN Tax Matters Agreement.

6.8 Continuation of Indemnity; D&O Tail Coverage.

- (a) The Buyer agrees that all rights to indemnification, advancement of expenses and exculpation by a Target Company now existing in favor of each Person who is now, has been at any time before the Execution Date, or who becomes prior to the Closing Date a manager, managing member, director or officer of a Target Company or the Seller as provided in the respective Organizational Documents of the Target Companies, in each case as in effect on the Execution Date, shall survive the Closing Date and shall continue in full force and effect in accordance with their respective terms for six (6) years or such shorter time provided for in such terms.
- (b) The Target Companies shall, and the Buyer shall cause the Target Companies to, obtain as of the Closing Date "tail" insurance policies with a claims period of six (6) years from the Closing Date with at least the same coverage and amounts, and containing terms and conditions that are not less advantageous to the managers, managing members, directors and officers of the Seller and the Target Companies as the "directors' and officers" liability insurance maintained by the Target Companies or any Person on behalf of the Target Companies immediately prior to the Closing Date with respect to claims arising out of or relating to events which occurred on or prior to the Closing Date (including in connection with the transactions contemplated by this Agreement) (collectively, "D&O Insurance").
- (c) The obligations of the Buyer and the Target Companies under this <u>Section 6.8</u> shall not be terminated or modified in such a manner as to adversely affect any manager, managing member, director or officer of the Seller or the Target Companies, as applicable, to whom this <u>Section 6.8</u> applies without the consent of such affected Person (it being expressly agreed that such managers, managing members, directors and officers to whom this <u>Section 6.8</u> applies shall be third-party beneficiaries of this <u>Section 6.8</u>, each of whom may enforce the provisions of this <u>Section 6.8</u>).
- (d) In the event the Buyer, the Target Companies or any of their respective successors or assigns (i) consolidates with or merges into any other Person and is not the continuing or surviving entity in such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any Person, then, and in either such case, proper provision shall be made so that the successors and assigns of the Buyer or the Target Companies, as the case may be, shall assume all of the obligations set forth in this <u>Section 6.8</u>.

6.9 R&W Insurance Policy.

(a) The Buyer has entered into a binder agreement (the "*Binder Agreement*") with a representation and warranty insurer (or agent thereto) providing for a summary of conditional coverage of a buyer-side representation and warranty insurance policy (the "*R&W Insurance Policy*") a copy of which is attached hereto as *Exhibit C*, naming the Buyer as the "named insured" and the Buyer Related Parties as "additional insureds". The Buyer shall use commercially reasonable efforts to satisfy the conditions set forth in the Binder Agreement to ensure that the R&W Insurance Policy is fully bound and in full force and effect at the Closing.

- (b) In connection with the R&W Insurance Policy, the Buyer shall not, and shall ensure that its Affiliates, officers, managers, members, equityholders, directors, agents and Representatives do not, amend or modify the R&W Insurance Policy in any manner adverse to the Seller (including the provision under the R&W Insurance Policy that, absent Fraud, waives subrogation rights of the insurer thereunder against the Seller) or terminate the R&W Insurance Policy once fully bound and in full force and effect at the Closing.
- (c) The cost of obtaining the R&W Insurance Policy, including all premiums and any related brokers fees, surplus line fees, and any Taxes payable thereto, shall be paid by the Buyer at or prior to the Closing.
- (d) To the extent that Buyer intends after the Closing to make or consider making a Claim under the R&W Insurance Policy and deems the assistance and cooperation of the Seller reasonably necessary or helpful in connection therewith, the Buyer shall so notify the Seller and the Seller shall use all reasonable efforts, at the Buyer's sole cost and expense, to provide such assistance and cooperation.

6.10 Employees.

- (a) <u>Available Employees</u>. Within one (1) Business Day following the Execution Date, the Seller shall update the Available Employee List to include the following information with respect to each Available Employee: (i) name; (ii) employer(s); (iii) work location; (iv) job title; (v) annualized salary or hourly wage; (vi) bonus eligibility/target for 2020; (vii) long-term incentive eligibility/target with respect to 2020; (viii) current accrued, but unused paid-time-off; (ix) hire date/start date; (x) leave status (including expected duration of any leave); (xi) details of any visa; and (xii) such other matters as the Buyer may reasonably request (the "*Updated Available Employee List*").
- (b) Offers of Employment. Within fifteen (15) Business Days of receiving the Updated Available Employee List, the Buyer or its Affiliate (including the Target Companies) shall make written offers of employment, conditioned upon the occurrence of the Closing, to each of the Available Employees to whom the Buyer elects in its sole discretion to make an offer of employment in accordance with the provisions of this Section 6.10, with all such offers subject to the satisfaction of the general hiring criteria applicable to the Buyer's hiring practices for similarly situated employees (the "Offer Employees"). Each offer of employment shall (i) be for a position that is comparable to the type of position held by each such Offer Employee immediately prior to the Closing Date (by function without regard to title), (ii) provide for a base salary or hourly rate that is no less than such Available Employee's base salary or hourly rate in effect as of immediately prior to the Closing Date, and (iii) be at a geographic location that is within fifty (50) miles of the primary location at which such Available Employee worked immediately prior to the Closing Date, in each case, through December 31, 2020. Each Offer Employee who accepts such offer of employment, including offers described in Section 6.10(c) from the Buyer or the Buyer's Affiliate (including the Target Companies), and commences such employment, is referred to herein as a "Continuing Employee".
- (c) <u>Inactive Employees</u>. With respect to Offer Employees who are Inactive Employees, the Buyer or the Buyer's Affiliate shall be required to offer employment to such employee, in accordance with the provisions of this <u>Section 6.10</u> notwithstanding their inactive or leave status (provided that such commencement of active employment occurs not later than ninety (90) days

following the Closing Date or such later time as may be required by applicable Law), and any such Inactive Employee who accepts such offer of employment shall become a Continuing Employee and the Buyer or the Buyer's Affiliate shall assume all obligations to provide compensation and benefits to each such Inactive Employee as of the Closing Date.

- (d) <u>Vacation</u>. As of the Closing Date, the Seller or one of its Affiliates (other than the Target Companies) shall pay to each Continuing Employee the amount of each such employee's (i) accrued bonus and (ii) accrued but unused vacation/paid time off, in each case, determined as of the Closing Date.
- (e) Employment-Related Liabilities. The Buyer and its Affiliates shall assume and be solely responsible for all employment related liabilities, obligations, claims or losses that (i) relate to any Continuing Employee that arise as a result of an event or events that occurred on or after the Closing Date, (ii) relate to severance payments or benefits provided to any Available Employee who does not become a Continuing Employee but becomes employed or contracted for service by the Buyer or its Affiliates (including the Target Companies) within twelve (12) months following the Closing Date, or (iii) are otherwise assumed by the Buyer or any of its Affiliates pursuant to this Section 6.10(e). The Buyer shall bear any costs related to, and shall indemnify, defend and hold harmless the Seller or any of its Affiliates (including the Target Companies) from and against any employment related liabilities, obligations, claims or losses arising out of or relating to (A) any discrimination or similar allegations stemming from the Buyer's or any of its Affiliate's selection of Offer Employees or failure to hire any Offer Employee, or (B) any breach by the Buyer or its Affiliates of this Section 6.10. The Seller shall indemnify and hold the Buyer and its Affiliates harmless from and against any and all Losses arising with respect to (1) the Continuing Employees, to the extent such claims, expenses, losses and liabilities arise from or relate to acts or omissions of the Seller or its Affiliates occurring prior to the Closing Date, (2) any Available Employees who do not become Continuing Employees (and who do not become employed or contracted for service by the Buyer or its Affiliates (including the Target Companies) within twelve (12) months following the Closing Date), (3) any Seller Benefit Plans, to the extent such claims, expenses, losses and liabilities arise from or relate to acts or omissions of the Seller or its Affiliates occurring prior to the Closing Date, (4) the Target Equity Awards in accordance with Section 2.6(d
- (f) <u>Buyer Health and Welfare Plans</u>. <u>Disclosure Schedule 6.10(f)</u> lists each material Buyer Plan. From and after Closing, the Buyer shall, and it shall cause its Affiliates (including the Target Companies) to, cause the Buyer Plans to give each Continuing Employee full credit (for all purposes, including eligibility to participate, vesting, vacation or PTO entitlement and severance benefits) for all service with the Seller or its Affiliates (including the Target Companies) and any of their predecessors prior to the Closing Date; *provided*, *however*, that (i) such service need not be credited to the extent it would result in a duplication of benefits, and (ii) such service credit shall not be given with respect to benefit accruals under any defined benefit pension plan. In addition, from and after Closing the Buyer shall, and it shall cause its Affiliates (including the Target Companies) to use commercially reasonable efforts to, (x) cause to be waived any eligibility waiting periods, any evidence of insurability requirements, any actively at work requirements, and the application of any pre-existing condition limitations under each of the Buyer Plans for the Continuing Employees and their spouses and dependents and (y) recognize any out-of-pocket expenses (including deductibles, co-pays, and out of pocket maximums) incurred by each of the

Continuing Employees and their eligible dependents under each of the health and welfare benefit plans offered to any Continuing Employee on or prior the Closing Date for purposes of determining deductibles, co-pays and out-of-pocket maximums under the Buyer Plans that are health and welfare benefit plans offered to any Continuing Employee after the Closing Date.

- (g) <u>401(k) Plan</u>. Effective as of the Closing Date, the Buyer shall, or shall cause one of its Affiliates (including a Target Company) to, permit each Continuing Employee who was a participant or eligible to participate in any Business Benefit Plan that provides defined contribution retirement benefits (the "*Seller's 401(k) Plan*") prior to the Closing to immediately following the Closing be eligible to participate in a tax-qualified defined contribution retirement plan, if any, established or designated by the Buyer (the "*Buyer's 401(k) Plan*"). The Buyer shall permit the Continuing Employees to make rollover contributions of their account balances from the Seller's 401(k) Plan that are "eligible rollover distributions" (as defined in Section 402(c)(4) of the Code) to the Buyer's 401(k) Plan, if any, including rollovers of notes evidencing any outstanding plan loans from the Seller's 401(k) Plan to an account under the Buyer's 401(k) Plan, if any.
- (h) <u>No Guaranteed Employment</u>. No provision of this <u>Section 6.10</u> shall be construed as a guarantee of continued employment of any Continuing Employee and this Agreement shall not be construed so as to prohibit the Buyer or any of its respective Subsidiaries from having the right to terminate the employment of any Continuing Employee, *provided*, that any such termination is effected in accordance with applicable Law.
- (i) No Third-Party Beneficiaries. The provisions of this Section 6.10 are for the sole benefit of the Parties hereto and nothing herein, express or implied, is intended or shall be construed to confer upon or give to any Person (including any Continuing Employee), other than the Parties to this Agreement and their respective successors and permitted assigns, any legal or equitable or other rights (including any third-party beneficiary rights) or remedies under or by reason of any provision of this Section 6.10. Nothing contained herein, express or implied: (i) shall be construed to establish, amend or modify any benefit plan, program, agreement (including any benefit plan of the Buyer); (ii) shall alter or limit the Buyer's ability to amend, modify or terminate any benefit plan, program, agreement or arrangement (including any benefit plan of the Buyer) subject to compliance with this Section 6.10; or (iii) is intended to confer upon any Continuing Employee any right to employment or continued employment for any period of time by reason of this Agreement, or any right to a particular term or condition of employment.

6.11 Use of Name.

(a) With respect to the phase-out of the use of the names, business names, trade names, brand names, registered and unregistered trademarks, logos, service marks or other source indicators of the Seller or any of its Affiliates that are used in the Business, including the use of the name and trademark "Riviera Resources" (collectively, the "Seller's Marks"), the Buyer agrees as follows: (i) as soon as reasonably practicable following the Closing, but in no event later than ninety (90) days after Closing, the Buyer shall, and after the Closing shall cause the Target Companies to, change the names of such Persons to names that do not include any of the Seller's Marks; and (ii) as soon as reasonably practicable following the Closing, but in no event later than ninety (90) days after Closing, the Buyer shall, and after the Closing shall cause the Target Companies to, cease and permanently discontinue any and all uses of any of the Seller's Marks,

and remove or cover all the Seller's Marks from or on, or destroy, any publications, signage, corporate letterhead, invoices, stationery, business cards, marketing materials, website content or other materials in the possession or under the control of the Target Companies bearing any of the Seller's Marks, and provide the Seller with written confirmation thereof; *provided*, *however*, that the Buyer shall have one hundred eighty (180) days after the Closing Date (but in any event, no longer than as required by applicable Laws, including Environmental Laws) to remove any of the Seller's Marks required to be on the Assets pursuant to any Environmental Law, health or safety Law or state or federal statutory Law.

- (b) In no event shall the Buyer or any of its Affiliates (including, after Closing, the Target Companies) use any of the Seller's Marks after Closing in any manner or for any purpose different from the use of the Seller's Marks by the Target Companies preceding the Closing, and none of them shall affix or include any of the Seller's Marks, or any colorable imitations thereof, on any publications, signage, corporate letterhead, invoices, stationery, business cards, marketing materials, website content or other materials that are created or produced after the Closing. The Buyer, for itself and its Affiliates (including, after Closing, the Target Companies), agrees that, after the Closing Date, the Buyer and its Affiliates (including, after Closing, the Target Companies) will not do business or represent themselves as having any affiliation or business relationship with the Seller or any Affiliates of the Seller (excluding, after Closing, the Target Companies), except in accordance with any separate agreement entered into, or assumed by, the Buyer or its Affiliates on one hand, and the Seller or its Affiliates, on the other hand.
- (c) The Buyer expressly acknowledges and confirms that the Seller is not transferring or assigning, and the Buyer shall not receive, any right, title or interest in or to the Seller's Marks, except the limited right to use for the sole purpose of permitting the Buyer to complete the phase-out in strict compliance with this Section 6.11. Notwithstanding anything to the contrary in this Agreement or in the other Transaction Documents, prior to the Closing Date, with the prior written consent of the Buyer (such consent not be unreasonably withheld, conditioned or delayed), the Seller shall be permitted to cause the Target Companies to assign and transfer to the Seller or its Affiliates (other than the Target Companies) any and all right, title and interest that the Target Companies have or may have in or to the Seller's Marks, including any goodwill therein.
- 6.12 Fees and Expenses. Except as expressly provided otherwise in this Agreement, all fees and expenses incurred in connection with the transactions contemplated hereby including all legal, accounting, financial advisory, consulting and all other fees and expenses of third parties incurred by a Party in connection with the negotiation and effectuation of the terms and conditions of this Agreement and the transactions contemplated hereby, shall be the obligation of the respective Party incurring such fees and expenses. Notwithstanding the foregoing, the Buyer shall pay and be fully responsible for (a) the premium, underwriting fees and any and all other fees, costs or expenses incurred in respect of the R&W Insurance Policy, as set forth in Section 6.9; (b) any and all fees, costs and expenses incurred in respect of any financing by the Buyer and its Affiliates of the transactions contemplated by this Agreement and the other Transaction Documents; (c) Transfer Taxes incurred in connection with this Agreement under Section 6.7(c); (d) fees, costs and expenses incurred with respect to the Escrow Agent and the Escrow Accounts; and (e) fees, costs, and expenses of the D&O Insurance in accordance with Section 6.8(b).

6.13 Retention of Records by the Seller.

(a) The Seller, its partners and their respective Affiliates (not including the Target Companies) shall have the right to retain (i) photocopies of all books and records, including all Tax Returns and other information and documents relating to Taxes, of the Target Companies, in each case, relating to Tax periods (or portions thereof) ending on or prior to the Closing Date (A) as required by any Governmental Authority, including any applicable Law or regulatory request or (B) as may be reasonably necessary for the Seller, its partners and their respective Affiliates to perform their respective obligations pursuant to this Agreement and the Transaction Documents, (ii) all data room materials, photocopies of bids and all books and records (including any financial analysis relating to such bids) prepared in connection with the transactions contemplated by this Agreement, including (A) any books and records that may be relevant in connection with the defense of disputes arising under this Agreement or the Transaction Documents or (B) financial information and all other accounting books and records prepared or used in connection with the preparation of financial statements of the Seller, its partners or their respective Affiliates, and (iii) any books and records to the extent reasonably necessary to prepare the Seller's, its partners' and their respective Affiliates' financial statements and Tax Returns (collectively, the "Records"). Additionally, for the shorter of (x) a period of seven (7) years from and after the Closing Date and (y) the wind-down, dissolution or liquidation of either Party, the Buyer will grant to the Seller (and/or its Representatives) reasonable access following reasonable advance notice at all reasonable times to the Records, and will afford the Seller and its Representatives the right to make copies thereof at the Seller's sole expense to the extent reasonably necessary to prepare the Seller's, its partners' or their respective Affiliates' financial statements or Tax Returns; provided, that such access shall not unreasonably interfere with the business or operations of the Buyer or its employees. The Buyer will maintain the Records in accordance with the Buyer's document retention policies.

(b) Without limiting the foregoing Section 6.13(a), at and after the Closing, the Buyer shall, and it shall cause its Affiliates (including the Target Companies) to, (i) reasonably cooperate with the Seller, its Affiliates and their respective Representatives, at the Seller's sole expense, in connection with the Seller's (or its Affiliate's) preparation of financial statements and the Seller's and its Affiliates compliance with their financial or other reporting requirements and audits (including inquiries), including those required under the Securities Act or the Exchange Act, or as otherwise required by any regulatory body or regime or stock exchange to which the Seller or any of its Affiliates is subject and (ii) upon reasonable request from the Seller, afford the Seller, its Affiliates and their respective Representatives, during normal business hours, upon reasonable notice, access to the books, records (including employee and personnel records) and employees of the Target Companies to the extent reasonably necessary in connection with the activities set forth in clause (i) above.

6.14 Public Announcements.

(a) No Party shall issue any public announcement, statement or other disclosure without first having obtained the prior written consent of the other Party, except as may otherwise be required by applicable Law or by any stock exchange listing requirement.

- (b) Except for disclosures permitted by the terms of the Confidentiality Agreement or Section 6.14(a), from the Execution Date until the Closing Date, each Party shall keep confidential in accordance with the terms of the Confidentiality Agreement all Pre-Closing Confidential Information. If the Closing occurs, the Confidentiality Agreement shall terminate other than with respect to information relating solely to the Seller and its Affiliates (other than the Target Companies).
- 6.15 No Solicitation of Other Bids. From the Execution Date until the earlier of the Closing Date or the termination of this Agreement, the Seller agrees that it will not, and will not authorize or permit any of its Affiliates or any of its or their Representatives to, directly or indirectly, engage in any negotiations with any Person other than the Buyer and its Affiliates regarding any Acquisition Proposals; provided, however, that the Buyer hereby acknowledges that prior to the date of this Agreement, the Seller, its Affiliates and its and their respective Representatives provided information relating to the Target Companies and the Business and afforded access to, and engaged in discussions with, other Persons in connection with an Acquisition Proposal. Immediately following the execution of this Agreement on the Execution Date, the Seller shall, and shall cause its Affiliates and its and their Representatives to, immediately cease any ongoing discussions with Persons other than the Buyer in connection with any Acquisition Proposal and immediately discontinue any access afforded to such Persons to information relating to the Target Companies and the Business. Notwithstanding anything to the contrary herein, the Seller, its Affiliates and its and their respective Representatives may respond to any unsolicited proposal regarding an Acquisition Proposal by indicating that the Seller and the Target Companies are subject to an exclusivity agreement and are unable to provide any information related to the Target Companies or the Business or entertain any proposals or offers or engage in any negotiations or discussions concerning an Acquisition Proposal for as long as this Agreement remains in effect. For purposes hereof, "Acquisition Proposal" will mean any inquiry, proposal or offer from any Person (other than the Buyer or any of its Affiliates) concerning (i) a merger, consolidation, liquidation, recapitalization, equity exchange or other business combination transaction involving the Target Companies; (ii) the issuance or acquisition of any Equity Interests in the Target Companies; (iii) the sale, lease, exchange or other disposition of any material portion of the Assets outside of the Ordinary Course of Business; or (iv) any transaction that would be inconsistent with the transactions contemplated by this Agreement or any Transaction Document, or interfere with the performance of any obligations hereunder or thereunder.
- 6.16 <u>Payoff Letters</u>. At least three (3) Business Days prior to the Closing Date, the Seller shall, and shall cause the Target Companies to, use commercially reasonable efforts to cause each payee of any portion of the Payoff Indebtedness to deliver a Debt Payoff Letter to the Seller or the Target Companies, copies of which shall promptly be delivered to the Buyer.
 - 6.17 Seller Credit Support; Termination of Agreements and Intercompany Arrangements.
 - (a) <u>Disclosure Schedule 6.17(a)</u> lists the guarantees, surety bonds and letters of credit provided by the Seller or any of its Affiliates (other than the Target Companies) for the benefit of the Target Companies. True and correct copies of such guarantees, surety bonds and letters of credit have been made available to the Buyer.
 - (b) Subject to Section 6.17(d), and without duplication of any adjustments to the Unadjusted Purchase Price included in the Adjustment Amount (including the Accrued Reimbursable Expenses Amount), the Seller shall, and shall cause the Target Companies to, take

such action and make such payments as may be necessary so that concurrently with the Closing, the Target Companies, on the one hand, and the Seller and its other Affiliates, on the other hand, shall settle, discharge, offset, pay, repay in full, terminate or extinguish all intercompany loans, notes, and advances regardless of their maturity and all intercompany receivables and payables for the amount due, including any accrued and unpaid interest to but excluding the date of payment.

- (c) Subject to Section 6.17(d), the Seller and the Target Companies shall take such actions as may be necessary to terminate, concurrently with the Closing Date, (i) all Contracts between any Target Company, on the one hand, and the Seller or any of its Affiliates (other than other Target Companies), on the other hand, and (ii) all other Contracts and arrangements set forth on Disclosure Schedule 4.16.
- (d) The Seller, the Buyer and the Target Companies shall use commercially reasonable efforts to cause, concurrently with Closing, each of the guarantees, surety bonds and letters of credit listed in <u>Disclosure Schedule 6.17(d)</u> (collectively, the "*Required Intercompany Bonds*") to be assigned and transferred to, or assumed by, Buyer or an Affiliate of Buyer (excluding the Target Companies), settled, discharged, offset, terminated or extinguished, in each case, in accordance with the terms and conditions of the Required Intercompany Bonds such that the Required Intercompany Bonds are no longer (i) provided by the Seller or any of its Affiliates for the benefit of the Target Companies and (ii) the obligation or liability of the Seller or any of its Affiliates. To the extent that any Required Intercompany Bonds are not assigned or transferred to, or assumed by, Buyer or an Affiliate of Buyer (excluding the Target Companies), settled, discharged, offset, terminated or extinguished concurrently with Closing, the Seller, the Buyer and the Target Companies shall continue to use commercially reasonable efforts to cause such assignment or transfer, or assumption, settlement, discharge, offset, termination or extinguishment as soon as reasonably practicable following the Closing, and the Buyer shall defend, indemnify and hold harmless the Seller Parties from and against any and all Losses, as incurred, arising out of or relating to any such remaining Required Intercompany Bonds.
- (e) Prior to Closing, each of the Target Companies shall, and the Seller shall cause each of the Target Companies to, assign to Seller or its Affiliate (other than a Target Company) any and all employment contracts to which such Target Company is a party or by which such Target Company is bound, including without limitation that certain Release Agreement dated April 27, 2020, executed by and between the Target and Greg Harper.
- 6.18 <u>Termination and Release of Claims</u>. Prior to the Closing, each of the Buyer Parties, the Seller Parties and their respective Affiliates, as applicable, shall take any and all actions as may be necessary or required to effect each of the transactions and other items contemplated by and set forth on <u>Schedule 6.18</u>, in each case as of and contingent upon the occurrence of Closing (collectively, the "*Termination and Release*").
- 6.19 <u>Litigation Cooperation</u>. For a period of seven (7) years from and after the Closing Date, the Seller Parties shall, at the Buyer's sole expense, (i) cooperate in good faith with the Buyer Parties and their counsel in providing information and assistance in connection with any contest or defense by the Buyer Parties of any Proceedings or Claims arising out of or related to the acts or omissions of the Target Companies prior to the Closing Date, including the matters set forth on <u>Disclosure Schedule 4.14(a), (b) and (c)</u>, and (ii) make reasonably available to the Buyer and its Representatives the personnel of the Seller

Parties and provide reasonable access following reasonable advance notice at all reasonable times to the books and records of the Seller Parties (including emails and electronic files), and afford the Buyer and its Representatives the right to make copies thereof, in each case, as shall be reasonably necessary in connection with such contest or defense. Notwithstanding the foregoing, nothing in this Section 6.19 shall in any way limit or interfere with the winding down, dissolution or liquidation of the Seller or any of its Affiliates and their respective businesses.

6.20 Financing Cooperation.

- (a) From the date hereof until the Closing (or the earlier termination of this Agreement pursuant to Section 9.1), subject to the limitations set forth in this Section 6.20(a), and unless otherwise agreed by the Buyer, the Seller Parties will use their reasonable best efforts to cooperate, and use their reasonable best efforts to cause their respective Subsidiaries (and each of their respective officers, directors, employees, accountants, consultants, legal counsel, affiliates and agents) to cooperate, with the Buyer as reasonably requested by the Buyer or the Debt Financing Sources in connection with the arrangement of the Debt Financing at the Buyer's sole cost and expense. Such cooperation will include and be limited to using reasonable best efforts to:
 - (i) make officers of appropriate seniority reasonably available, with appropriate advance notice and at times reasonably acceptable to the Seller Parties, for participation in virtual bank meetings, additional bank calls during normal business hours at times reasonably agreed, virtual due diligence sessions, reasonable assistance in the preparation of confidential information memoranda and similar customary documents as may be reasonably requested by the Buyer or any Debt Financing Source, in each case, with respect to information relating to the Target Companies in connection with customary marketing efforts of the Buyer for all or any portion of the Debt Financing;
 - (ii) furnish the Buyer and the Debt Financing Sources with copies of such historical financial data with respect to the Target Companies which are prepared by the Seller Parties in the ordinary course of business, but in any case, limited to: (A) audited consolidated balance sheets and related statements of operations, stockholders' equity and cash flows of the Target for the three most recently completed fiscal years ended at least 120 days prior to the Closing Date, and (B) unaudited consolidated balance sheet and related unaudited consolidated statements of operations, stockholders' equity and cash flows of the Target Companies, on a consolidated basis, as of and for each subsequent fiscal quarter (other than the fourth fiscal quarter of any fiscal year) at least 60 days prior to the Closing Date (in each case, together with the corresponding comparative period from the prior fiscal year, it being agreed and understood that no such financial data is available with respect to any fiscal year prior to the 2018 fiscal year);
 - (iii) provide information reasonably requested by Buyer or the Debt Financing Sources regarding the Target Companies at least four (4) Business Days prior to the Closing Date under applicable "know your customer," anti-money laundering rules and regulations and the USA PATRIOT Act of 2001, in each case, requested in writing at least nine (9) Business Days prior to the Closing Date;

(iv) provide reasonable and customary authorization letters to the Debt Financing Sources authorizing the distribution of information relating to the Target Companies to prospective lenders (and not containing any other confirmations or undertakings other than containing confirmation that public side versions of documents to be distributed to prospective lenders containing such information, if any, do not contain material nonpublic information regarding the Target Companies or their securities) and subject to customary confidentiality provisions;

(v) assist with the production of factual information required in connection with the preparation of any credit agreement, pledge and security documents, perfection certificates, mortgages, deeds of trust, hedging agreements, legal opinion support certificates or other definitive financing documents or other documents related to the Debt Financing (including schedules, insurance certificates, and evidence of corporate authority) as may be reasonably requested by the Buyer;

(vi) provide all existing items related to the Real Property Assets required by the Debt Financing that are requested by the Buyer or any Debt Financing Source prior to the date hereof, which are in the possession of the Seller or any of its affiliates (for the avoidance of doubt, it is agreed and understood that all such items have been delivered to the Buyer Parties prior to the date hereof) and facilitate the pledging of collateral owned by the Target Companies as reasonably requested by the Buyer; *provided*, that no pledge shall be effective until the Closing; and

(vii) (A) allow the usual and customary use of the logos of the Target Companies in connection with any debt financing (provided such logos shall be used solely in a manner that is not intended or reasonably likely to harm, disparage or otherwise adversely affect the Target Companies' reputation or goodwill) and (B) in connection with the Closing, allow the placement of customary advertisements in financial and other newspapers and periodicals or on a home page or similar place for dissemination of customary information on the Internet or worldwide web as the Debt Financing Sources may choose, and circulate similar promotional materials in the form of a "tombstone" or otherwise describing aspects of the transactions contemplated hereby and the Debt Financing;

provided, that Buyer shall promptly upon receipt of a reasonably detailed invoice therefor, reimburse the Seller Parties for any documented out of pocket expenses and costs incurred in connection with the obligations of the Target Companies under this Section 6.20; provided, further, that nothing in this Agreement shall require the Target Companies to cause the delivery of (A) legal opinions or reliance letters, (B) any financial information in a form not customarily prepared by the Seller Parties with respect to such period or (C) any financial information with respect to a fiscal period that has not yet ended or has ended less than sixty (60) days prior to the date of such request (or, in the case of annual financial statements, one hundred twenty (120) days prior to such request).

- (b) Notwithstanding anything to the contrary contained in this Agreement (including this Section 6.20):
 - (i) Nothing in this Agreement (including this Section 6.20) shall require any such cooperation to the extent that it would: (A) require the Seller Parties to pay any commitment or other fees, reimburse any expenses or otherwise incur any liabilities or give any indemnities prior to the Closing for which it has not received prior reimbursement by the Buyer; (B) materially interfere with the ongoing business or operations of the Target Companies; (C) require the Target Companies to enter into any agreement or other document effective prior to the Closing (other than authorization letters, confirmations and undertakings described in this Section 6.20) or agree to any change or modification of any existing agreement that would be effective prior to the Closing (other than customary payoff letters) (for the avoidance of doubt, any contracts or amendments contemplated to be entered into by the Seller Related Parties in connection with the Debt Commitment Letter (other than this Agreement, the Transaction Documents, or any amendments hereto or thereto) shall be effective no earlier than concurrent with the Closing); (D) require, prior to the Closing, the Target, any of its Subsidiaries or any of their respective boards of directors (or equivalent bodies) to approve or authorize the Debt Financing; (E) require any action that would conflict with or violate the organizational documents of the Target Companies or any Laws, orders or result in the contravention of, or that would reasonably be expected to result in a violation or breach of, or default under, confidentiality or similar provision of any contract (including any Contract) to which any Target Company is a party; (F) cause any director, officer, employee or stockholder of any Target Company to incur any personal liability; (G) provide access to or disclose information that would jeopardize any attorney-client privilege of any Target Company; (H) prepare separate financial statements for any Subsidiary of the Target or change any fiscal period or prepare any financial statements or information that are not available to it and prepared in the ordinary course of its financial reporting practice or (I) require the Seller Parties to make any representations in connection with the Debt Financing except as provided in Section 6.20(a)(iv); and
 - (ii) no action, liability or obligation (including any obligation to pay any commitment or other fees or reimburse any expenses) of the Target, its Subsidiaries, or any of their respective Representatives under any certificate, agreement, arrangement, document or instrument relating to the Debt Financing (other than with respect to customary authorization letters) shall be effective until the Closing.
- (c) Buyer shall indemnify and hold harmless the Target Companies against any claim, loss, damage, injury, liability, judgment, award, penalty, fine, cost (including cost of investigation), reasonable and documented out-of-pocket expenses (including reasonable and documented out-of-pocket fees and expenses of counsel and third-party accountants) or settlement payment incurred as a result of such cooperation or the Debt Financing (or, if applicable, Substitute Debt Financing) and any information used in connection therewith. All non-public or other confidential information provided by the Target and its Affiliates and Representatives pursuant to this <u>Section 6.20</u> shall be kept confidential in accordance with <u>Section 6.2</u>. Notwithstanding anything to the contrary contained in this Agreement, in no event shall the reasonable best efforts of the Buyer require or be deemed or construed to require the Buyer to pay any fees in excess of those contemplated by the Debt Commitment Letter (whether to secure waiver of any conditions contained therein or otherwise).

6.21 Financing Efforts.

(a) The Buyer shall use its reasonable best efforts to take (or cause to be taken), all actions and do (or cause to be done), all things necessary, proper or advisable to maintain in effect, arrange and consummate the Debt Financing on or before the Closing Date on the terms described in the Debt Commitment Letter, including using reasonable best efforts to: (i) satisfy, or cause the satisfaction of, or obtain a waiver of, on a timely basis all conditions applicable to the Buyer under the Debt Commitment Letter on or prior to the Closing Date; (ii) negotiate and enter into definitive agreements (to the extent not already entered into) with respect to the Debt Financing on the terms and subject to the conditions contemplated by the Debt Commitment Letter (including all "market flex" provisions thereof) (the "Definitive Agreements"); (iii) subject to clause (iv) of this Section 6.21, consummate the Debt Financing at or prior to or substantially concurrently with Closing and (iv) enforce its rights under the Debt Commitment Letter to the extent necessary to cause the Debt Financing Sources to provide the proceeds of the Debt Financing to the Buyer in accordance with the Debt Commitment Letter in the event of a breach by the Debt Financing Sources or other parties thereto to the extent such breach results in a failure or material delay to consummate the transactions under this Agreement. The Buyer shall not, without the prior written consent of the Seller, permit any amendment or modification to be made to, consent to any waiver of any provision or remedy of, or replace the Debt Commitment Letter if such amendment, modification, waiver or replacement (x) reduces the aggregate amount of the Debt Financing (including by changing the amount of fees to be paid or original issue discount thereof) below the amount necessary to consummate the transaction contemplated by this Agreement, unless Buyer will otherwise have available cash sufficient to consummate the Closing, or (y) imposes any new or additional condition to the Debt Financing, or otherwise amends, modifies or expands any condition to the Debt Financing in any respect. For the avoidance of doubt, subject to clauses (x) and (y) above, nothing herein shall prevent the Buyer from replacing or amending the Debt Commitment Letter in order to add lead arrangers, bookrunners, syndication agents or similar entities which had not executed the Debt Commitment Letter as of the date hereof and amend the economic or other arrangements among the existing and additional lenders, lead arrangers, bookrunners, syndication agents or similar entities, or to replace the commitment parties with new commitment parties if the addition of such additional parties does not reduce the aggregate net amount to be funded under the Debt Commitment Letter at the Closing as compared to the amount to be funded at the Closing under the Debt Commitment Letter as in effect on the date hereof.

(b) The Buyer shall keep the Seller reasonably informed in reasonable detail of the status of its efforts to arrange the Debt Financing and shall provide senior members of management of the Company a reasonable opportunity to review on any materials that are to be presented to them during any meetings conducted in connection with the Debt Financing. Without limiting the generality of the foregoing, the Buyer shall give the Seller prompt (and in any event, within three (3) Business Days) written notice (i) of any default or breach that would materially delay, impede or prevent the funding of the Debt Financing (or any event that, with our without notice, lapse of time or both, would reasonably be expected to give rise to any default or breach that would materially delay, impede or prevent the funding of the Debt Financing) by any party to the Debt Financing of which the Buyer becomes aware, and (ii) of the receipt of any written notice or other communication from any Person with respect to any (A) actual default, breach, termination or repudiation by any party to the Debt Commitment Letter or (B) material dispute or disagreement between or among any parties to the Debt Commitment Letter or any other event or condition, in each case, which would cause or reasonably be expected to cause any condition to the Debt Financing to fail to be satisfied. In the event that any portion of the Debt Financing is otherwise

unavailable, the Buyer shall give the Seller prompt (and in any event, within three (3) Business Days) written notice thereof (and the reasons therefor) and shall use its reasonable best efforts to arrange and to obtain (having conditions to closing that are not more onerous to the Buyer, taken as a whole, than those set forth in the Debt Commitment Letter) a Substitute Debt Financing as promptly as practicable following the occurrence of such event, but, in no event, later than the date that the Buyer is required to consummate the transactions contemplated hereby in accordance with this Agreement. The Buyer shall deliver promptly to the Seller true, accurate and complete copies of all agreements (provided that any fee letter will be redacted in a customary manner) pursuant to which any such alternative sources of the Substitute Debt Financing shall have committed to provide any portion of the Debt Financing (the "Substitute Financing Commitments"). The terms of Section 6.21(a) and Section 6.21(b) with respect to the Debt Financing and the Debt Commitment Letter, shall apply to the Substitute Debt Financing and the Substitute Financing Commitments, *mutatis mutandis*.

- (c) Notwithstanding anything to the contrary, it is understood and agreed that the Buyer's obligations hereunder, and Closing, are not in any manner subject to or conditioned upon the Buyer's receipt of all or any portion of the Debt Financing or any Substitute Debt Financing.
- 6.22 <u>Services</u>. The Buyer and the Seller will work in good faith to finalize Schedule 1.1 to the Transition Services Agreement prior to the Closing; *provided*, *however*, in no event will the Seller or its Affiliates be obligated to perform services that they are not performing as of the Execution Date. The Parties shall make appropriate revisions to the Transition Services Agreement to correspond to the agreed-upon Services.

ARTICLE VII CONDITIONS TO THE CLOSING

- 7.1 <u>Conditions to Each Party's Obligation</u>. The respective obligations of each Party to consummate the transactions contemplated by this Agreement shall be subject to the satisfaction (or waiver, if permissible under applicable Law), at or prior to Closing, of the following condition:
 - (a) No Law, injunction, Order, judgment or ruling enacted, promulgated, issued, entered, amended or enforced by any Governmental Authority shall be in effect enjoining, restraining, preventing or prohibiting consummation of the transactions contemplated by this Agreement or making the consummation of the transactions contemplated by this Agreement illegal (a "*Restraint*").
 - (b) No Proceeding initiated by a Governmental Authority shall be pending before any Governmental Authority seeking a Restraint.
- 7.2 <u>Seller's Conditions</u>. The respective obligations of the Seller to consummate the transactions contemplated by this Agreement shall be subject to the satisfaction (or waiver, if permissible under applicable Law), at or prior to Closing, of each of the following conditions:
 - (a) (i) The Buyer Fundamental Representations shall be true and correct in all material respects (other than those portions of any such representations and warranties that are qualified as to "materiality", which portions shall be true and correct in all respects) as of the Execution Date and on and as of the Closing Date as if made on and as of the Closing Date (other than any such

representation or warranty that is made as of a specified date, which shall be true and correct as of such specified date), and (ii) all representations and warranties of the Buyer contained in <u>Article V</u>, other than the Buyer Fundamental Representations, shall be true and correct as of the Execution Date and on and as of the Closing Date as if made on and as of the Closing Date (other than any such representation or warranty that is made as of a specified date, which shall be true and correct as of such specified date), except, in the case of this clause (ii), for such breaches, if any, of representations and warranties as would not, individually or in the aggregate, prevent, impede or otherwise adversely affect in any material respect the consummation at Closing of the transactions contemplated by this Agreement (without giving effect to any "materiality" qualifiers contained therein).

- (b) The Buyer shall have performed and complied in all material respects with all covenants required by this Agreement to be performed or complied with by the Buyer on or prior to the Closing, including the receipt by the Seller of all amounts required to be paid by the Buyer at the Closing pursuant to Section 8.2(b).
 - (c) The Buyer shall have delivered the items required to be delivered by the Buyer pursuant to Section 8.2(b).
- 7.3 <u>Buyer's Conditions</u>. The obligations of the Buyer to consummate the transactions contemplated by this Agreement shall be subject to the satisfaction (or waiver, if permissible under applicable Law), at or prior to Closing, of each of the following conditions:
 - (a) (i) The Seller Fundamental Representations shall be true and correct in all material respects (other than those portions of any such representations and warranties that are qualified as to "materiality" or a "Target Material Adverse Effect", which portions shall be true and correct in all respects) as of the Execution Date and on and as of the Closing Date as if made on and as of the Closing Date (other than any such representation or warranty that is made as of a specified date, which shall be true and correct as of such specified date), (ii) Section 4.17(y) (Absence of Target Material Adverse Effect) shall be true and correct in all respects as of the Execution Date and on and as of the Closing Date as if made on and as of the Closing Date, and (iii) all representations and warranties of the Seller contained in Article III and Article IV other than the Seller Fundamental Representations and Section 4.17(y) shall be true and correct as of the Execution Date and on and as of the Closing Date as if made on and as of the Closing Date (other than any such representation or warranty that is made as of a specified date, which shall be true and correct as of such specified date), except, in the case of this clause (iii), for such breaches, if any, of representations and warranties as would not, individually or in the aggregate, have a Target Material Adverse Effect (without giving effect to any "materiality" or "Target Material Adverse Effect" qualifiers contained therein).
 - (b) The Seller shall have performed and complied in all material respects with all covenants required by this Agreement to be performed or complied with by the Seller on or prior to the Closing.
 - (c) Each Required Third-Party Consent set forth in <u>Disclosure Schedule 7.3(c)</u> and each Authorization set forth in <u>Disclosure Schedule 7.3(c)</u> shall have been obtained and each Required Notification set forth in <u>Disclosure Schedule 7.3(c)</u> shall have been delivered.

- (d) The Seller shall have delivered the items required to be delivered by the Seller pursuant to Section 8.2(a).
- 7.4 <u>Frustration of Closing Conditions</u>. Neither the Buyer nor the Seller may rely on the failure of any condition set forth in this <u>Article VII</u> to be satisfied if such failure was primarily caused by such Party's failure to act in good faith or to use its commercially reasonable efforts to cause the Closing to occur as required by <u>Section 6.6(a)</u>.

ARTICLE VIII CLOSING

8.1 <u>Time and Place of Closing</u>. The consummation of the transactions contemplated hereby (the "*Closing*") shall take place at the offices of Kirkland & Ellis LLP, 609 Main Street, 47th Floor, Houston, Texas 77002 (or remotely via exchange of electronic documents) at 10:00 a.m. local time on the later to occur of: (a) October 1, 2020 and (b) the second (2nd) Business Day after the conditions to Closing set forth in <u>Article VII</u> (other than actions to be taken or items to be delivered at Closing) have been satisfied or waived, or such other time and place as to which the Parties agree in writing (the date on which Closing occurs, the "*Closing Date*").

8.2 Deliveries at the Closing.

- (a) <u>Seller's Closing Deliverables</u>. Subject to the terms and conditions of this Agreement, at the Closing, the Seller will execute and deliver (or cause to be executed and delivered) each of the following documents (where the execution and delivery of the documents is contemplated), deliver (or cause to be delivered) each of the following items (where the delivery of the items is contemplated) and take (or cause to be taken) each of the following actions (where the taking of action is contemplated):
 - (i) to the Buyer, a certificate, dated as of the Closing Date duly executed by an officer of the Seller certifying the satisfaction of the conditions specified in <u>Section 7.3(a)</u> and <u>Section 7.3(b)</u> (the "*Seller's Closing Certificate*");
 - (ii) to the Buyer, a counterpart of joint written release instructions to the Escrow Agent, duly executed by an authorized officer of the Seller authorized to deliver such instructions under the Escrow Agreement, instructing the Escrow Agent to disburse from the Deposit Escrow Account to the Seller an amount equal to the remainder of the Deposit;
 - (iii) to the Buyer, a counterpart of the Target Assignment Agreement, duly executed by an authorized Representative of the Seller;
 - (iv) to the Buyer, a counterpart of the Transition Services Agreement, duly executed by an authorized Representative of Riviera Operating;
 - (v) to the Buyer, a valid Form W-9 from the Seller's parent company, Riviera Resources, Inc.;

- (vi) to the Buyer, a certificate of the Secretary of State of the State of Delaware or Oklahoma, as applicable, as to the legal existence and good standing of the Target Companies;
- (vii) to the Buyer, a certificate of the secretary from each Target Company certifying the Organizational Documents of such Target Company;
- (viii) to the Buyer, a resignation letter, executed by each officer, director or manager of each Target Company, effective immediately upon consummation of the Closing;
- (ix) to the Buyer, all agreements, instruments, documents and other evidence as may be reasonably requested by the Buyer that are necessary to effect the Termination and Release as required by <u>Section 6.18</u>;
- (x) to the Buyer, all agreements, instruments, documents and other evidence as may be reasonably requested by the Buyer that are necessary to effect the termination of the Contracts and other arrangements required by <u>Section 6.17(c)</u>;
- (xi) to the Buyer, the LINN Tax Matters Termination Agreement, duly executed by an authorized representative of the Seller and each of Seller's Affiliates that is a party to the LINN Tax Matters Agreement; and
- (xii) to the Buyer, a payoff letter duly executed by each holder of Payoff Indebtedness, in which the payee shall (A) provide the dollar amount required to be paid with respect to such Payoff Indebtedness in order to fully pay off such Payoff Indebtedness as of the Closing and to release all Liens thereunder upon such payment; and (B) either (1) agree to either file all necessary UCC and Lien terminations following the Closing Date, or (2) expressly authorize the Seller or the Buyer (or its designee) to have any such documents filed on the lenders' behalf (each such payoff letter, a "Debt Payoff Letter").
- (b) <u>Buyer's Closing Deliverables</u>. Subject to the terms and conditions of this Agreement, at the Closing, the Buyer will execute and deliver (or cause to be executed and delivered) each of the following documents (where the execution and delivery of the documents is contemplated), deliver (or cause to be delivered) each of the following items (where the delivery of the items is contemplated) and take (or cause to be taken) each of the following actions (where the taking of action is contemplated):
 - (i) to the Seller (or its designee), in cash by wire transfer of immediately available funds, to the account(s) designated by the Seller at least two (2) Business Days prior to the Closing, the Closing Amount;
 - (ii) the payoff amounts set forth in the Debt Payoff Letters (the "*Payoff Amounts*") in cash by wire transfer of immediately available funds, on behalf of the Target Companies, in such amounts and to such accounts as set forth in the Debt Payoff Letters;

- (iii) to the Seller, a certificate executed by an officer of the Buyer certifying the satisfaction of the conditions specified in Section 7.2(a) and Section 7.2(b) (the "Buyer's Closing Certificate");
- (iv) to the Seller, a counterpart of joint written release instructions to the Escrow Agent, duly executed by an authorized officer of the Buyer authorized to deliver such instructions under the Escrow Agreement, instructing the Escrow Agent to disburse from the Deposit Escrow Account to the Seller an amount equal to the remainder of the Deposit;
 - (v) to the Seller, a counterpart of the Target Assignment Agreement, duly executed by an authorized Representative of the Buyer;
 - (vi) to the Seller, a counterpart of the Transition Services Agreement, duly executed by an authorized Representative of the Buyer;
- (vii) to the Seller, all agreements, instruments, documents and other evidence as may be reasonably requested by the Seller that are necessary to effect the Termination and Release as required by <u>Section 6.18</u>; and
- (viii) to the applicable Person or Persons, as required by the Binder Agreement, the premium for (A) the effective R&W Insurance Policy and (B) the R&W Insurance Policy, along with any related brokers' fees or other similar fees, by wire transfer of immediately available funds in U.S. dollars.

ARTICLE IX TERMINATION

- 9.1 Termination at or Prior to the Closing. This Agreement may be terminated prior to the Closing:
 - (a) by the mutual written consent of the Seller and the Buyer;
- (b) by the Seller or the Buyer at any time after the Outside Date, upon written notice to the other Party, if the Closing has not been consummated on or before the Outside Date; *provided*, that, if on the Outside Date the condition to Closing set forth in either Section 7.1(a) (No Restraints) or Section 7.1(b) (No Proceedings Seeking Restraints) has not been satisfied, but all other conditions to Closing in Article VII have been satisfied or are capable of being satisfied, then the Outside Date may be extended on one or more occasions to a date not later than ninety (90) days after the Outside Date at the option of the Buyer by providing written notice to the Seller at or before 11:59 p.m. Houston, Texas time on the Outside Date; and *provided*, *further*, that the right to terminate this Agreement under this Section 9.1(b) will not be available to a Party if the other Party has filed (and is then pursuing) an action seeking specific performance as permitted by Section 11.12;
- (c) by the Seller or the Buyer, upon written notice to the other Party, if any Restraint having the effect set forth in Section 7.1(a) is in effect and has become final and nonappealable;

- (d) by the Seller, upon written notice to the Buyer, if the Buyer has breached any of its representations or warranties set forth in this Agreement (or if any such representations or warranties fail to be true) or the Buyer has failed to perform any of its covenants or agreements set forth in this Agreement (other than the Buyer's covenant to deliver the Deposit to the Escrow Agent pursuant to Section 2.2(a), which shall be handled exclusively in Section 9.1(f) below), which breach or failure (i) would (if it occurred or was continuing as of the Closing Date) give rise to the failure of a condition set forth in Section 7.2(a) or Section 7.2(b) and (ii) is incapable of being cured, or is not cured by the Buyer by the earlier of (A) the Outside Date and (B) thirty (30) days following receipt of written notice from the Seller of such breach or failure;
- (e) by the Buyer, upon written notice to the Seller, if the Seller has breached any of its representations or warranties set forth in this Agreement (or if any such representations or warranties fail to be true) or the Seller has failed to perform any of its covenants or agreements set forth in this Agreement, which breach or failure (i) would (if it occurred or was continuing as of the Closing Date) give rise to the failure of a condition set forth in Section 7.3(a) or Section 7.3(b) and (ii) is incapable of being cured, or is not cured by the Seller by the earlier of (A) the Outside Date and (B) thirty (30) days following receipt of written notice from the Buyer of such breach or failure; or
- (f) by the Seller, upon written notice to the Buyer, if the Buyer does not deposit the Deposit with the Escrow Agent pursuant to Section 2.2(a) at or before 11:59 p.m. Houston, Texas time on the first Business Day after the Execution Date.

9.2 Effect of Termination.

- (a) In the event of the termination of this Agreement as provided in <u>Section 9.1</u>, written notice thereof will be given to the other Party, specifying the provision of this Agreement pursuant to which such termination is made, and this Agreement shall forthwith become void and there shall be no further obligation on the part of any Party or its respective Representatives, except that the provisions of <u>Section 6.2(d)</u>, <u>Section 6.2(e)</u>, <u>Section 6.12</u>, this <u>Section 9.2</u> and <u>Article XI</u> (other than <u>Section 11.12</u>), and the Confidentiality Agreement (which shall continue pursuant to its terms) shall survive any termination of this Agreement.
- (b) If the Seller validly terminates this Agreement under Section 9.1(d), then the Seller shall be entitled to receive (x) the entire Deposit from the Deposit Escrow Account, plus (y) an additional amount in cash equal to five percent (5%) of the Unadjusted Purchase Price (items (x) and (y) collectively, the "Termination Fee") for the sole account and use of the Seller as liquidated damages under this Agreement, which remedy shall be the sole and exclusive remedy available to the Seller for any breach by the Buyer. The Parties acknowledge and agree that if the Seller receives the Termination Fee under this Section 9.2(b), the actual amount of damages resulting from such a termination would be difficult if not impossible to determine accurately because of, among other things: (w) the unique nature of this Agreement; (x) the unique nature of the Assets; (y) the uncertainties of applicable commodity markets; and (z) differences of opinion with respect to such matters that inherently make damages difficult to ascertain. Consequently, in such case, the Parties further acknowledge and agree that the liquidated damages associated with receipt by the Seller of the Termination Fee are a reasonable estimate by the Parties of such damages under the circumstances and do not constitute a penalty.

- (c) If this Agreement is terminated under Section 9.1 and the Seller is not entitled to receive the Termination Fee under Section 9.2(b), then the Buyer shall be entitled to receive the entirety of the Deposit. If the Buyer validly terminates this Agreement under Section 9.1(e), then the Buyer shall be entitled to seek any and all rights or remedies against the Seller available at law, in equity or otherwise with respect to any willful and intentional breach of this Agreement by the Seller, in which case the Buyer shall only be entitled to recover from the Seller the actual Losses suffered by the Buyer as a consequence of such termination; provided that Buyer's recovery shall not exceed an amount equal to fifteen percent (15%) of the Unadjusted Purchase Price.
- (d) Once the Buyer or the Seller becomes entitled to receive the Deposit in accordance with Section 2.2(b) or this Section 9.2, each of the Buyer and the Seller covenants and agrees that such Party shall promptly, and in any event within three (3) Business Days, execute and deliver to the Escrow Agent joint written instructions instructing the Escrow Agent to disburse from the Deposit Escrow Account to the Buyer or the Seller, as applicable, an amount equal to the Deposit. If the Seller becomes entitled to receive the Termination Fee in accordance with this Section 9.2, the Buyer covenants and agrees that it shall promptly, and in any event within three (3) Business Days, make available to the Seller via wire transfer of immediately available funds in such account specified by Seller an amount equal to the remainder of the Termination Fee (net of the Deposit, which shall be disbursed to the Seller in accordance with the immediately preceding sentence).
- (e) Notwithstanding anything to the contrary in this Agreement (other than as set forth in Section 9.2(h)), payment of the Termination Fee (to the extent payable pursuant to Section 9.2(b)) to the Seller shall be deemed liquidated damages and constitute the sole and exclusive monetary remedy of the Seller, the Target Companies, each of their respective Affiliates and the respective Representatives of each of the foregoing (collectively, the "Seller Related Parties") against Buyer, Sponsor, the Debt Financing Sources, and each of the foregoing Person's former, current or future Affiliates, portfolio companies, investment vehicles, controlling persons, officers, directors, employees, equityholders, partners, members, managers, agents, representatives, successors and assigns (collectively, the "Buyer Related Parties") for, and the Seller Related Parties shall be deemed to have waived all other remedies with respect to, any Losses suffered as a result of the failure of the Closing to occur or for a breach or failure to perform hereunder or otherwise (in any case, whether willfully, intentionally, unintentionally or otherwise) and none of the Buyer Related Parties shall have any further Liability relating to or arising out of this Agreement or the transactions contemplated herein. No Seller Related Party shall be entitled to bring, and the Seller shall cause all other Seller Related Parties not to bring, and shall in no event support, facilitate or encourage, the bringing of any Proceeding (under any legal theory, whether sounding in law or in equity (in each case whether for breach of contract, in tort or otherwise)) against a Buyer Related Party with respect to, arising out of, or in connection with, the failure of the Closing to occur or for a breach or failure to perform hereunder or otherwise (in any case, whether willfully, intentionally, unintentionally or otherwise), other than a Proceeding to seek to obtain the payment of the Termination Fee or injunction, specific performance or other equitable relief solely to the extent permitted by Section 11.12, and the Seller shall cause any such Proceeding pending as of any termination of this Agreement to be dismissed with prejudice as promptly as practicable after such termination. Notwithstanding anything to the contrary in this Agreement (including this Section 9.2(e)), if the Termination Fee is paid to the Seller as liquidated damages in accordance with Section 9.2(b), under no circumstances will any Seller Related Party, or the Seller Related Parties in the aggregate, be entitled to monetary damages or monetary

remedies for any claims, damages or other Losses suffered as a result of the failure of the transactions contemplated by this Agreement to be consummated or for a breach or failure to perform hereunder or thereunder or for any representation made or alleged to have been made in connection herewith or therewith, in excess of the amount of the Termination Fee.

- (f) Without limiting the foregoing, if the Buyer fails to effect the Closing for any or no reason or otherwise breaches this Agreement or fails to perform hereunder (in any case, whether willfully, intentionally, unintentionally or otherwise), and in each case the Closing has not occurred, in no event shall the Seller seek or permit to be sought any monetary damages from any Buyer Related Party in connection with this Agreement or any of the transactions contemplated hereby, other than to the extent that the entire amount of the Termination Fee has not been paid in accordance with this Section 9.2 (provided, that in such case, the aggregate amount payable under this Agreement or in connection with the transactions contemplated hereby shall not exceed an amount equal to the portion of the Termination Fee that has not already been paid, if any).
- (g) Without limiting the foregoing, if the Seller fails to effect the Closing for any or no reason or otherwise breaches this Agreement or fails to perform hereunder (in any case, whether willfully, intentionally, unintentionally or otherwise), and in each case the Closing has not occurred, in no event shall the Buyer seek or permit to be sought any monetary damages from the Seller in connection with this Agreement or any of the transactions contemplated hereby, other than (without duplication) from the Seller to the extent provided in Section 9.2(c) (provided, that in such case, the aggregate amount payable under this Agreement or in connection with the transactions contemplated hereby shall be the actual Losses suffered by the Buyer as a consequence of such failure; provided that Buyer's recovery shall not exceed an amount equal to fifteen percent (15%) of the Unadjusted Purchase Price (in addition to the return of any portion of the Deposit from the Deposit Escrow Account that has not already been paid to the Buyer, if any)).
- (h) Notwithstanding the foregoing provisions of this <u>Section 9.2</u>, for the avoidance of doubt, nothing in this <u>Section 9.2</u> shall limit either Party's rights under Section 11.12 to specifically enforce this Agreement prior to Closing in lieu of terminating this Agreement.
- 9.3 <u>Return of Documentation and Confidentiality</u>. Upon termination of this Agreement, the Buyer shall comply with the obligation in the Confidentiality Agreement regarding the return or destruction of the "Confidential Information" (as defined therein), in each case in accordance with (and subject to the exceptions set forth in) the Confidentiality Agreement.

ARTICLE X SURVIVAL AND REMEDIES

10.1 <u>Survival</u>. None of the representations, warranties, covenants, agreements or undertakings set forth in this Agreement or in any instrument, document or certificate delivered in accordance with this Agreement shall survive the Closing, other than each covenant and agreement set forth in this Agreement that by its terms is to be performed following the Closing, which shall survive the Closing until fully performed. Absent Fraud, no Party or any of its respective Affiliates shall have any Liability with respect to any representation, warranty, covenant, agreement or undertaking from and after the time that such representation, warranty, covenant, agreement or undertaking ceases to survive hereunder (*provided*, that the foregoing shall not limit any Claim or recovery that may be available to any Buyer Related Party under

the R&W Insurance Policy). The Parties agree that, absent Fraud, (a) none of the Seller or its Affiliates shall be liable to the insurer under the R&W Insurance Policy for subrogation claims pursuant to the R&W Insurance Policy, and (b) the R&W Insurance Policy will include a waiver of such subrogation claims for the benefit of the Seller and its Affiliates.

10.2 Exclusive Remedy. From and after the Closing, except for claims of Fraud, the remedies provided or contemplated in Section 2.4, Section 2.8, Section 6.7, Section 10.1, and Section 11.12 shall be the sole and exclusive remedies (whether at law, in equity, in contract, in tort or otherwise) for any and all claims against any Party to the extent arising under, out of, related to or in connection with this Agreement. Without limiting the generality of the foregoing and subject to Article IX, this Article X, and Section 11.12, each of the Parties hereby waives, to the fullest extent permitted under applicable Law, any and all other rights, claims and causes of action (except for claims of Fraud) that it or any of its respective Affiliates may have against each of the other Parties or any of its Affiliates or its or their respective Representatives with respect to the subject matter of this Agreement, whether under any contract, misrepresentation, tort, or strict liability theory, or under applicable Law (including CERCLA and Environmental Law), and whether at law or in equity.

10.3 <u>Bold or Capitalized Letters</u>. THE PARTIES ACKNOWLEDGE THAT THE BOLD AND/OR CAPITALIZED LETTERS IN THIS AGREEMENT CONSTITUTE CONSPICUOUS LEGENDS.

10.4 Disclaimer.

(a) WITHOUT DIMINISHING THE REPRESENTATIONS AND WARRANTIES MADE BY THE SELLER IN ARTICLE III AND ARTICLE IV OR IN ANY OTHER TRANSACTION DOCUMENT, THE BUYER ACKNOWLEDGES THAT: (i) THE ASSETS HAVE BEEN USED FOR HYDROCARBON MIDSTREAM OPERATIONS AND PHYSICAL CHANGES IN THE ASSETS AND IN THE LANDS BURDENED THEREBY MAY HAVE OCCURRED AS A RESULT OF SUCH USES; (ii) THE ASSETS INCLUDE ABOVE-GROUND AND BURIED PIPELINES AND OTHER EQUIPMENT, THE LOCATIONS OF WHICH MAY NOT BE READILY APPARENT BY A PHYSICAL INSPECTION OF THE ASSETS OR THE LANDS BURDENED THEREBY; AND (iii) THE ASSETS HAVE BEEN USED FOR THE TRANSPORTATION AND PROCESSING OF OIL AND GAS AND THAT THERE MAY BE PETROLEUM, WASTES OR OTHER SUBSTANCES OR MATERIALS LOCATED IN, ON OR UNDER OR ASSOCIATED WITH THE ASSETS; NORM MAY AFFIX OR ATTACH ITSELF TO THE INSIDE OF PIPE, MATERIALS AND EQUIPMENT AS SCALE, OR IN OTHER FORMS; THE EQUIPMENT LOCATED ON THE REAL PROPERTY ASSETS OR INCLUDED IN THE ASSETS MAY CONTAIN ASBESTOS, NORM AND OTHER WASTES OR HAZARDOUS MATERIALS; NORM-CONTAINING MATERIAL OR OTHER WASTES OR HAZARDOUS MATERIALS MAY HAVE COME IN CONTACT WITH VARIOUS ENVIRONMENTAL MEDIA, INCLUDING WATER, SOILS OR SEDIMENT; AND SPECIAL PROCEDURES MAY BE REQUIRED FOR THE ASSESSMENT, REMEDIATION, REMOVAL, TRANSPORTATION OR DISPOSAL OF ENVIRONMENTAL MEDIA, WASTES, ASBESTOS, NORM AND OTHER HAZARDOUS MATERIALS FROM THE ASSETS.

(b) EXCEPT FOR THE SPECIFIC REPRESENTATIONS AND WARRANTIES EXPRESSLY MADE BY THE SELLER IN ARTICLE III AND THE SPECIFIC REPRESENTATIONS AND WARRANTIES EXPRESSLY MADE BY THE SELLER WITH RESPECT TO THE TARGET COMPANIES SET FORTH IN ARTICLE IV, (1) THE BUYER ACKNOWLEDGES AND AGREES THAT (A) NONE OF THE SELLER, THE TARGET COMPANIES OR ANY OF THEIR RESPECTIVE REPRESENTATIVES, NOR ANY OTHER PERSON, ARE MAKING OR HAS MADE, AND THE BUYER HAS NOT RELIED UPON, ANY OTHER REPRESENTATION OR WARRANTY OF ANY KIND OR NATURE WHATSOEVER, ORAL OR WRITTEN, EXPRESSED OR IMPLIED, AT LAW OR IN EQUITY, IN RESPECT OF THE TARGET COMPANIES, THE SELLER, THEIR RESPECTIVE AFFILIATES, THE INTERESTS, THE ASSETS, THIS AGREEMENT, THE TRANSACTION DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY, THE TARGET COMPANIES' RESPECTIVE BUSINESSES, LIABILITIES, OPERATIONS, PROSPECTS OR CONDITION (FINANCIAL OR OTHERWISE), INCLUDING WITH RESPECT TO MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE OF ANY ASSETS, THE NATURE OR EXTENT OF ANY LIABILITIES, INCLUDING LIABILITIES ARISING UNDER ENVIRONMENTAL LAWS (INCLUDING WITH RESPECT TO THE USE, PRESENCE, DISPOSAL OR RELEASE OF HAZARDOUS MATERIALS, ANY LIABILITIES ARISING UNDER OR WITH RESPECT TO CERCLA OR ANY OTHER ANALOGOUS LAW OR REGULATION AND ANY CHANGES IN ENVIRONMENTAL LAWS), THE BUSINESS OR FINANCIAL PROSPECTS, THE EFFECTIVENESS OR THE SUCCESS OF ANY OPERATIONS OR THE ACCURACY OR COMPLETENESS OF ANY CONFIDENTIAL INFORMATION MEMORANDA, DOCUMENTS, PROJECTIONS, MATERIAL OR OTHER INFORMATION (FINANCIAL OR OTHERWISE) REGARDING THE TARGET FURNISHED TO THE BUYER OR THE BUYER'S REPRESENTATIVES OR MADE AVAILABLE TO THE BUYER OR THE BUYER'S REPRESENTATIVES IN ANY "DATA ROOMS," "VIRTUAL DATA ROOMS," MANAGEMENT PRESENTATIONS OR IN ANY OTHER FORM IN EXPECTATION OF, OR IN CONNECTION WITH, THE TRANSACTIONS CONTEMPLATED HEREBY, OR IN RESPECT OF ANY OTHER MATTER OR THING WHATSOEVER, AND (B) NO REPRESENTATIVE OF THE SELLER OR THE TARGET COMPANIES (OR ANY OF THEIR RESPECTIVE AFFILIATES) HAS ANY AUTHORITY, EXPRESS OR IMPLIED, TO MAKE ANY REPRESENTATIONS, WARRANTIES OR AGREEMENTS NOT SPECIFICALLY SET FORTH IN THIS AGREEMENT AND SUBJECT TO THE LIMITED REMEDIES HEREIN PROVIDED; (2) THE BUYER SPECIFICALLY DISCLAIMS, ON ITS OWN BEHALF AND ON BEHALF OF ITS AFFILIATES, ANY OTHER REPRESENTATIONS OR WARRANTIES OF THE SELLER OR THE TARGET COMPANIES, WHETHER MADE BY THE SELLER, THE TARGET COMPANIES, ANY OF THEIR RESPECTIVE REPRESENTATIVES OR ANY OTHER PERSON, WITH RESPECT TO THE TARGET COMPANIES, THE SELLER, THEIR RESPECTIVE AFFILIATES, THE INTERESTS, THE ASSETS, THIS AGREEMENT, THE TRANSACTION DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY, ANY RELIANCE THEREON, OR THAT IT IS RELYING UPON OR HAS RELIED UPON ANY SUCH OTHER REPRESENTATIONS OR WARRANTIES THAT MAY HAVE BEEN MADE BY ANY PERSON, AND ACKNOWLEDGES AND AGREES THAT THE SELLER AND THE TARGET COMPANIES HAVE SPECIFICALLY DISCLAIMED AND DO SPECIFICALLY DISCLAIM ANY SUCH OTHER REPRESENTATION OR WARRANTY

MADE BY ANY PERSON; (3) THE BUYER SPECIFICALLY DISCLAIMS ANY OBLIGATION OR DUTY BY THE SELLER OR THE TARGET COMPANIES TO MAKE ANY DISCLOSURES OF FACT NOT REQUIRED TO BE DISCLOSED PURSUANT TO THE SPECIFIC REPRESENTATIONS AND WARRANTIES SET FORTH IN <u>ARTICLE III</u> AND <u>ARTICLE IV</u> OF THIS AGREEMENT AND THE BUYER HAS NOT RELIED UPON THE ABSENCE OF A DISCLOSURE OF ANY SPECIFIC FACT; AND (4) THE BUYER IS ACQUIRING THE INTERESTS AND THE TARGET COMPANIES SUBJECT ONLY TO THE SPECIFIC REPRESENTATIONS AND WARRANTIES SET FORTH IN <u>ARTICLE III</u> AND <u>ARTICLE IV</u> OF THIS AGREEMENT AS FURTHER LIMITED BY THE SPECIFICALLY BARGAINED-FOR EXCLUSIVE REMEDIES AS SET FORTH IN THIS ARTICLE X.

ARTICLE XI MISCELLANEOUS PROVISIONS

- 11.1 <u>Assignment</u>. No Party may assign this Agreement or any of its rights or obligations arising hereunder without the prior written consent of the other Party; *provided*, *however*, that without the consent of the Seller, the Buyer may, without relieving the Buyer from its Liabilities or obligations hereunder, assign this Agreement, and its rights and obligations hereunder, to an Affiliate of Buyer or to an entity formed, controlled and primarily owned by Buyer. Subject to the preceding sentence of this <u>Section 11.1</u>, this Agreement shall be binding upon, inure to the benefit of, and be enforceable by, the Parties and their respective successors and permitted assigns. Any purported assignment not permitted under this <u>Section 11.1</u> shall be null and void.
- 11.2 Amendments and Waiver. This Agreement may be amended, superseded or canceled only by a written instrument duly executed by each Party, specifically stating that it amends, supersedes or cancels this Agreement. Any of the terms of this Agreement and any condition to a Party's obligations hereunder may be waived only in writing by that Party specifically stating that it waives a term or condition hereof. No waiver by a Party of any one or more conditions or defaults by the other in performance of any of the provisions of this Agreement shall operate or be construed as a waiver of any future conditions or defaults, whether of a like or different character, nor shall the waiver constitute a continuing waiver unless otherwise expressly provided in writing. Notwithstanding anything to the contrary contained in this Agreement, Section 9.2(e), Section 9.2(f), this Section 11.2, Section 11.11, Section 11.14(b) and the definitions of "Debt Financing Sources" and "Debt Finance Related Parties" (or to any other provision or definition of this Agreement to the extent that such amendment or waiver would modify the substance of any such foregoing Section) may not be amended, modified, supplemented, waived or terminated in a manner that is materially adverse to the Debt Financing Sources without the prior written consent of the Debt Financing Sources party to the Debt Commitment Letter.
- 11.3 Entire Agreement. This Agreement (together with any Exhibits and Schedules hereto), the Transaction Documents, and all other certificates, documents, instruments, and writings that are delivered pursuant to this Agreement contain the entire understanding of the Parties with respect to the transactions contemplated hereby and supersede all prior agreements, arrangements, and understandings relating to the subject matter hereof other than the Confidentiality Agreement, which is hereby ratified by the Parties.

- 11.4 Severability. If any term or other provision of this Agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other terms, provisions and conditions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby are not affected in any manner materially adverse to either Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible to the fullest extent permitted by applicable law in an acceptable manner to the end that the transactions contemplated by this Agreement are fulfilled to the extent possible.
- 11.5 <u>Counterparts</u>. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall together constitute one and the same instrument. This Agreement may be executed and delivered by email in portable document format (.pdf), and delivery of the executed signature page by such method will be deemed to have the same effect as if the original signature had been delivered to the other Parties.

11.6 Governing Law and Dispute Resolution.

- (a) <u>Governing Law</u>. THIS AGREEMENT AND THE PERFORMANCE OF THE TRANSACTIONS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT WILL BE GOVERNED BY AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH THE LAW OF THE STATE OF DELAWARE, WITHOUT REGARD TO THE CONFLICTS OF LAW PRINCIPLES OF SUCH STATE; PROVIDED, HOWEVER, THAT NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY ALL CLAIMS OR CAUSES OF ACTION (WHETHER AT LAW, IN EQUITY, IN CONTRACT, IN TORT OR OTHERWISE) AGAINST ANY DEBT FINANCING SOURCE IN ANY WAY RELATING TO THIS AGREEMENT AND ANY RIGHT OR OBLIGATION WITH RESPECT TO ANY DEBT FINANCING SOURCE IN CONNECTION WITH THE DEBT FINANCING OR THE DEBT COMMITMENT LETTER SHALL BE INTERPRETED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO PRINCIPLES OR RULES OR CONFLICT OF LAWS TO THE EXTENT SUCH PRINCIPLES OR RULES WOULD REQUIRE OR PERMIT THE APPLICATION OF LAWS OF ANOTHER JURISDICTION.
- (b) <u>Consent to Jurisdiction</u>. Subject to clause (e) of this <u>Section 11.6</u>, the Parties hereto irrevocably submit to the exclusive jurisdiction of the Delaware Chancery Courts located in Wilmington, Delaware, or, if such court shall not have jurisdiction, any federal court of the United States or other Delaware state court located in Wilmington, Delaware, and appropriate appellate courts therefrom, over any dispute arising out of or relating to this Agreement or any of the transactions contemplated by this Agreement, and each Party irrevocably agrees that all claims in respect of such dispute may be heard and determined in such courts. Subject to clause (e) of this <u>Section 11.6</u>, the Parties irrevocably waive, to the fullest extent permitted by applicable Law, any objection which they may now or hereafter have to the laying of venue of any dispute arising out of or relating to this Agreement or any of the transactions contemplated by this Agreement brought in such courts or any defense of inconvenient forum for the maintenance of such dispute. Each of the Parties hereto agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. Each of the Parties hereto consents to process being served by any Party to this Agreement in any suit, action, Proceeding or counterclaim of the nature specified in this <u>Section 11.6(b)</u> in the manner specified by the provisions of <u>Section 11.7</u>.

- (c) Waiver of Jury Trial. NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY, EACH OF THE PARTIES ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY THAT MAY ARISE UNDER THIS AGREEMENT, THE DEBT FINANCING THAT THE BUYER PROCURES IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY, THE DEBT COMMITMENT LETTER OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY OR THE PERFORMANCE OF ANY SERVICES THEREUNDER IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH OF THE PARTIES KNOWINGLY AND INTENTIONALLY, IRREVOCABLY AND UNCONDITIONALLY WAIVES TRIAL BY JURY AGAINST THE DEBT FINANCING SOURCES DIRECTLY OR INDIRECTLY IN ANY PROCEEDING, SUIT OR ACTION UNDER, ARISING OUT OF, OR IN CONNECTION WITH THIS AGREEMENT, THE DEBT FINANCING THAT THE BUYER PROCURES IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY, THE DEBT COMMITMENT LETTER OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY OR THE PERFORMANCE OF ANY SERVICES THEREUNDER AND FOR ANY COUNTERCLAIM.
- (d) <u>Settlement Proceedings</u>. All aspects of any settlement Proceedings, including discovery, testimony and other evidence, briefs, negotiations, communications and any award pursuant to this <u>Section 11.6</u> shall be held confidential by each Party and shall be treated as compromise and settlement negotiations for the purposes of the federal and state rules of evidence.
- (e) <u>Financing Sources</u>. Notwithstanding anything in this Agreement to the contrary, (i) each of the Parties hereto agrees that it will not bring or support (or permit any of their controlled Affiliates to bring or support) any claim or cause of action (whether in contract, tort or otherwise), against any Debt Financing Source, in any way relating to this Agreement or any of the transactions contemplated by this Agreement, including any dispute arising out of or relating in any way to the Debt Commitment Letter, the Debt Financing or the Definitive Agreements executed in connection therewith or the performance thereof or services provided thereunder, in any forum other than any federal court in the Borough of Manhattan, New York, New York and any appellate court thereof, and each Party waives any objection which such Party may now or hereafter have to the laying of the venue of any such action, suit or proceeding, and irrevocably submits to the jurisdiction of any such court in any such action, suit or proceeding and (ii) under no circumstances shall any Party to this Agreement be entitled to recovery from any Debt Financing Source consequential, indirect, punitive, exemplary or special damages arising out of or relating to the transactions contemplated by this Agreement, the Debt Commitment Letter or the Debt Financing.
- 11.7 <u>Notices and Addresses</u>. Any notice, request, instruction, waiver, or other communication to be given hereunder by any Party shall be in writing and shall be considered duly delivered if personally delivered, mailed by certified mail with the postage prepaid (return receipt requested), sent by messenger or overnight delivery service, or sent by e-mail (without notice of failed delivery) to the addresses of the Parties as follows:

If to the Buyer or, after the Closing, the Target Companies:

Rheinbund, LLC

320 S. Boston Ave., Suite 900

Tulsa, Oklahoma 74103 Attention: Tim Helms

E-mail: Tim@ce2ok.com

with a copy to (which shall not constitute notice):

Shearman & Sterling LLP 2828 North Harwood Street

18th Floor

Dallas, Texas 75201

Attention: Nathan D. Meredith

E-mail: nathan.meredith@shearman.com

If to Parent:

Citizen Energy Operating, LLC 320 S. Boston Ave., Suite 900 Tulsa, Oklahoma 74103 Attention: Tim Helms E-mail: Tim@ce2ok.com

with a copy to (which shall not constitute notice):

Shearman & Sterling LLP 2828 North Harwood Street 18th Floor Dallas, Texas 75201

Attention: Nathan D. Meredith

E-mail: nathan.meredith@shearman.com

If to the Seller or, prior to the Closing, the Target Companies:

Linn Energy Holdco II LLC c/o Riviera Resources, Inc. 717 Texas St., Suite 2000 Houston, TX 77002

David B. Rottino, Chief Executive Officer Attention:

Holly Anderson, General Counsel

drottino@rvraresources.com Email:

> handerson@rvraresources.com legal@rvraresources.com

with a copy to (which shall not constitute notice):

Kirkland & Ellis LLP 609 Main Street 45th Floor

Houston, Texas 77002

Attention: Julian J. Seiguer, P.C.

Kim Hicks, P.C.

E-mail: julian.seiguer@kirkland.com

kim.hicks@kirkland.com

or at such other address as a Party may designate by written notice to the other Party in the manner provided in this Section 11.7. Notice shall be deemed given on the date of actual delivery. If a date specified herein for giving any notice or taking any action is not a Business Day (or if the period during which any notice is required to be given or any action expires on a date which is not a Business Day), then the date for giving such notice or taking such action (and the expiration date of such period during which notice is required to be given or action taken) shall be the next day which is a Business Day.

11.8 Conflict Wavier; Attorney-Client Privilege.

- (a) It is acknowledged by each of the Parties that Kirkland & Ellis LLP ("Seller Counsel") represented the Seller and the Target Companies in connection with the negotiation of this Agreement and the Transaction Documents. The Buyer agrees, on its own behalf and, after the Closing, on behalf of the Target Companies, that, following consummation of the transactions contemplated under this Agreement, such representation and any prior representation of the Target Companies by Seller Counsel shall not preclude Seller Counsel from serving as counsel to the Seller or any director, member, shareholder, partner, officer or employee of the Seller, in connection with any litigation, Claim or obligation arising out of or relating to this Agreement or the transactions contemplated by this Agreement.
- (b) The Buyer shall not, and after the Closing, shall cause the Target Companies not to, seek or have Seller Counsel disqualified from any such representation based upon the prior representation of the Target Companies by Seller Counsel prior to Closing. Each of the Parties consents thereto and waives any conflict of interest arising from such prior representation, and each of such Parties shall cause any of its Affiliates to consent to waive any conflict of interest arising from such representation. The Parties acknowledge and agree that the waiver and consent set forth in the preceding sentence only applies to the Seller and not to any Third Person. Each of the Parties acknowledges that such consent and waiver is voluntary, that it has been carefully considered, and that the Parties have consulted with counsel or have been advised they should do so in connection herewith. The covenants, consent and waiver contained in this Section 11.8 shall not be deemed exclusive of any other rights to which Seller Counsel is entitled whether pursuant to law, contract or otherwise.
- (c) The Buyer and the Seller agree that any attorney-client privilege, attorney work-product protection, and expectation of client confidence attaching as a result of Seller Counsel's representation of the Target Companies in connection with the negotiation of this Agreement and the Transaction Documents and in connection with the transactions contemplated hereby and thereby, and all information and documents covered by such privilege or protection, shall belong to and be controlled by the Seller and may be waived only by the Seller, and not by any Target Company, and shall not pass to or be claimed or used by the Buyer or the Target Companies, except as provided in Section 11.8(d).

(d) In the event that a dispute arises between the Buyer or the Target Companies, on the one hand, and a Third Person other than the Seller or any of its Affiliates, on the other hand, the Target Companies may assert the attorney-client privilege on behalf of the Seller to the extent necessary to prevent disclosure of privileged materials described in <u>Section 11.8(c)</u> to such Third Person; *provided*, *however*, that such privilege may be waived only with the prior written consent of the Seller (such consent not to be unreasonably withheld, conditioned or delayed).

11.9 Rules of Construction; Joint Drafting.

- (a) In construing this Agreement, the following principles will be followed, in each case unless expressly provided otherwise in a particular instance: (i) the singular includes the plural and vice versa; (ii) reference to a Person includes such Person's successors and assigns but, in the case of a Party, only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity; (iii) reference to any gender includes each other gender; (iv) references to any Exhibit, Schedule, Section, Article, Annex, subsection and other subdivision refer to the corresponding Exhibits, Schedules, Sections, Articles, Annexes, subsections and other subdivisions of this Agreement, unless expressly provided otherwise; (v) references in any Section, Article or definition to any clause means such clause of such Section, Article or definition; (vi) "hereunder," "hereof," "hereby" and words of similar import are references to this Agreement as a whole and not to any particular provision of this Agreement; (vii) the word "or" is not exclusive, and the word "including" (in its various forms) means "including without limitation"; (viii) each accounting term not otherwise defined in this Agreement has the meaning commonly applied to it in accordance with GAAP; (ix) references to "days" are to calendar days, unless the term "Business Days" is used, and whenever any action must be taken hereunder on or by a day that is not a Business Day, then such action may be validly taken on or by the next day that is a Business Day; (x) all references to money refer to the lawful currency of the United States; (xi) references to the "other Party" from the perspective of the Buyer refers to the Seller, and from the perspective of the Seller refer the Buyer; (xii) any reference to any federal, state, local or foreign Law shall be deemed also to refer to all rules and Regulations promulgated under such Law; (xiii) all monetary figures shall be in U.S. dollars unless otherwise specified, (xiv) the table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement and (xv) the phrase "provided", "made available", "delivered" or words of similar import shall mean that the documents or information referred to have been posted no later than three (3) Business Days prior to the Execution Date, to the electronic data site hosted by Donnelley Financial Solutions and established by the Seller and its Representatives for the purpose of providing due diligence materials and information to the Buyer and its respective Representatives, or otherwise delivered to the Seller.
- (b) The Parties have participated jointly in the negotiation and drafting of this Agreement with the assistance of counsel. Consequently, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement or interim drafts of this Agreement.

- 11.10 Offset. Nothing contained herein shall impair or constitute a waiver of any right of offset or setoff for any Party.
- 11.11 No Partnership; Third-Party Beneficiaries. Nothing in this Agreement shall be deemed to create a joint venture, partnership, Tax partnership, or agency relationship between or among the Parties. This Agreement shall not confer upon any Person other than the Parties any rights (including third-party beneficiary rights or otherwise) or remedies under this Agreement, except for (i) the provisions of Section 6.2(d) (Access Indemnification), Section 6.8 (Continuation of Indemnity; D&O Tail Coverage) Section 9.2 (Effect of Termination), Article X (Survival and Remedies), this Section 11.11 (No Partnership; No Third-Party Beneficiary), and Section 11.15 (Release); and (ii) notwithstanding anything herein to the contrary, the Debt Financing Sources shall be express third party beneficiaries of Section 11.2 (Amendments and Waiver) (as it relates to the Debt Financing Sources), Section 11.6(e), Section 11.6(e), this Section 11.11 (No Partnership; No Third-Party Beneficiary) (as it relates to the Debt Financing Sources), and Section 11.14(b), and each of such Sections shall expressly inure to the benefit of the Debt Financing Sources and the Debt Financing Sources shall be entitled to rely on and enforce the provisions of such Sections.
- 11.12 <u>Specific Performance</u>. Each Party agrees that irreparable damage would occur and that the Parties would not have any adequate remedy at Law in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Accordingly, in addition to any other remedies available under this Agreement, the Parties agree that, prior to the termination of this Agreement in accordance with <u>Section 9.1</u>, each Party will be entitled to an injunction or injunctions, specific performance and other equitable relief to prevent the other Party's breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement. Each Party agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief when expressly available pursuant to the terms of this Agreement, and hereby waives (a) any defenses in any Proceeding for an injunction, specific performance or other equitable relief, including the defense that the other Party has an adequate remedy at Law or an award of specific performance is not an appropriate remedy for any reason at Law or equity and (b) any requirement under Law to post a bond, undertaking or other security as a prerequisite to obtaining equitable relief. The Parties acknowledge and agree that if the Buyer or the Seller exercises its right to terminate this Agreement under <u>Section 9.1</u>, then such Person shall not thereafter have the right to specific performance under this <u>Section 11.12</u>.
- 11.13 <u>Disclosure Schedule</u>. The Disclosure Schedules to this Agreement are arranged in sections corresponding to those contained in this Agreement merely for convenience, and the disclosure of any fact or item in one section or subsection of such Disclosure Schedules to any particular covenant, representation or warranty shall be deemed adequately disclosed with respect to all other covenants, representations or warranties, notwithstanding the presence or absence of an appropriate section or subsection of such Disclosure Schedules with respect to such other covenants, representations or warranties or an appropriate cross-reference thereto, in each case to the extent relevancy of such disclosure to such other covenants, representations or warranties is reasonably apparent on the face of such disclosure that such disclosed information is applicable thereto. Additionally, for each of the Disclosure Schedules, the mere inclusion of an item in such Disclosure Schedules as an exception to a representation or warranty shall not be deemed an admission or acknowledgment, in and of itself and solely by virtue of the inclusion

of such information in such Disclosure Schedules, that such information is required to be listed in such Disclosure Schedules or that such item (or any non-disclosed item or information of comparable or greater significance) represents a material exception or fact, event or circumstance, that such item has had, or is expected to result in, a Target Material Adverse Effect, that such item actually constitutes noncompliance with, or a violation of, any Law, Permit or Contract or other topic to which such disclosure is applicable or that such item is outside the Ordinary Course of Business. The specification of any dollar amount in the representations and warranties contained in this Agreement or the inclusion of any specific item in the Disclosure Schedules is not intended to imply that such amounts (or higher or lower amounts) are or are not material, and no Party shall use the fact of the setting of such amounts or the fact of the inclusion of any such item in the Disclosure Schedules in any dispute or controversy between the Parties as to whether any obligation, item, or matter not described in this Agreement or included in a Disclosure Schedule is or is not material for purposes of this Agreement. Capitalized terms used in the Disclosure Schedules, unless otherwise defined therein, shall have the meanings assigned to them in this Agreement.

11.14 Non-Recourse.

(a) All causes of action or Proceedings (whether in contract or in tort, in equity or at Law, or granted by statute) that may be based upon, in respect of, arise under, out or by reason of, be connected with, or relate in any manner to this Agreement, or the negotiation, preparation, execution, delivery, performance or breach of this Agreement (including any representation or warranty made in, in connection with, or as an inducement to, this Agreement), may be brought only against (and are those solely of) the Seller Parties and the Buyer Parties (each, a "Recourse Party"). No Person who is not a Recourse Party, including any past, present or future direct or indirect equity holder, Affiliate (other than Subsidiaries of such Recourse Party) or Representative of such Recourse Party, such equity holder or such Affiliate (each, a "Non-Recourse Party"), shall have any Liability or other obligation (whether in contract or in tort, in equity or at Law, or granted by statute) for any cause of action or Proceeding arising under, out of, in connection with, or related in any manner to this Agreement or based on, in respect of, or by reason of this Agreement or its negotiation, preparation, execution, delivery, performance, or breach; and, to the maximum extent permitted by applicable Law, each Party, on behalf of itself and, in the case of the Seller, the other Seller Parties, and in the case of the Buyer, the other Buyer Parties, hereby waives and releases all such causes of action and Proceedings against any such Non-Recourse Party. Without limiting the generality of the foregoing, to the maximum extent permitted by applicable Law, (a) each Party, on behalf of itself and, in the case of the Seller, the other Seller Parties, and in the case of the Buyer, the other Buyer Parties, hereby waives and releases any and all causes of action or Proceedings that may otherwise be brought in equity or at Law, or granted by statute, to avoid or disregard the entity form of a Recourse Party or otherwise impose Liability or other obligation of any Recourse Party on any Non-Recourse Party, whether granted by statute or based on theories of equity, agency, control, instrumentality, alter ego, domination, sham, single business enterprise, piercing the veil, unfairness, undercapitalization, or otherwise; and (b) each Party, on behalf of itself and, in the case of the Seller, the other Seller Parties, and in the case of the Buyer, the other Buyer Parties, disclaims any reliance upon any Non-Recourse Party with respect to the performance of this Agreement or any representation or warranty made in, in connection with, or as an inducement to this Agreement. Notwithstanding the foregoing, nothing in this Section 11.14 shall preclude any party to the Escrow Agreement, the Confidentiality Agreement or any other Transaction Document from making any claim thereunder, to the extent permitted therein and pursuant to the terms thereof (and subject to the applicable limitations set forth therein).

(b) Notwithstanding anything herein to the contrary, Seller Parties shall not have any claim against any Debt Financing Source (solely in its capacity as a Debt Financing Source) (or any of its respective former, current or future general or limited partners, stockholders, managers, members, agents, Representatives, Affiliates, successors or assigns (collectively, "Debt Finance Related Parties")) nor shall any Debt Financing Source (solely in its capacity as a Debt Financing Source) or any Debt Finance Related Parties (solely in its capacity as a Debt Finance Related Party) have any liability whatsoever to the Seller Parties, in connection with the Debt Financing, the Debt Commitment Letter or in any way relating to this Agreement or any of the transactions or services contemplated thereby or hereby, whether at law, in equity, in contract, in tort or otherwise.

11.15 Release.

- (a) Effective as of the Closing, the Buyer, on its own behalf and on behalf of its direct and indirect equity holders, Affiliates (including the Target Companies following the Closing) and Representatives, and its and their respective Affiliates and Representatives, and each of the respective heirs, executors, administrators, successors and permitted assigns of each of the foregoing (each, a "Buyer Releasing Person"), hereby absolutely and unconditionally releases and forever discharges each of the Seller, its past, present and future direct and indirect equity holders, Affiliates and Representatives, and each of its and their respective Affiliates and Representatives, and each of the respective heirs, executors, administrators, successors and permitted assigns of each of the foregoing (each, a "Seller Released Person") from, and agrees not to assert any cause of action or Proceeding with respect to, any Losses or Liabilities whatsoever, of any kind or nature, whether at law or in equity ("Released Claims"), which have been or could have been asserted against any Seller Released Person, which any Buyer Releasing Person has or ever had, which arises out of or in any way relates to events, circumstances or actions occurring, existing or taken prior to or as of the Closing Date in respect of matters relating to the Seller's ownership of the Interests or ownership or operation of the Target Companies, or their respective businesses, or the negotiation of this Agreement. Nothing contained in this Section 11.15(a) is intended to, nor does it, limit, impair or otherwise modify or affect any rights or obligations of the Seller Released Persons expressly set forth in this Agreement, the other Transaction Documents or the Contribution Agreement, or any facts, circumstances or claims to the extent entitling a Buyer Releasing Person to any recovery under this Agreement, the other Transaction Documents or the Contribution Agreement.
- (b) Effective as of the Closing, the Seller, on its own behalf and on behalf of its direct and indirect equity holders, Affiliates (including the Target Companies prior to the Closing) and Representatives, and its and their respective Affiliates and Representatives, and each of the respective heirs, executors, administrators, successors and permitted assigns of each of the foregoing (each, a "Seller Releasing Person"), hereby absolutely and unconditionally releases and forever discharges each of the Buyer, its past, present and future direct and indirect equity holders, Affiliates and Representatives, and each of its and their respective Affiliates and Representatives, and each of the respective heirs, executors, administrators, successors and permitted assigns of each of the foregoing (each, a "Buyer Released Person") from, and agrees not to assert any Released Claim, which have been or could have been asserted against any Buyer Released Person, which any Seller Releasing Person has or ever had, which arises out of or in any way relates to events, circumstances or actions occurring, existing or taken after or as of the Closing Date in respect of matters relating to the Buyer's ownership of the Interests or ownership or operation of

the Target Companies, or their respective businesses. Nothing contained in this <u>Section 11.15(b)</u> is intended to, nor does it, limit, impair or otherwise modify or affect any rights or obligations of the Buyer Released Persons expressly set forth in this Agreement or the other Transaction Documents, or any facts, circumstances or claims to the extent entitling a Seller Releasing Person to any recovery under this Agreement or the other Transaction Documents.

11.16 Parent Guaranty.

- (a) For good and valuable consideration, and to induce Seller to enter into this Agreement, Parent absolutely, unconditionally and irrevocably guarantees to the Seller the punctual and complete performance of all obligations of the Buyer under this Agreement, including the payment obligations of the Buyer pursuant to Section 2.2, Section 2.3, Section 2.4 and Section 9.2 (and as also referenced in Section 8.2(b) (the "Buyer Obligations")). The limited guaranty set out in this Section 11.16 (the "Parent Guaranty") shall remain in full force and effect until the earlier of (i) the performance by the Buyer or Parent to the Seller of the Buyer Obligations and (ii) termination of this Agreement in accordance with the terms of Article IX.
- (b) Except as set forth in this Section 11.16 but in all instances subject to the limitations set forth in Section 10.2 of this Agreement, Parent's liability under this Section 11.16 shall be absolute, unconditional, irrevocable and continuing irrespective, without limitation, of (i) any lack of validity or enforceability of this Agreement as a result of the application of any bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity to the Buyer; (ii) any modification, amendment, consent, extension, forbearance or waiver of or any consent to departure from this Agreement that may be agreed to by the Parties in accordance with the terms of this Agreement; (iii) any action or inaction by the Seller under or in respect of this Agreement, any failure, lack of diligence, omission or delay on the part of the Seller to enforce, assert or exercise any right, power or remedy conferred on the Seller in this Agreement; (iv) any merger or consolidation of the Buyer, Parent or any of their respective Affiliates into or with any Person, or any sale, lease or transfer of any of the assets of Parent, the Buyer, Seller or any other Person to any other Person; or (v) any change in the ownership of any of Parent, the Buyer, Seller or any other Person.
- (c) Subject to the limitations set forth in Section 10.2 of this Agreement, upon default by the Buyer, or failure by the Buyer to perform of any of the Buyer Obligations, Parent shall be liable for the obligations of the Buyer arising under this Agreement as if it were a primary obligor, and Seller may immediately proceed directly against Parent without first proceeding against the Buyer or any other Person or pursuing any other remedy.
- (d) Parent waives any defenses which it may have with respect to the performance of the Buyer Obligations, other than defenses that the Buyer would have under the terms of this Agreement. Parent further waives notice of the acceptance of this Parent Guaranty, presentment, demand, protest, and notices of protest, nonpayment, default or dishonor of the Buyer Obligations. The other provisions of this Section 11.16(d) shall be deemed incorporated and to apply to this Section 11.16(d), and Parent shall be a "Party" for purposes of Article XI and Section 10.4.

- (e) Parent represents and warrants to the Seller the following:
- (i) Parent is a limited liability company duly organized, validly existing and in good standing under the laws of Delaware. Parent has all requisite limited liability company power and authority to execute and deliver this Parent Guaranty and to perform its obligations under this Parent Guaranty.
- (ii) The execution, delivery and performance of this Parent Guaranty, and the consummation of the transactions contemplated by this Parent Guaranty, have been duly and validly authorized by all necessary action of Parent or any of its Affiliates, as applicable.
- (iii) This Parent Guaranty has been duly executed and delivered by Parent and constitutes, assuming the due authorization, execution and delivery of this Parent Guaranty, as applicable, by the other Persons that are party thereto, the valid and binding obligations of Parent, as applicable, enforceable against Parent, as applicable, in accordance with their respective terms, except as enforceability may be limited by bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium or other Laws relating to or affecting creditors' rights generally or by equitable principles (regardless of whether enforcement is sought at law or in equity).
- (iv) The execution, delivery and performance of this Parent Guaranty, and the consummation of the transactions contemplated by this Parent Guaranty do not violate, breach or contravene (a) Parent's Organizational Documents or (b) any Law or contractual restriction binding on or affecting such Parent or its properties except where such violation, breach or contravention, individually or in the aggregate, could not reasonably be expected to have a material adverse effect on such Parent's ability to perform its obligations under this Parent Guaranty.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the Parties have and solely for purposes of <u>Section 11.16</u>, Parent has, caused this Agreement to be duly executed and delivered as of the date first above written.

BUYER:

RHEINBUND, LLC

By: /s/ Tim Helms
Name: Tim Helms

Title: Chief Financial Officer

SIGNATURE PAGE
MEMBERSHIP INTEREST PURCHASE AGREEMENT

PARENT, SOLELY FOR PURPOSES OF <u>SECTION 11.16</u>:

CITIZEN ENERGY OPERATING, LLC

By: <u>/s/ Tim Helms</u>

Name: Tim Helms

Title: Chief Financial Officer

SIGNATURE PAGE
MEMBERSHIP INTEREST PURCHASE AGREEMENT

SELLER:

LINN ENERGY HOLDCO II LLC

By: /s/ Darren Schluter

Name: Darren Schluter

Title: Executive Vice President, Finance, Administration and Chief Accounting

Officer

SIGNATURE PAGE
MEMBERSHIP INTEREST PURCHASE AGREEMENT

List of Exhibits and Schedules Omitted from the Membership Interest Purchase Agreement Referenced in Exhibit 2.1 Above

Pursuant to Regulation S-K, Item 601(b)(2), the Exhibits and Schedules to the Membership Interest Purchase Agreement referenced in Exhibit 2.1 above, as listed below, have not been filed. The Registrant agrees to furnish supplementally a copy of any omitted Exhibit or Schedule to the Securities and Exchange Commission (the "Commission") upon request; provided, however, that the Registrant may request confidential treatment of omitted items.

F.X	hı	bits	

Exhibit A Target Assignment Agreement
Exhibit B: Transition Services Agreement
Exhibit C: R&W Insurance Policy
Exhibit D: Estimated Closing Statement
Exhibit E: Escrow Agreement

Schedules

Schedule 1.1 (a) CapEx Budget

Schedule 1.1 (b) Illustrative Working Capital Schedule

Schedule 1.1 (c) Payoff Indebtedness

Schedule 1.1 (d) Liens

Schedule 1.1 (e) Post-Closing Notification

Schedule 1.1 (f)

Gathering and Processing Systems

Schedule 1.1 (g) SWD Wells

Schedule 1.1 (h)

Schedule 1.1 (i)

Reimbursable Expenses

Schedule 1.1 (j)

Designated Agreements

Schedule 1.1 (k) Current Tax Assets and Liabilities

Schedule 6.3 Customers

Schedule 6.18 Termination and Release of Claims

Disclosure Schedules

Disclosure Schedule 3.3(a) Consents (Seller)
Disclosure Schedule 3.3(b) No Conflicts (Seller)

Disclosure Schedule 3.4 Litigation

Disclosure Schedule 3.5 Broker's or Finder's Fees (Seller)

Disclosure Schedule 4.1(b) Equity Interests (Target)
Disclosure Schedule 4.1(c) Target Companies

Disclosure Schedule 4.2 No Conflicts (Target Companies)

Disclosure Schedule 4.3(b) Required Notifications

Disclosure Schedule 4.3(c) Required Third-Party Consents

Disclosure Schedule 4.3(d)

Disclosure Schedule 4.4

Disclosure Schedule 4.5(1)

Tax liability

Disclosure Schedule 4.6 Compliance with Laws
Disclosure Schedule 4.7(a) Material Contracts

Disclosure Schedule 4.7(b) Material Contract Exceptions

Disclosure Schedule 4.8

Disclosure Schedule 4.9

Disclosure Schedule 4.10(a)

Disclosure Schedule 4.10(c)

Target Company Benefit Plan

Disclosure Schedule 4.10(e)

Disclosure Schedule 4.10(f)

Disclosure Schedule 4.11

Disclosure Schedule 4.12(a)

Disclosure Schedule 4.12(d)

Disclosure Schedule 4.12(d)

Material Payments

Employee Matters

Financial Statements

Disclosure Schedule 4.12(d)

Disclosure Schedule 4.13 Environmental Matters

Disclosure Schedule 4.14(a)

Disclosure Schedule 4.14(b)

Disclosure Schedule 4.14(c)

Disclosure Schedule 4.14(c)

Disclosure Schedule 4.14(c)

Disclosure Schedule 4.14(c) Disputes

Disclosure Schedule 4.15(a)(i)

Owned Real Property
Disclosure Schedule 4.15(a)(ii)

Leased Real Property

Disclosure Schedule 4.15(e) Personal Property Leases Exceptions

Disclosure Schedule 4.15(f) Assets

Disclosure Schedule 4.16 Affiliate Transactions
Disclosure Schedule 4.17 Absence of Certain Changes

Disclosure Schedule 4.18 Insurance Disclosure Schedule 4.19 Bank Accounts Disclosure Schedule 4.24 System Data Disclosure Schedule 4.25 Imbalances Disclosure Schedule 4.26 Regulatory Status SWD Wells (Capacity) Disclosure Schedule 4.29(a) Disclosure Schedule 4.29(b) SWD Wells (Integrity) Disclosure Schedule 6.1(a) Conduct of Business Disclosure Schedule 6.10(f) **Buyer Plans**

Disclosure Schedule 6.17(a)

Disclosure Schedule 6.17(d)

Disclosure Schedule 7.3(c)

Intercompany Guarantees

Required Intercompany Bonds

Required Third-Party Consent

Schedule A Defined Terms

- "Accounting Firm" is defined in Section 2.4(d).
- "Accounting Principles" means GAAP consistently applied with the principles, practices, methodologies and procedures used by the Seller in preparation of the Financial Statements; *provided*, that, with respect to any calculation of Working Capital, "Accounting Principles" shall mean the principles, practices, methodologies and procedures used by in the preparation of the Illustrative Working Capital Schedule.
- "Accrued Reimbursable Expenses Amount" means the aggregate amount of Reimbursable Expenses accrued but not yet paid by any of the Target Companies to the Seller or any of its Affiliates (other than the Target Companies) as of the Closing.
 - "Acquisition Proposal" is defined in Section 6.15.
 - "Adjusted Purchase Price" is defined in Section 2.3.
- "Adjustment Amount" means an amount of U.S. dollars (expressed as a positive or a negative number, as applicable) equal to the sum of (without duplication) the following determined as of the Measurement Time: (i) the Cash and Cash Equivalents, <u>plus</u> (ii) the Working Capital Surplus, if any, <u>minus</u> (iii) the Working Capital Deficit, if any, <u>minus</u> (iv) the Indebtedness of the Target Companies (including, for the avoidance of doubt, any Payoff Amounts), <u>minus</u> (v) the Company Transaction Cost Amount, <u>plus</u> (vi) the Accrued Reimbursable Expenses Amount.
 - "Adjustment Notice" is defined in Section 2.4(b).
- "Affiliate" means, when used with respect to a specified Person, any other Person directly or indirectly (through one or more intermediaries or otherwise) controlling, controlled by, or under common control with the specified Person; provided, however, that none of Sponsor, any fund formed or managed by such Person or its Affiliates for the purposes of making investments, or any portfolio company of any such funds shall be considered an Affiliate of the Buyer for purposes of this Agreement other than for purposes of Section 10.1 and Section 10.2. For purposes of this definition, "control," when used with respect to any specified Person, means the power to direct or cause the direction of the management and policies of such Person whether through the ownership of voting securities, by contract, or otherwise; and the term "controlled" has the meanings correlative to the foregoing. For the purposes of this Agreement, the Target Companies will be deemed to be Affiliates of the Seller prior to the Closing and will be deemed to be Affiliates of the Buyer after the Closing.
 - "Agreement" is defined in the Preamble.
 - "Allocation Schedule" is defined in Section 2.5(b).
- "Anti-Money Laundering Legislation" means the Currency and Foreign Transactions Reporting Act of 1970, as amended, the Bank Secrecy Act, as amended by Title III of the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001 (USA PATRIOT ACT), the United States Department of the Treasury Office of Foreign Assets Control regulations and the money laundering laws of all applicable jurisdictions where the Target Companies conduct business.

- "Assets" means the Target Companies' assets and properties, including the Target Company Systems.
- "*Authorization*" means any franchise, Permit, license, authorization, Order, certificate, registration or other consent or approval that a Governmental Authority has the legal authority to grant or issue.
 - "Available Employee List" is defined in Section 4.11(a).
- "Available Employees" means those employees of the Seller or its Affiliates (i) who spend substantially all of their business time providing services relating to the Business and (ii) to whom the Buyer or its Affiliates may, but shall not be obligated to, make an offer of employment.
 - "Balance Sheets" is defined in Section 4.12(a)(i).
 - "Balance Sheet Date" is defined in Section 4.12(a)(ii).
 - "Binder Agreement" is defined in Section 6.9(a).
- "Business" means the business of the Target Companies as of the Execution Date, including the oil, natural gas and produced water midstream business in the Anadarko Basin of Oklahoma.
 - "Business Benefit Plan" means (a) any Target Company Benefit Plan or (b) any Seller Benefit Plan.
- "Business Day" means any day, other than Saturday and Sunday, on which federally insured commercial banks in Houston, Texas, are generally open for business and capable of sending and receiving wire transfers.
 - "Buyer" is defined in the Preamble.
 - "Buyer's 401(k) Plan" is defined in Section 6.10(g).
- "Buyer Fundamental Representations" means Section 5.1 (Organization, Good Standing, and Authority), Section 5.2(b)(i) (No Conflicts), Section 5.5 (Broker's or Finder's Fees) and Section 5.6 (Investment Intent).
 - "Buyer Obligations" is defined in Section 11.16(a).
 - "Buyer Parties" means Parent and its Subsidiaries, including the Buyer and, after the Closing, the Target Companies.

"Buyer Plans" means the benefit or compensation plans, policies, programs, arrangements or agreements maintained by the Buyer or any of its Affiliates, including after the Closing, the Target Companies.

"Buyer Related Parties" is defined in Section 9.2(e).

"Buyer Released Person" is defined in Section 11.15(b).

"Buyer Releasing Person" is defined in Section 11.15(a).

"Buyer's Closing Certificate" is defined in Section 8.2(b)(iii).

"CapEx Amount" means the amount of capital expenditures actually made or incurred on behalf of the Target Companies or the Assets by the Seller or any of its Affiliates (other than the Target Companies) during the period between the Execution Date and the Closing, provided that the CapEx Amount shall not exceed 110% of the CapEx Budget.

"CapEx Budget" means the schedule of budgeted capital expenditures for the Target Companies with respect to the period between the Execution Date and the Closing, attached hereto as Schedule 1.1(a), subject to daily proration for the applicable monthly budget for the month in which the Closing occurs.

"Cash and Cash Equivalents" means the net amount, determined as of the Measurement Time, of (a) the Target Companies' money, currency or a credit balance in a deposit account at a financial institution, net of checks outstanding as of the time of determination and (b) (i) marketable direct obligations issued or unconditionally guaranteed by the United States government or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing within one (1) year from the date of acquisition, (ii) marketable direct obligations issued by any state of the United States or any political subdivision of any such state or any public instrumentality thereof maturing within one (1) year from the date of acquisition and, at the time of acquisition, having the highest rating obtainable from either Standard & Poor's Corporation or Moody's Investors Service, Inc., (iii) commercial paper issued by any bank or any bank holding company owning any bank maturing no more than one (1) year from the date of its creation and, at the time of acquisition, having the highest rating obtainable from either Standard & Poor's Corporation or Moody's Investors Service, Inc. and (iv) certificates of deposit or bankers' acceptances maturing within one (1) year from the date of acquisition issued by any commercial bank organized under the Laws of the United States.

"Casualty Loss" means any damage, destruction, eminent domain taking or other casualty loss (whether or not covered by insurance) affecting the Business, the Assets or the Target Companies.

"CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980.

"Claim" means any claim, Liability, Loss, demand, damage, cause of action of any kind, Order, subpoena, obligation, cost, fee, assessment, duty, charge, penalty, fine, or judgment (including recoverable legal counsel fees and costs of litigation of the Person asserting the Claim), whether arising in equity, under law, contract, tort, voluntary settlement or in any other manner.

"Class B Units" means the Class B Units of the Target.

"Class B Unit Value" means the Fair Market Value (as defined in the Target Stock Plan) of a Class B Unit of the Target as of the Closing, as determined by the Board of Managers of the Target, in accordance with the relevant governing documents.

"Closing" is defined in Section 8.1.

"Closing Amount" is defined in Section 2.3(b)(iv).

"Closing Date" is defined in Section 8.1.

"Closing Indebtedness" means, determined as of the Measurement Time, an amount equal to all Indebtedness of the Target Companies after giving effect to the payment of the Payoff Amounts at Closing.

"Closing Working Capital" is defined in Section 2.4(b).

"Code" means the Internal Revenue Code of 1986, as amended.

"Company Transaction Cost Amount" means the aggregate amount of all Company Transaction Costs that will remain unpaid as of the consummation of Closing.

"Company Transaction Costs" means all investment banking, legal, and accounting fees and other costs and expenses incurred by the Seller and/or any Target Company in connection with the preparation for, negotiating or consummation of the transactions contemplated by this Agreement and the other Transaction Documents.

"Confidentiality Agreement" means that Confidentiality and Non-Disclosure Agreement, dated as of October 16, 2019, by and between the Target and Citizen Energy Holdings, LLC.

"Continuing Employee" is defined in Section 6.10(b).

"Contract" means any contract, agreement, understanding, option, right to acquire, preferential purchase right, preemptive right, warrant, indenture, debenture, note, bond, loan, loan agreement, lease, mortgage, franchise, license, purchase order, bid, commitment, letter of credit, guaranty, surety or any other legally binding arrangement excluding any tariffs and Authorizations.

"Contribution Agreement" means that certain Contribution Agreement by and among Linn Energy Holdings, LLC, Linn Operating, LLC, Citizen Energy II, LLC and Roan Resources LLC dated June 27, 2017.

"Current Assets" means the sum of all current assets of the Target Companies as of the Measurement Time as determined in accordance with the Accounting Principles, including (a) accounts receivable (including trade receivables, unbilled receivables, and other receivables) that are reasonably expected to be realized in cash or consumed in the Ordinary Course of Business, (b) condensate inventory, (c) prepaid expenses, (d) deposits, (e) current Tax assets set forth on Schedule 1.1(k) (as updated in connection with the determination of the Estimated Adjustment Amount and the Final Adjustment Amount) and (f) other current assets, but excluding in all cases, (u) for the avoidance of doubt, Tax assets attributable to the Sales Tax Contest, (v) deferred Tax assets, (w) all Cash and Cash Equivalents, (x) any prepaid expenses that do not relate to the corresponding obligations of the Target Companies after the Closing, including any prepaid insurance amounts, (y) inventory reclassified from products, plant and equipment to current assets in the Interim Financial Statements, and (z) any amounts receivable owed to any Target Company by the Seller or any of its Affiliates (other than the Target Companies). An illustrative computation of Current Assets as of the Balance Sheet Date is set forth on the Illustrative Working Capital Schedule.

"Current Liabilities" means the sum of all current liabilities of the Target Companies as of the Measurement Time as determined in accordance with the Accounting Principles, including (a) accounts payable, (b) current Tax Liabilities set forth on Schedule 1.1(k) (as updated in connection with the determination of the Estimated Adjustment Amount and the Final Adjustment Amount), (c) any other short term Liabilities or accruals, and (d) the Accrued Reimbursable Expenses Amount, but excluding in all cases, (i) the Closing Indebtedness, (ii) the Payoff Amounts, (iii) deferred Tax Liabilities, (iv) all obligations of any of the Target Companies under any swaps, hedges or similar instruments, and (v) any CapEx Amounts incurred but not yet paid. An illustrative computation of Current Liabilities as of the Balance Sheet Date is set forth on the Illustrative Working Capital Schedule. For purposes of determining Current Liabilities to be used in the determination of Working Capital, no reserves, allowances or accrued Liability of any Target Company reflected in the Financial Statements shall be reduced or eliminated, except in the case of a reduction or elimination by reason of payment or credit occurring in the Ordinary Course of Business.

"D&O Insurance" is defined in Section 6.8(b).

"Debt Commitment Letter" means that certain executed commitment letter, dated as of the date of this Agreement (together with each exhibit, annex, schedule or other attachment thereto, as amended, supplemented, replaced, extended or otherwise modified in accordance with Section 6.21(a) from time to time after the date of this Agreement) from the lenders party thereto (including any lenders who become party thereto by joinder in accordance with the terms thereof) (collectively, the "Lenders") pursuant to which the Lenders have agreed, subject to the terms and conditions thereof, to provide the debt amounts set forth therein (such debt financing and any alternative debt financing, the "Debt Financing").

"Debt Finance Related Parties" is defined in Section 11.14(b).

"Debt Financing" is defined in the term "Debt Commitment Letter".

"*Debt Financing Source*" means each Lender (as defined in the definition of "Debt Commitment Letter"), or financing provider under any alternative debt financing, agent, arranger, underwriter, initial purchaser and placement agent that has committed to provide or arrange Debt Financing or any replacement debt financing contemplated by <u>Section 6.21</u> or any Affiliates or Representatives of any such Person.

- "Debt Payoff Letter" is defined in Section 8.2(a)(xii).
- "Definitive Agreements" is defined in Section 6.21(a).
- "Deposit" is defined in Section 2.2(a).
- "Deposit Escrow Account" means the account designated as the "Deposit Escrow Account" maintained by the Escrow Agent pursuant to the Escrow Agreement.
 - "Designated Agreements" means the Contracts listed on Schedule 1.1(j).
 - "Disclosure Schedule" means the disclosure schedules attached to this Agreement and incorporated herein.
- "*Employee Benefit Plan*" means any "employee benefit plan" (within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("*ERISA*")) and any other material benefit or compensation plan, policy, program, Contract, agreement or arrangement, but excluding any Multiemployer Plan.
- "Environment" means (a) land, including surface land, soil, sub-surface strata, sediment, sea bed and river bed under water (as defined in subclause (b)); (b) water, including coastal and inland water, surface waters (permanent or ephemeral); and ground waters; (c) ambient air; and (d) indoor air.
 - "Environmental Claims" is defined in Section 4.13(e).
- "Environmental Law" means any Law or legally-binding interpreting guidance, to the extent applicable to the person or properties in context of which the term is used, pertaining to the pollution or protection of the Environment, or regulating pollution, environmental protection, waste management, the use, storage, generation, treatment, remediation, removal, disposal, transport of Hazardous Material, regulating or prohibiting Releases of Hazardous Materials into the Environment, or pertaining to the protection of natural resources, the Environment or, to the extent relating to exposure of Hazardous Materials, human health or safety, as such Laws have been and may be amended or supplemented through the date of this Agreement.
 - "Environmental Permits" is defined in Section 4.13(b).
- "Equity Commitment Letter" means that certain executed commitment letter, dated as of the date of this Agreement (together with each exhibit, annex, schedule or other attachment thereto, as amended, supplemented, replaced, extended or otherwise modified from time to time after the date of this Agreement) from each of the investor parties thereto (collectively, the "Investors") pursuant to which the Investors have agreed, subject to the terms and conditions thereof, to provide the amounts set forth therein (such financing, the "Equity Financing").

"*Equity Interest*" means (a) the equity ownership rights in a business entity, whether a corporation, company, joint stock company, limited liability company, general or limited partnership, joint venture, bank, association, trust, trust company, land trust, business trust, sole proprietorship, or other business entity or organization, and whether in the form of capital stock, ownership unit, limited liability company interest, limited or general partnership interest, or any other form of ownership, and (b) also includes all Equity Interest Equivalents.

"*Equity Interest Equivalents*" means all rights, warrants, options, convertible securities or Indebtedness, exchangeable securities or other instruments, or other rights that are outstanding and exercisable for or convertible or exchangeable into, directly or indirectly, any Equity Interest described in <u>clause (a)</u> of the definition thereof at the time of issuance or upon the passage of time or occurrence of some future event.

"ERISA" is defined in this Schedule A in the definition of "Employee Benefit Plan."

"ERISA Affiliate" means any entity or trade or business (whether or not incorporated) that together with any of the Target Companies is considered under common control and is currently treated as a single employer under Section 414(b), (c), (m) or (o) of the Code.

"Escrow Accounts" means, collectively, the Deposit Escrow Account and the Indemnity Escrow Account.

"Escrow Agent" means JPMorgan Chase Bank, National Association, or any successor thereto.

"*Escrow Agreement*" means that certain Escrow Agreement, in substantially the form attached hereto as <u>Exhibit E</u>, dated as of the Execution Date, by and among the Buyer, the Seller and the Escrow Agent, as may be amended, modified, supplemented and restated from time to time.

"Estimated Adjustment Amount" is defined in Section 2.4(a).

"Estimated Closing Statement" is defined in Section 2.4(a).

"Estimated Working Capital" is defined in Section 2.4(a).

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Execution Date" is defined in the Preamble.

"Exhibits" means any or all of the exhibits attached to and made a part of this Agreement.

"Existing Credit Facility" means that certain Credit Agreement, dated as of August 10, 2018, by and among the Target, Royal Bank of Canada as administrative agent and the lenders from time to time party thereto, as amended, modified, supplemented and restated from time to time.

"FCC" means the Federal Communications Commission.

"FCC Licenses" means any licenses, permits, certificates, approvals, franchises, consents, waivers, registrations or other authorizations issued by the FCC.

"Federal Trade Commission Act" means the Federal Trade Commission Act of 1914, as amended, and the rules and Regulations promulgated thereunder.

"Fee Letter" is defined in Section 5.11(b).

"Final Adjustment Amount" is defined in Section 2.4(e).

"Final Allocation Schedule" is defined in Section 2.5(b).

"Financial Statements" is defined in Section 4.12(a)(ii).

"Fraud" means (a) with respect to the Seller (as determined pursuant to a final non-appealable order of a court of competent jurisdiction), a false representation and warranty contained in Article IV or in the Seller's Closing Certificate made with the Knowledge of the Seller that such representation and warranty is false when made and with the intent to induce the Buyer to enter into this Agreement or to act in reliance upon the false representation and warranty that causes the Buyer, in reliance upon such false representation and warranty, to enter into or consummate this Agreement and to take action and suffer Losses by reason of such reliance and (b) with respect to the Buyer (as determined pursuant to a final non-appealable order of a court of competent jurisdiction), a false representation and warranty contained in Article V or in the Buyer Closing Certificate made with the Knowledge of the Buyer that such representation and warranty is false when made and with the intent to induce the Seller to enter into this Agreement or to act in reliance upon the false representation and warranty that causes the Seller, in reliance upon such false representation and warranty, to enter into or consummate this Agreement and to take action and suffer Losses by reason of such reliance. For the avoidance of doubt, "Fraud", with respect to the Seller or the Buyer, does not include any fraud based on constructive knowledge, negligent misrepresentation, recklessness, or any similar theory.

"GAAP" means generally accepted Accounting Principles used in the United States, in effect from time to time and applied consistently with Financial Statements and such Party's past practices.

"Gathering Systems and Plants" means any pipeline system or processing plant owned or leased by a Target Company that is used in the Business.

"Governmental Authority" means any instrumentality, subdivision, court, tribunal, administrative or arbitral body, agency, commission, commonwealth, parish, official or other authority of any country or any state, province, prefect, municipal or local jurisdiction or other government or political subdivision thereof, or any quasi-governmental or private body exercising any regulatory, taxing or other governmental or quasi-governmental authority.

"Hazardous Material" means: (a) any chemicals, materials, wastes, substances or any contaminant or pollutant defined as "hazardous" or "toxic" under any Law pertaining to the Environment or health or safety or otherwise classified by a Governmental Authority as "hazardous" or "toxic" (including per-fluorinated compounds); (b) any radioactive materials (including naturally occurring radioactive materials or technologically enhanced naturally occurring radioactive materials), asbestos or asbestos containing materials and polychlorinated biphenyls; or (c) petroleum and petroleum derivatives, including, but not limited to, crude oil and its fractions.

"Hydrocarbon" means oil, gas, casinghead gas, drip gasoline, natural gasoline, condensate, distillate, liquid hydrocarbons, gaseous hydrocarbons and all products refined or separated therefrom.

"Illustrative Working Capital Schedule" means the Illustrative Working Capital Schedule attached hereto as Schedule 1.1(b).

"*Inactive Employee*" shall mean any Offer Employee who is absent from active employment on the Closing Date on account of vacation, paid time off, ordinary sick leave reasonably expected to result in an absence of short duration, workers' compensation leave, short-term or long-term disability leave, leave under the U.S. federal Family and Medical Leave Act or leave under any other Law, or any other approved leave of absence.

"Indebtedness" means, without duplication, the aggregate amount, of the following obligations: (a) any indebtedness for borrowed money; (b) any obligations evidenced by bonds, debentures, notes or other similar instruments; (c) any obligations in the nature of accrued interest, prepayment penalties, expenses, or fees in respect of any of the foregoing; (d) any reimbursement obligations and obligations with respect to letters of credit, bankers' acceptances and bank guarantees; (e) any obligations secured by a Lien on the property or assets of such Person, whether or not the obligation so secured is a primary obligation of, or has been assumed by, such Person or similar facilities; (f) any obligations of such Person to pay the deferred purchase or acquisition price of any property, including any "earn-out" payments or similar obligations (other than trade those trade payables incurred in the Ordinary Course of Business); (g) any obligations of such Person under any currency, commodity or interest rate swap, future, hedge or similar protection device, the principal purpose of which is to benefit from or reduce or eliminate the risk of fluctuations in interest rates or currencies; (h) any capital lease obligations required to be classified as such on a balance sheet prepared in accordance with GAAP; and (i) any obligations described in the foregoing clauses (a) through (b) of any Person other than one of the Target Companies, the payment of which is guaranteed by one of the Target Companies; provided, however, that Indebtedness will expressly exclude trade account payables incurred in the Ordinary Course of Business.

Notwithstanding the foregoing, "Indebtedness" does not include any operating or lease obligations (other than as set forth in the foregoing clause (b)), the Required Intercompany Bonds, performance bonds, whether or not matured, any obligations secured by a Lien listed in Schedule 1.1(d), or intercompany obligations between or among the Target Companies.

"Indemnity Escrow Account" means the account designated as the "Indemnity Escrow Account" maintained by the Escrow Agent pursuant to the Escrow Agreement.

- "Indemnity Escrow Fund" is defined in Section 2.3(b)(i).
- "Indemnity Escrow Fund Release Date" means December 23, 2020.
- "Inspection Indemnitees" is defined in Section 6.2(d).
- "Intellectual Property" is defined in Section 4.8.
- "Interests" is defined in the Recitals.
- "Interim Financial Statements" is defined in Section 4.12(a)(ii).
- "IRS" means the United States Internal Revenue Service.
- "Knowledge" means (a) with respect to the Buyer, the actual knowledge of any of Robbie Woodard, Tim Helms and James Woods, after due inquiry, and (b) with respect to the Seller, the actual knowledge of any of David Rottino, David Weathers, James G. Frew and Chris Ditzel, after due inquiry.
- "*Law*" or "*Laws*" means (i) common law, (ii) any law, statute, treaty, code, ordinance, Order, rule, Regulation, judgment, injunction, or other legally-binding requirement of any Governmental Authority in effect at such time or (iii) any legally-binding obligation included in any Permit or resulting from binding arbitration, including any requirement under common law.
 - "Leased Personal Property" is defined in Section 4.15(a).
 - "Leased Real Property" is defined in Section 4.15(a).
 - "Lenders" is defined in term "Debt Commitment Letter".
- "*Liability*" means any direct or indirect liability, Indebtedness, obligation, commitment, expense, Claim, deficiency, guaranty, or endorsement of or by any Person of any type, whether known or unknown, and whether accrued, absolute, contingent, matured, or unmatured.
- "*Lien*" means any lien (contractual, statutory, constitutional or otherwise), adverse Claim, production payment, mortgage, security interest, pledge, charge, encumbrance, hypothecation or deposit arrangement, restriction, burden, defect in title, right of purchase, or right of a vendor under any title retention or conditional sale agreement or lease or other arrangement substantially equivalent thereto.
- "LINN Tax Matters Agreement" means that certain Tax Matters Agreement entered into as of August 7, 2018 by and among Linn Energy, Inc., a Delaware corporation, Riviera Resources, Inc., a Delaware corporation, and certain other parties identified therein.
 - "LINN Tax Matters Termination Agreement" is defined in Section 6.10(d).
- "Loss" or "Losses" means any and all damages, payments, penalties, assessments, disbursements, costs, and expenses, including interest, awards, judgments, settlements, fines, costs of remediation, fees, costs of defense and reasonable attorneys' fees, costs of accountants, expert witnesses and other professional advisors, and costs of investigation and preparation of any kind or nature whatsoever.

"Material Contract" means each of the following Contracts to which a Target Company is a party or by which any Assets are bound, excluding Rights-of-Way:

- (a) Material Gathering Contracts;
- (b) Material Sales Contracts;
- (c) Each transportation Contract that provides for aggregate payments by any Target Company during any fiscal year of such Target Company in excess of \$1,000,000;
- (d) Contracts under which any Target Company has (i) created, incurred, assumed or guaranteed, directly or indirectly, any outstanding Indebtedness, (ii) granted a Lien on its assets to secure such Indebtedness or (iii) extended credit to any Person;
- (e) each Contract involving a remaining commitment, as of the Execution Date, to pay capital expenditures in excess of \$250,000 in the aggregate;
 - (f) partnership or joint venture Contracts with Persons other than the Seller or another Target Company;
- (g) each Contract relating to the disposition of Assets (other than in the Ordinary Course of Business or Material Sales Contracts or Material Gathering Contracts) for consideration in excess of \$250,000, and each Contract for the grant to any Person of any preferential rights to purchase any Equity Interests or material Assets of the Target Companies or that grants any right of first refusal or first offer with respect to any Equity Interests or material Assets of the Target Companies;
- (h) each Contract providing for the acquisition or disposition of any business or division of any business (whether by merger, purchase or sale of Equity Interests or assets or otherwise);
- (i) each Contract for lease of personal property or real property involving aggregate payments in excess of \$250,000 in any future calendar year and of a term of one (1) year or more;
- (j) each Contract providing for aggregate future payments to or from any Target Company in excess of \$250,000 in any calendar year, other than such Contracts that can be terminated without material penalty by any such Target Company upon ninety (90) days' notice or less; (excluding, for the avoidance of doubt, gathering, treating, transportation, processing and sales contracts);
- (k) (i) any capital lease or (ii) any other lease or other Contract relating to equipment held or used by any Target Company providing for aggregate annual rental payments in excess of \$250,000;

- (l) any Contract that contains any covenant of such Target Company that materially limits or purports to limit the ability of such Target Company to conduct or compete in any line of business or with any Person in any geographic area;
- (m) except with respect to any Organizational Document of the Target Companies, any agreement (A) with the Seller or any Affiliate of the Seller or (B) between any Target Company, on the one hand, and any other Target Company, on the other hand;
 - (n) any Contract that is a currency, commodity or interest rate swap, future, hedge or similar protection device;
- (o) any Contract that provides for the acquisition or construction of processing plants, gathering systems, pipeline systems or other related assets;
 - (p) any Contract that provides for the acquisition, construction or operation of saltwater disposal wells;
- (q) any Contract with any financial advisor or consultant under which there are remaining indemnity or other obligations after the Closing, including any financial advisory, oversight or similar agreement with the Seller or any of its Affiliates; and
- (r) any Contract the breach or termination of which would, individually or in the aggregate, reasonably be expected to have a Target Material Adverse Effect.
- "Material Gathering Contract" means each gathering, treating, transportation, disposal or processing Contract for natural gas, crude oil or water entered into by a Target Company or by which a Target Company is bound.
 - "Material Sales Contract" means each Hydrocarbon sales Contract entered into by a Target Company or by which a Target Company is bound.
 - "Measurement Time" means 12:01 a.m. (Central Time) on the Closing Date.
- "*Measurement Time Working Capital*" means, determined as of the Measurement Time, the Working Capital of the Target Companies as determined in accordance with the Accounting Principles.
 - "Multiemployer Plan" means a "multiemployer plan" as defined in Section 4001(a)(3) or 3(37) of ERISA.
 - "NORM" means naturally occurring radioactive material.
- "Notification" means any notice to or filing with any Person or Governmental Authority required under the terms of any Contract to which the Seller or a Target Company is a party, by the terms of any Authorization held by or applicable to the Seller or a Target Company or by Law that is necessary for the Seller to execute, deliver, and perform its obligations under this Agreement and the Transaction Documents to which it is or shall be a party or is otherwise required in connection with the consummation by the Seller of the transactions contemplated hereby or thereby.

"Offer Employees" is defined in Section 6.10(b).

"Order" means all applicable writs, judgments, injunctions, decrees, and other official acts of or by any Governmental Authority.

"Ordinary Course of Business" means the ordinary course of business consistent with the past practice of the Business or the Target Companies, as applicable, during the two (2)-year period prior to the date hereof (or such shorter period referenced in the provision in which this term is used), including (i) taking any actions contemplated by this Agreement in connection with the transactions contemplated hereby, (ii) shut-ins of producing wells, (iii) impacts of, and matters relating to, the COVID-19 pandemic and (iv) changes in management and resulting management and accounting practices beginning in April of 2020.

"Organizational Documents" means with respect to any particular entity: (a) if a corporation, its articles or certificate of incorporation and its bylaws; (b) if a limited partnership, its limited partnership agreement and its articles or certificate of limited partnership; (c) if a limited liability company, its articles of organization or certificate of formation and its limited liability company agreement or operating agreement; (d) all related equity holders' agreements, voting agreements, voting trust agreements, joint venture agreements or registration rights agreements; and (e) any amendment or supplement to any of the foregoing.

"Outside Date" means ninety (90) calendar days after the Execution Date.

"Owned Real Property" is defined in Section 4.15(a).

"Parent" means Citizen Energy Operating, LLC.

"Parent Guaranty" is defined in Section 11.16(a).

"Party" and "Parties" are defined in the Preamble.

"Payment Deadline" is defined in Section 2.6(c).

"Payoff Amounts" is defined in Section 8.2(b)(ii).

"Payoff Indebtedness" means any Indebtedness described on Schedule 1.1(c).

"*Permits*" means any permits, licenses, approvals, notices, waivers, registrations, authorizations, variances, exemptions, Orders, franchises, certifications, certificates or other authorizations of any Governmental Authority.

"Permitted Encumbrances" means the following:

- (a) terms, conditions, restrictions, exceptions, reservations, limitations, and other matters contained in any document filed or recorded in the appropriate county or parish to reflect title thereto, creating, transferring, limiting, encumbering, or reserving or granting any rights in (including rights of reverter, reservation and life estates) any Real Property Assets that, singly or in the aggregate, do not materially adversely affect the Business of the Target and/or the Target Companies;
- (b) statutory Liens for Taxes that are not yet due and payable or Liens for Taxes that are being contested in good faith in the Ordinary Course of Business by appropriate Proceedings, in each case for which adequate reserves have been established on the Financial Statements in accordance with GAAP:
- (c) mechanics', materialmen's, repairmen's and other statutory Liens arising in the Ordinary Course of Business and securing obligations incurred prior to the Closing Date that are not delinquent, that will be paid and discharged in the Ordinary Course of Business and for which adequate reserves have been made in the Financial Statements;
- (d) easements, restrictive covenants, defects, and other irregularities in title, that, singly or in the aggregate, do not materially adversely affect the value of the assets to which such matters relate or materially interfere with the ownership, use, or operation of such assets and that are of a nature that would be reasonably acceptable to a prudent pipeline or plant operator;
- (e) Required Third-Party Consents to assignment, preferential purchase rights, and other similar agreements with respect to which consents or waivers are obtained from the appropriate Person prior to the Closing for the transactions contemplated hereby, or as to which the appropriate time for asserting such rights has expired as of the Closing without an exercise of such right;
 - (f) any Post-Closing Notification;
 - (g) Liens paid or discharged at or prior to the Closing;
 - (h) Liens created by either the Buyer or its respective successors or assigns;
 - (i) the Liens listed in Schedule 1.1(d);
 - (j) all zoning, entitlement and other land use restrictions under applicable Laws (including Environmental Laws);
 - (k) defects or irregularities that have been cured or remedied by applicable statutes of limitation;
- (l) defects or irregularities in the chain of title consisting of omissions of heirship Proceedings if the matters to be covered by such omitted heirship Proceedings are covered by appropriate affidavits of heirship;
 - (m) defects or irregularities in the chain of title consisting of the failure to recite marital status in documents;

- (n) all rights reserved to or vested in any Governmental Authority to control or regulate any of the Assets in any manner or to assess Tax with respect to the Assets, the ownership, use or operation thereof, or revenue, income or capital gains with respect thereto, and all obligations and duties under all applicable Laws of any such Governmental Authority or under any franchise, grant, license or Permit issued by any Governmental Authority; and
 - (o) Liens created by this Agreement.

"Permitted Equity Encumbrances" means:

- (a) Liens in favor of the holders of the Interests pursuant to the terms of the Organizational Documents of the applicable entity (which will be terminated, waived, complied with or conveyed in full to the Buyer at Closing);
 - (b) restrictions on sales of securities under generally applicable securities Laws; and
 - (c) Liens created by either the Buyer or its respective successors or assigns.
- "*Person*" means any natural person, corporation, company, partnership (general or limited), limited liability company, trust, joint venture, joint stock company, unincorporated organization, Governmental Authority, or other entity or association.
- "Personal Property Assets" means all fixtures, equipment, improvements, and tangible personal property Assets of the Target Companies, including those Assets described in clauses (a) through (d) of the definition of "Target Company Systems".
- "*Post-Closing Notification*" means any Notification to or with or consent from any Person or Governmental Authority for the assignment, transfer or change of control of the Interests, Target Companies, Assets or operations that are customarily made or obtained after the closing of a similar assignment, transfer or change of control, including those listed in <u>Schedule 1.1(e)</u>, but such term shall not include any Notification that constitutes a Required Third-Party Consent or a Required Notification.
 - "Post-Closing Statement" is defined in Section 2.4(b).
 - "Pre-Closing Confidential Information" means "Confidential Information" as defined in the Confidentiality Agreement.
 - "Pre-Closing Tax Period" is defined in Section 6.7(a)(i).
- "*Proceeding*" means any action, suit, charge, Claim, investigation, review, or other judicial or administrative proceeding, at Law or in equity, before or by any Governmental Authority or arbitration proceeding.

"PSU Consideration" is defined in Section 2.6(b).

- "R&W Insurance Policy" is defined in Section 6.9(a).
- "Real Property Assets" is defined in Section 4.15(d).
- "Real Property Leases" is defined in Section 4.15(a).
- "Records" is defined in Section 6.13.
- "Recourse Party" is defined in Section 11.14.
- "Regulation" means any rule or regulation of any Governmental Authority having the effect of Law or of any rule or regulation of any self-regulatory organization.
- "*Reimbursable Expenses*" means, without duplication, all intercompany payables or other amounts owed by any of the Target Companies to the Seller or any of its Affiliates (other than the Target Companies) in connection with any of the items or matters described on <u>Schedule 1.1(i)</u>.
- "*Release*" means any release, spill, emission, leaking, pumping, pouring, emitting, escaping, injection, deposit, disposal, discharge, dispersal or leaching into the Environment.
 - "Released Claims" is defined in Section 11.15(a).
- "Remedial Action" means all actions pursuant to Environmental Law to (a) clean up, respond to, remove, treat, or in any other way address any Hazardous Material in the Environment; (b) prevent the Release, or minimize the further Release, of any Hazardous Material so it does not endanger or threaten to endanger human health or the Environment; or (c) perform pre-remedial studies and investigations or post-remedial monitoring and care pertaining or relating to a Release of Hazardous Material.
- "Representatives" with respect to any Person means such Person's Affiliates and its and their sponsors, managers, members, partners, directors, officers, employees, agents or advisors (including attorneys, accountants, consultants, bankers, financial advisors and any representatives of those advisors).
 - "Required Intercompany Bonds" is defined in Section 6.17(d).
 - "Required Notifications" is defined in Section 4.3.
 - "Required Third-Party Consent" is defined in Section 4.3.
 - "Restraint" is defined in Section 7.1(a).
 - "Rights-of-Way" is defined in Section 4.15(b).
 - "Riviera Operating" means Riviera Operating, LLC, a Delaware limited liability company.
 - "RSU Consideration" is defined in Section 2.6(a).

"Sales Tax Contest" means that certain Oklahoma sales tax audit of or with respect to a Target Company for sales tax paid on exempt electricity used/consumed in a manufacturing operation over taxes due between July 2018 and April 2019, and to which Ernst & Young has agreed to pursue on a contingency basis.

"Schedules" means the schedules referenced in this Agreement and attached hereto.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Seller" is defined in the Preamble.

"Seller's 401(k) Plan" is defined in Section 6.10(f).

"Seller's Marks" is defined in Section 6.11.

"Seller Benefit Plan" means any Employee Benefit Plan that is sponsored or maintained by the Seller or any of its Subsidiaries in which any Available Employee participates, or for which the Seller or any of its Subsidiaries has any Liability with respect to any Available Employee, but in each case, excluding any Target Company Benefit Plan.

"Seller Counsel" is defined in Section 11.8(a).

"Seller Fundamental Representations" means the representations and warranties set forth in Section 3.1 (Organization, Good Standing, and Authority of the Seller), Section 3.2 (Title to the Interests), Section 3.3(b)(i) (No Conflicts), Section 3.5 (Broker's or Finder's Fees), Section 4.1 (Organization, Good Standing, and Authority; Capitalization of the Target Companies), Section 4.2(a) (No Conflicts), and Section 4.9 (Broker's or Finder's Fees).

"Seller Insurance" is defined in Section 6.5.

"Seller Parties" means Riviera Resources, Inc. and its Subsidiaries, including the Seller and, prior to the Closing, the Target Companies.

"Seller Related Parties" is defined in Section 9.2(e).

"Seller Released Person" is defined in Section 11.15(a).

"Seller Releasing Person" is defined in Section 11.15(b).

"Seller's Closing Certificate" is defined in Section 8.2(a)(i).

"Solvent" means, with regard to any Person, that (a) the sum of the assets of such Person, both at a fair valuation and at present fair salable value, exceeds its Liabilities, including contingent, subordinated, unmatured, unliquidated and disputed Liabilities, (b) such Person has sufficient capital with which to conduct its business, and (c) such Person has not incurred debts beyond its ability to pay such debts as they mature. For purposes of this definition, "debt" means any Liability on a claim, and "claim" means (x) a right to payment, whether or not such right is

reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured or unsecured or (y) a right to an equitable remedy for breach of performance to the extent such breach gives rise to a payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, disputed, undisputed, secured or unsecured. With respect to any such contingent Liabilities, such Liabilities shall be computed at the amount which, in light of all the facts, conditions and circumstances existing at the time, represents the amount which can reasonably be expected to become an actual or matured Liability.

"Sponsor" means Warburg Pincus LLC.

"Straddle Period" means any Tax period beginning on or before and ending after the Closing Date.

"Subsidiary" means, with respect to any Person, any corporation or other organization, whether incorporated or unincorporated, of which (a) such Person or any other subsidiary of such Person is a general partner, managing member, or sole or controlling member or (b) at least a majority of the Equity Interests or other interests having by their terms ordinary voting power to elect a majority of the board of directors, managers or others performing similar functions with respect to such corporation, partnership, limited partnership, limited liability company, or other organization is, directly or indirectly, owned or controlled by such Person or by any one or more of its subsidiaries, or by such Person and any one or more of its subsidiaries.

"Substitute Debt Financing" means alternative Debt Financing or such portion of the Debt Financing, on terms (including structure, covenants and pricing) not materially less favorable (taken as a whole) to Buyer (as reasonably determined by Buyer) or as otherwise acceptable to Buyer and with lenders or holders, as applicable, reasonably satisfactory to Buyer, from the same or alternative sources, which may include one or more of a senior secured debt financing or other financing, or any combination thereof, in an amount such that the aggregate funds that would be available to Buyer at the Closing, including cash on hand, will be sufficient to satisfy Buyer's obligations under this Agreement and to consummate the transactions contemplated hereby and to pay all fees and expenses required to be paid in connection herewith and with the Debt Financing.

"SWD Wells" means the salt water disposal wells owned or operated by any Target Company and associated property.

"Synthetic Equity Interest Agreement" means that certain Synthetic Equity Interest Agreement, dated as of December 22, 2017, among (i) Gaedeke Merge, LP, (ii) Gaedeke Holdings VII, Ltd. and (iii) the Target.

"System Data" is defined in Section 4.24.

"Target" is defined in the Recitals.

"Target Assignment Agreement" means an assignment instrument, in the form of Exhibit A, evidencing the Seller's assignment and transfer to the Buyer of the Interests.

"Target Company" or "Target Companies" means any of the Target or any of the Target Subsidiaries, or, collectively, Target and all of Target Subsidiaries.

"*Target Company Benefit Plan*" means all Employee Benefit Plans that are sponsored, maintained, contributed to or required to be contributed to by any Target Company for the benefit of current or former employees, directors or individual consultants of any Target Company and for which a Target Company will have Liability following the Closing.

"Target Company Systems" means (a) the natural gas, water and crude pipeline gathering systems described on Schedule 1.1(f), (b) the processing plants described on Schedule 1.1(f), (c) the SWD Wells described on Schedule 1.1(g), (d) the treating facilities, compressor stations, separators, storage tanks or facilities, meters, pumps, valves, gauges, fittings, machinery, equipment, improvements, fixtures, and other personal property owned by any Target Company or located on or used in connection with the Real Property Assets or the properties described in clauses (a) through (c), (e) the Real Property Assets, including surface leases and other rights to use the surface and other interests in real property relating to the Assets, and all buildings, structures, fixtures and improvements thereon and appurtenances thereto owned or otherwise held by the Target Companies or used in connection with the Business or Assets, (f) the Permits relating to the Assets, and (g) the Contracts relating to the Assets.

"Target Equity Awards" means, collectively, the Target PSUs and the Target RSUs.

"Target Material Adverse Effect" means, any result, change, effect, event, circumstance, development, condition or occurrence that, individually or in the aggregate, would reasonably be expected to be, materially adverse to the assets, properties, business, financial condition, results of operations or prospects of the Target Companies, taken as a whole; provided, however, that in no event will any result, change, effect, event, circumstance, development, condition or occurrence that arises out of or results from any of the following be deemed to constitute, or be taken into account in determining whether there has been, a Target Material Adverse Effect: (a) the execution, pendency or announcement of this Agreement and the transactions contemplated hereby or by any Transaction Document (including any delay or any loss of, or adverse change in, the relationship of the Target Companies with their respective customers, partners, employees, financing sources or suppliers caused by the pendency or announcement of the transactions contemplated by this Agreement or the Transaction Documents), (b) changes in commodity prices or conditions affecting the oil and gas industry generally or in the general geographic areas where the Gathering Systems and Plants are located (including changes in commodity prices, general market prices and regulatory changes affecting such industry or geographic areas generally and drilling, producing, gathering, processing, transportation, storing, disposal and marketing activity, costs or margins), (c) changes in general economic, capital markets, regulatory or political conditions in the United States or elsewhere (including interest rate fluctuations or governmental shutdowns or slowdowns), (d) changes in Law, including Environmental Laws (exclusive, however, of any enforcement actions or Claims related to non-compliance or violations of Law), GAAP or regulatory accounting requirements or interpretations thereof, applicable to the Target Companies, (e) fluctuations in currency exchange rates, (f) acts of war, insurrection, sabotage or terrorism, the outbreak or escalation of hostilities involving the United States, the declaration by the United States of a national emergency or war, (g) epidemic, pandemic or disease outbreak (including the COVID-19 virus), (h) any act or omission to act by the Seller, the **Target Companies**

or their respective Affiliates that is expressly permitted by this Agreement or any Transaction Document, or taken (or omitted to be taken) at the written request of the other Parties hereto or thereto (other than acts or omissions in the Ordinary Course of Business of the Seller, the Target Companies or their respective Affiliates), (i) any failure of the Target Companies to meet any budget, projections, forecasts or predictions of financial performance or estimates of revenue, earnings, cash flow or cash position, for any period (*provided*, that this <u>clause (i)</u> shall not prevent a determination that any change or effect underlying such failure to meet projections or forecasts has resulted in a Target Material Adverse Effect) or (j) the matters described on <u>Schedule 1.1(h)</u>; *provided*, *further*, that any result, change, effect, event, circumstance, development, condition or occurrence referred to in the immediately preceding <u>clauses (b)</u>, (c), (d), (f) or (g) will be taken into account for purposes of determining whether there has been a Target Material Adverse Effect to the extent (and only to the extent) such result, change, effect, event, circumstance, development, condition or occurrence adversely affects Target Companies, taken as a whole, in a disproportionately adverse manner as compared to other participants in the domestic midstream oil and gas industry in the general geographic areas where the Gathering Systems and Plants are located.

"*Target PSUs*" means all rights to receive Target Class B Units (or a cash amount equal to the Fair Market Value (as defined in the Target Stock Plan) thereof) subject to time and performance-vesting conditions, granted pursuant to the Target Stock Plan.

"*Target RSUs*" means all rights to receive Target Class B Units (or a cash amount equal to the Fair Market Value (as defined in the Target Stock Plan) thereof) subject solely to time-vesting conditions, granted pursuant to the Target Stock Plan.

"Target Stock Plan" means the Blue Mountain Midstream LLC 2018 Omnibus Incentive Plan.

"*Target Subsidiaries*" means (i) Wildcat Water Gathering LLC, an Oklahoma limited liability company, (ii) Wildcat SWD 1 LLC, an Oklahoma limited liability company, and (iii) Wildcat Oil Gathering LLC, an Oklahoma limited liability company.

"Target Working Capital Amount" means an amount equal to -\$1,500,000.

"Taxe" or "Taxes" means (a) any and all U.S. federal, state, local, foreign, provincial or other taxes, charges, imposts, levies or other duties or assessments of any kind whatsoever in the nature of a tax, including all net income, gross receipts, capital, sales, use, ad valorem, value added, transfer, franchise, margin, profits, net proceeds, alternative or add-on minimum, inventory, goods and services, capital stock, license, registration, leasing, user, withholding, payroll, employment, social security, unemployment, disability, excise, severance, stamp, occupation, property, fuel, excess profits, premium, windfall profit, unclaimed property, escheat, deficiency and estimated taxes, including any and all interest, penalties, fines, additions to tax or additional amounts imposed by any Governmental Authority in connection with respect thereto or in lieu thereof, whether disputed or not, and (b) any Liability for the payment of any item described in clause (a) as a result of being a member of an affiliated, aggregated, combined, consolidated or unitary group (including pursuant to Treasury Regulations Section 1.1502-6 or any analogous or similar provision of state, local or foreign Law), as a transferee or successor, as a result of an express or implied obligation to indemnify any Person or any obligation under any Contract or other agreement or arrangement, by operation of Law or otherwise.

"Tax Consideration" means the Adjusted Purchase Price, as adjusted under this Agreement and made final as of the Determination Date, plus any liabilities or other items treated as amounts realized for U.S. federal income tax purposes, minus any consideration not allocable in an applicable asset acquisition for purposes of Section 1060 of the Code. For avoidance of doubt, neither the PSU Consideration nor the RSU Consideration payable by the Seller or its Affiliates in respect of the Target Equity Awards in accordance with the procedures described in Section 2.6(c) will be treated as part of the Tax Consideration.

"*Tax Return*" means any return, report or similar filing (including any attached schedules, supplements and additional or supporting material) of or relating to Taxes, including any information return, Claim for refund, amended return or declaration of estimated Taxes, and including any amendments with respect thereto.

"Termination and Release" is defined in Section 6.18.

"Termination Fee" is defined in Section 9.2(b).

"*Third-Party Consent*" means any consent, waiver, permission, authorization, or approval of, or exemption by, any Third Person (other than a Governmental Authority).

"Third Person" means any Person other than a Party or its Affiliates.

"*Transaction Documents*" means the Target Assignment Agreement, the Transition Services Agreement, the Escrow Agreement and any other Contract among the Parties that is expressly agreed by the Parties to constitute a Transaction Document for purposes of this Agreement.

"*Transfer Taxes*" means all transfer, documentary, sales, use, stamp, registration, value-added, recording, registration, conveyance, stock transfer, or other similar Taxes incurred as a result of the purchase and sale of the Interests.

"*Transition Services Agreement*" means the Transition Services Agreement between the Buyer, on the one hand, and Riviera Operating, on the other hand, to be entered into at the Closing in substantially the form attached hereto as <u>Exhibit B</u>.

"*Treasury Regulations*" means the regulations promulgated by the United States Department of the Treasury pursuant to and in respect of provisions of the Code.

"Unadjusted Purchase Price" is defined in Section 2.3.

"United States" means the United States of America.

"Updated Available Employee List" is defined in Section 6.10(a).

"Working Capital" means (without duplication), at any time of determination, with respect to the Target Companies, the aggregate value (expressed as a positive or negative number) of (a) the Current Assets of the Target Companies, minus (b) the Current Liabilities of the Target Companies, in each case, calculated in accordance with the line items set forth on the Illustrative Working Capital Schedule.

"Working Capital Deficit" means the amount, if any, by which the Estimated Working Capital or the Closing Working Capital (as applicable) is less than the Target Working Capital Amount; provided, however, that such amount shall be \$0 if the Estimated Working Capital or the Closing Working Capital (as applicable) is equal to the Target Working Capital Amount.

"Working Capital Surplus" means the amount, if any, by which the Estimated Working Capital or the Closing Working Capital (as applicable) exceeds the Target Working Capital Amount; provided, however, that such amount shall be \$0 if the Estimated Working Capital or the Closing Working Capital (as applicable) is equal to the Target Working Capital Amount.

"Year End Financial Statements" is defined in Section 4.12(a)(i).



NEWS RELEASE

RIVIERA RESOURCES ANNOUNCES CASH DISTRIBUTION OF \$1.35 PER SHARE; CLOSES THE SALE OF BLUE MOUNTAIN MIDSTREAM LLC, AND CERTAIN INTERESTS IN UPSTREAM ASSETS; BOARD APPROVAL OF PLAN OF DISSOLUTION; AND LEADERSHIP CHANGES

HOUSTON, October 13, 2020 - Riviera Resources, Inc. (OTCQX: RVRA) ("Riviera" or the "Company") announces the following:

- The Riviera Board of Directors (the "Board") has approved a cash distribution of \$1.35 per share
- The Company closed the following sales:
 - Blue Mountain Midstream LLC ("Blue Mountain") for net proceeds of approximately \$111 million
 - Remaining North Louisiana assets for net proceeds of approximately \$22 million
 - Interests in certain properties located in the Anadarko Basin in Oklahoma for net proceeds of approximately \$13 million
- The Board has approved the dissolution, winding up and liquidation of the Company, and adopted a Plan of Liquidation and Distribution (the "Plan of Liquidation")
- The Company will make changes in leadership in connection with the Plan of Liquidation

Cash Distribution

The Company's Board has approved a cash distribution to shareholders of \$1.35 per share, funded with its cash on hand. The distribution is payable on or around October 27, 2020 to all shareholders of record as of the close of business on October 23, 2020. The Company currently expects that the cash distribution should not constitute a taxable dividend for U.S. federal and state income tax purposes. Rather, the cash distribution would generally constitute a non-taxable return of capital, and a reduction to the tax basis of each recipient's ownership interest in the Company, with any amount exceeding the holder's basis subject to capital gain treatment.

The \$1.35 per share distribution represented approximately 70% of the closing price per share on October 12, 2020, the last trading day before this announcement. Therefore, the ex-dividend date for the cash distribution is expected to be October 28, 2020, the first trading day following the payment date. Shareholders of record on the record date who sell their shares prior to the ex-dividend date will not be entitled to, and will also sell their right to, receive the cash distribution. As of October 12, 2020, there were 58,170,727 shares outstanding of the Company's common stock eligible to receive distributions. Based on the \$1.35 per share distribution, the aggregate cash amount of the distribution is expected to be approximately \$79 million.

Blue Mountain and Upstream Asset Sales Update

On October 8, 2020, the Company closed on the previously announced Blue Mountain sale in an all cash transaction with net proceeds of approximately \$111 million after closing adjustments and transaction costs. Proceeds from the sale were used to repay the total outstanding borrowings of \$80 million under Blue Mountain's credit facility. Following repayment in full of all liabilities, including the Blue Mountain credit facility, the remaining sale proceeds were added to cash on Riviera's balance sheet, and will be used, in part, to fund the cash distribution of \$1.35 per share. With the closing of the transaction, the Company has completely exited the midstream business.

In addition, the Company closed the previously announced sales of its interest in properties located in North Louisiana in September 2020, and interests in certain properties located in the Anadarko Basin in October 2020, for combined proceeds of approximately \$35 million. The Company has also previously announced plans to sell its remaining upstream assets, primarily located in the Mid-Continent region, with plans to close in the fourth quarter of 2020.

Share count includes 263,118 restricted stock units of the Company either granted or designated for issuance to certain employees (the "Riviera RSUs") that vested with the completed sale of Blue Mountain Midstream LLC, which (in combination with the sales of upstream assets) constituted a Change of Control under the Riviera Resources, Inc. 2018 Omnibus Incentive Plan.

Dissolution Plans

On October 12, 2020, the Board approved the dissolution, winding up and liquidation of the Company, and adopted a Plan of Liquidation. Also on October 12, 2020, the required percentage of the Company's shareholders approved, in accordance with the bylaws of the Company, the Plan of Liquidation.

Pursuant to the Plan of Liquidation, the Company will have to pay or provide for the following items:

- i. All claims and obligations known to it,
- ii. Any claim which is the subject of a pending action, suit or proceeding to which it is a party,
- iii. Claims that have not been known to it but are likely to arise within ten (10) years after the date of dissolution,
- iv. Any obligations, to the extent owed, to the Company's existing creditors, after providing for (i) through (iii)

After providing for the items above, the Company will distribute any remaining assets, which the Company currently expects to be between \$0 and \$40 million, to the Company shareholders in accordance with the Company's Certificate of Incorporation, as amended.

The Plan of Liquidation also provides that the Company will make applicable filings and take such other actions as are found to be appropriate to carry out the Plan of Liquidation. The Company intends to file a certificate of dissolution with the State of Delaware, at which time the Company's transfer books and records will be closed and the Company's common stock will cease trading on the OTCQX Market, no later than the close of business on December 31, 2020. After such time, the Company intends to file with the SEC a Certification on Form 15 under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), requesting the suspension of the Company's reporting obligations under Sections 13 and 15(d) of the Exchange Act (the "Deregistration").

Management Changes

In connection with the plan of dissolution, James G. Frew will resign as Executive Vice President and Chief Financial Officer of the Company effective October 23, 2020. Darren Schluter, who is currently the Company's Executive Vice President of Finance and Administration and Chief Accounting Officer will replace Mr. Frew upon his departure.

In addition, Daniel Furbee, Executive Vice President and Chief Operating Officer of the Company, plans to depart at or prior to the planned closing of the sale of the remaining upstream assets in the fourth quarter of 2020. Mr. Frew's and Mr. Furbee's decisions to resign as Officers of the Company was not the result of any disagreement with the Company on any matter relating to the operations, internal controls, policies or practices of the Company.

"We are very pleased to announce the appointment of Darren Schluter as Executive Vice President and Chief Financial Officer of the Company," said David Rottino, President and Chief Executive Officer of Riviera. "He brings a wealth of knowledge and experience and will enhance our efforts to keep the shareholders' best interests in mind as we wind up the Company. I would also like to thank Jim, Dan, and all the Company's employees who will be moving on. Their leadership and contributions were significant, especially given the difficult market environment."

Board Changes

Thanasi Skafidas who in his capacity as an employee of York Capital Management ("York"), served as a member of the Company's Board, the Audit Committee, and the Compensation Committee of the Board, has resigned from the Board, and Win Rollins, who in his capacity as an employee of Elliott Management Corporation ("Elliott"), served as a member of the Company's Board, and the Compensation Committee of the Board, has resigned from the Board, each effective October 12, 2020. Mr. Skafidas and Mr. Rollins resignations were not the result of any disagreement with the Company on any matter relating to the operations, internal controls, policies or practices of the Company.

The Board does not intend to select members to replace Mr. Skafidas and Mr. Rollins, as permitted by the Company's bylaws, the Board plans to approve a decrease in the number of authorized directorships of the Board from five members to three members following the effective date.

David Rottino, Chief Executive Officer of Riviera, said: "On behalf of the Board, I would like to thank Thanasi and Win for all of the insights and support they have provided to both myself and the Company during their tenure as members of the Board."

Distributions to Shareholders

As a C corporation, distributions to common shareholders of current or accumulated earnings and profits are qualified dividends eligible for the 23.8% maximum federal income tax rate, inclusive of the 3.8% Medicare tax rate applicable to net investment income. Any distributions in excess of current or accumulated earnings and profits would be reported as returns of capital instead of qualified dividends. Distributions that are classified as returns of capital are nontaxable to the extent they do not exceed a shareholder's adjusted tax basis in the Company's stock, or as a capital gain to the extent that the amount of the distribution exceeds a shareholder's adjusted tax basis in the Company's stock. As of October 8, 2020, the Company estimates it will have zero current and accumulated earnings and profits for the tax year ended December 31, 2020. An updated estimate of Riviera's E&P will be provided in connection with publishing Form 8937 (which publication will occur within 45 days of the Distribution).

With the expected elimination of any current and accumulated earnings and profits, the Company reasonably estimates that a cash distribution should not constitute a taxable dividend for U.S. federal income tax purposes. Rather, a cash distribution would generally constitute non-taxable return of capital, and a reduction to the tax basis of each recipient's ownership interest in the Company, with any amount exceeding the holder's basis subject to capital gain treatment.

Under the Foreign Investment in Real Property Tax Act, non-U.S. persons who hold (or have held, during a certain measuring period) more than 5% of the Company's stock will be subject to withholding at a 15% rate on the full amount of the distribution. We have asked certain of our large, identifiable holders to certify that they are not subject to withholding under these rules or otherwise made withholding arrangements with such holders. We have not otherwise identified any additional non-U.S. 5% shareholders and do not intend to directly withhold from the distribution to any other holders on the basis of that diligence. However, while withholding under these rules is, under our circumstances, generally performed directly by the distributing company (to the extent withholding is necessary), we cannot guarantee that no intermediary will withhold from the distribution to the extent a particular beneficial holder is unable to demonstrate to the satisfaction of such intermediary that they are not subject to withholding under these rules.

Information regarding tax matters in this press release is for general information purposes only and does not constitute tax advice. Shareholders should consult with their tax advisors as to the specific U.S. federal and state, and non-U.S. tax consequences to such holder related to a return of capital distribution.

Forward-Looking Statements

Statements made in this press release that are not historical facts are "forward-looking statements." These statements are based on certain assumptions and expectations made by the Company which reflect management's experience, estimates and perception of historical trends, current conditions, and anticipated future developments. These statements include, among others, statements regarding the return of capital to shareholders, estimates of accumulated earnings and profits, our financial position, business strategy and other plans and objectives for future operations. Such statements are subject to a number of assumptions, risks and uncertainties, many of which are beyond the control of the Company, which may cause actual results to differ materially from those implied or anticipated in the forward-looking statements. These and other important factors could cause actual results to differ materially from those anticipated or implied in the forward-looking statements. Please read "Risk Factors" in the Company's Annual Report on Form 10-K, Quarterly Reports on Form 10-Q and other public filings. The Company undertakes no obligation to publicly update any forward-looking statements, whether as a result of new information or future events.

CONTACT:

Investor Relations (281) 840-4168 IR@RVRAresources.com

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL INFORMATION

On October 8, 2020, Riviera Resources, Inc., ("Riviera" or the "Company") completed the sale of its wholly owned subsidiary, Blue Mountain Midstream, LLC ("Blue Mountain" and "the Blue Mountain Divestiture") to Citizen Energy under a Purchase and Sale Agreement dated August 23, 2020, for net cash proceeds of approximately \$111 million. During the six months ended June 30, 2020, the Company recorded a noncash impairment charge of approximately \$18 million associated with Blue Mountain's crude oil gathering system assets. The Company anticipates recording an additional noncash impairment charge of approximately \$218 million in the third quarter of 2020 associated with Blue Mountain assets.

In addition, as previously reported, on October 1, 2020, the Company completed the sale of its interest in Mid-Continent properties located in Oklahoma (the "Mid-Continent Asset Sale") to Staghorn Petroleum II, LLC under a Purchase and Sale Agreement dated August 4, 2020, for net cash proceeds of approximately \$13 million. During the six months ended June 30, 2020, the Company recorded a noncash impairment charge of approximately \$88 million to reduce the carrying value of these assets to fair value.

The unaudited pro forma condensed consolidated balance sheet as of June 30, 2020, gives effect to the Blue Mountain Divestiture and the Mid-Continent Asset Sale as if they were completed as of June 30, 2020. The unaudited pro forma condensed consolidated statements of operations for the six months ended June 30, 2020, and for the year ended December 31, 2019, give effect to the Blue Mountain Divestiture and the Mid-Continent Asset Sale as if they were completed as of January 1, 2019.

The following unaudited pro forma condensed consolidated financial statements are derived from historical consolidated financial statements of the Company. The unaudited pro forma condensed consolidated financial statements are for informational and illustrative purposes only and are not necessarily indicative of the financial results that would have occurred if the transactions reflected therein had occurred on the dates indicated, nor are such financial statements necessarily indicative of the results of operations in future periods. The assumptions and estimates underlying the adjustments to the unaudited pro forma condensed consolidated financial statements are described in the accompanying notes. The unaudited pro forma condensed consolidated financial information should also be read in conjunction with the Company's historical financial statements and the notes thereto included in its Annual Report on Form 10-K for the year ended December 31, 2019, and its Quarterly Report on Form 10-Q for the quarter ended June 30, 2020.

The preparation of the unaudited pro forma condensed consolidated financial information is based on financial statements prepared in accordance with accounting principles generally accepted in the United States of America. These principles require the use of estimates that effect the reported amounts of revenues and expenses. Actual results could differ from those estimates.

The unaudited pro forma condensed consolidated financial information does not purport to represent what the actual results of operations would have been had the transaction occurred on the date assumed, nor is it necessarily indicative of the Company's future operating results. However, the pro forma adjustments reflected in the accompanying unaudited pro forma condensed financial information reflect estimates and assumptions that Company management believes to be reasonable.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET June 30, 2020 $\,$

(in thousands, except per share amounts)

	Riviera Resources, Inc. Historical	Blue Mountai Divestitur Pro Form Adjustmer	e a	A P	Mid- Continent Assets Sale Pro Forma Adjustments		Riviera Resources, Inc. Pro Forma	
ASSETS			_					
Current assets:								
Cash and cash equivalents	\$ 55,641	\$ 109,0		\$	13,418	(a)	\$ 102,63	37
		(75,4	, , ,					
Accounts receivable – trade, net	22,551	(10,1	40) <i>(c)</i>		(6,453)	(b)	5,95	
Derivative instruments	7,059	_	-		_		7,05	
Restricted cash	24,139	-	-		_		24,13	39
Other current assets	13,057	(6,7	68) (d)		(943)	(c)	5,34	1 6
Total current assets	122,447	16,6	70		6,022		145,13	39
Noncurrent assets:								
Oil and natural gas properties (successful efforts method), net	37,333	_	_		(11,554)	(d)	25,77	79
Other property and equipment, net	324,185	(318,1	25) (e)		(1,887)	(d)	4,17	73
Other noncurrent assets	5,555	(4,0			<u> </u>		1,47	73
Total noncurrent assets	367,073	(322,2	07)		(13,441)		31,42	25
Total assets	\$ 489,520	\$ (305,5	37)	\$	(7,419)		\$ 176,56	54
LIABILITIES AND EQUITY							-	=
Current liabilities:								
Accounts payable and accrued expenses	\$ 36,294	\$ (11,9	06) <i>(c)</i>	\$	(5,191)	(b)	\$ 19,19) 7
Derivative instruments	766	_	- ` ` `		`— ´	` '	76	6
Other accrued liabilities	9,490	(38) <i>(b)</i>		(1,131)	(c)	8,32	21
Total current liabilities	46,550	(11,9	44)		(6,322)		28,28	34
Noncurrent liabilities:			<u>—</u>					
Credit facilities	75,400	(75,4	00) (b)		_		_	-
Asset retirement obligations and other noncurrent liabilities	19,317	(1,8	32) (e)		(13,189)	(d)	4,29) 6
Total noncurrent liabilities	94,717	(77,2	32)		(13,189)		4,29) 6
Commitments and contingencies								
Equity:								
Preferred Stock, \$0.01 par value	_	_	_		_		_	-
Common Stock, \$0.01 par value	579	_	-		_		57	79
Additional paid-in capital	759,186				_		759,18	36
Accumulated deficit	(411,512)	(250,8	80) (f)		12,092	(e)	(615,78	31)
		34,5						
Total equity	348,253	(216,3			12,092		143,98	34
Total liabilities and equity	\$ 489,520	\$ (305,5	37)	\$	(7,419)		\$ 176,56	54

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF OPERATIONS SIX MONTHS ENDED JUNE 30, 2020

(in thousands, except per share amounts)

	Riviera Resources, Inc. Historical	Blue Mountain Divestiture Pro Forma Adjustments		Mid- Continent Assets Sale Pro Forma Adjustments		Riviera Resources, Inc. Pro Forma
Revenues and other:	d 05 500	ф		Φ (44.005)	(0)	ф. 4400 -
Oil, natural gas and natural gas liquids sales	\$ 25,732	\$ —		\$ (11,665)	(f)	\$ 14,067
Gains on commodity derivatives	6,721	(466)	(g)	_		6,255
Marketing revenues	55,786	(53,088)	(g)		(0)	2,698
Other revenues	40			(8)	(f)	32
	88,279	(53,554)		(11,673)		23,052
Expenses:						
Lease operating expenses	7,845	_		(4,904)	(f)	2,941
Transportation expenses	3,383	_		(2,191)	(f)	1,192
Marketing expenses	38,147	(38,083)	(g)	_		64
General and administrative expenses	21,123	(3,246)	(g)	_		17,877
Depreciation, depletion and amortization	15,112	(5,845)	(h)	(4,763)	(g)	4,504
Impairment of assets held for sale and long-lived assets	121,658	(17,856)	(i)	(87,683)	(h)	16,119
Taxes, other than income taxes	2,590	(869)	(j)	(819)	(i)	902
(Gains) losses on sale of assets and other, net	(2,031)	1,091	(j)			(940)
	207,827	(64,808)		(100,360)		42,659
Operating loss	(119,548)	11,254		88,687		(19,607)
Other income and (expenses):						
Interest expense, net of amounts capitalized	(1,668)	1,401	(k)	_		(267)
Other, net	(1,008)	290	(k)			(718)
	(2,676)	1,691		_		(985)
Reorganization items, net	(494)					(494)
Loss before income taxes	(122,718)	12,945		88,687		(21,086)
Income tax expense			(1)		(j)	
Net loss	\$ (122,718)	\$ 12,945		\$ 88,687		\$ (21,086)
Loss per share:						
Basic and diluted	\$ (2.11)					\$ (0.36)
Weighted average shares outstanding – basic and diluted	58,098					58,098

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF OPERATIONS YEAR ENDED DECEMBER 31, 2019

(in thousands, except per share amounts)

	Riviera Resources, Inc. Historical	Blue Mountain Divestiture Pro Forma Adjustments	Mid- Continent Assets Sale Pro Forma Adjustments			Riviera Resources, Inc. Pro Forma	
Revenues and other:							
Oil, natural gas and natural gas liquids sales	\$ 236,053	\$ —		\$	(43,036)	(f)	\$ 193,017
Gains on commodity derivatives	10,091	541	(g)		_		10,632
Marketing revenues	214,280	(159,706)	(g)		(7)	(f)	54,567
Other revenues	19,355				(93)	(f)	19,262
	479,779	(159,165)			(43,136)		277,478
Expenses:							
Lease operating expenses	77,719	_			(16,319)	(f)	61,400
Transportation expenses	64,149	_			(4,830)	(f)	59,319
Marketing expenses	166,651	(126,262)	(g)				40,389
General and administrative expenses	61,671	(16,634)	(g)		_		45,037
Exploration costs	5,122	_			_		5,122
Depreciation, depletion and amortization	77,089	(9,525)	(h)		(10,171)	(g)	57,393
Impairment of assets held for sale and							
long-lived assets	208,376	_					208,376
Taxes, other than income taxes	15,374	(1,383)	(j)		(1,966)	(i)	12,025
(Gains) losses on sale of assets and other, net	(20,743)	1,776	(j)				(18,967)
	655,408	(152,028)			(33,286)		470,094
Operating loss	(175,629)	(7,137)			(9,850)		(192,616)
Other income and (expenses):							
Interest expense, net of amounts capitalized	(6,997)	1,984	(k)		_		(5,013)
Other, net	(444)	798	(k)		_		354
	(7,441)	2,782					(4,659)
Reorganization items, net	13,359	_			_		13,359
Loss from continuing operations before income taxes	(169,711)	(4,355)			(9,850)		(183,916)
Income tax expense	127,859	27,775	(l)		(33,275)	<i>(</i> j)	122,359
Loss from continuing operations	\$ (297,570)	\$ (32,130)		\$	23,425	-	\$ (306,275)
Loss per share:							
Loss from continuing operations per share - basic and diluted	\$ (4.71)						\$ (4.85)
Weighted average shares outstanding - basic and diluted	63,118						63,118

The accompanying notes are an integral part of these unaudited pro forma condensed consolidated financial statements.

NOTES TO UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

Historical Riviera Resources, Inc.

These amounts represent the Company's condensed consolidated historical balance sheet and statements of operations. Amounts as of June 30, 2020, and for the six months ended June 30, 2020, were derived from the Company's Quarterly Report on Form 10-Q for the quarter ended June 30, 2020. Amounts for the year ended December 31, 2019, were derived from the Company's Annual Report on Form 10-K for the year ended December 31, 2019. The unaudited pro forma condensed consolidated financial information should be read in conjunction with the Company's historical financial statements and the notes thereto. During 2019 and 2020, the Company completed divestitures for which the related activity is not required to be reflected in the adjustments to the unaudited pro forma condensed consolidated financial statements included herein. Substantially all of the revenues reported on the historical condensed consolidated statements of operations included herein relates to properties that have been divested.

The historical consolidated financial statements have been adjusted in the unaudited pro forma condensed consolidated financial statements to give effect to pro forma events that are (i) directly attributable to the transaction, (ii) factually supportable and (iii) with respect to the unaudited pro forma condensed consolidated statements of operations, expected to have a continuing impact on the results following the transactions and events.

Note 1. Blue Mountain Divestiture Pro Forma Adjustments

- (a) Reflects cash proceeds received of approximately \$110 million, net of Blue Mountain cash balance of approximately \$941,000.
- (b) Reflects payment in full of the Blue Mountain senior secured revolving loan facility and accrued interest.
- (c) Reflects elimination of outstanding accounts receivable trade, net and accounts payable and accrued expenses.
- (d) Reflects elimination of other current assets.
- (e) Reflects elimination of property, plant and equipment, other noncurrent assets, asset retirement obligations and other noncurrent liabilities.
- (f) Reflects elimination of the Company's equity in Blue Mountain and the loss recognized from the divestiture.
- (g) Reflects elimination of revenues and operating expenses.
- (h) Reflects elimination of depreciation and amortization expenses.
- (i) Reflects elimination of noncash impairment charge recorded during the six months ended June 30, 2020, associated with Blue Mountain's crude oil gathering system assets.
- (j) Reflects elimination of taxes, other than income taxes and (gains) losses on sale of assets and other, net.
- (k) Reflects elimination of interest expense and other, net.
- (l) Reflects elimination of deferred income tax benefit, determined using the Company's statutory tax rate and the book and tax basis difference associated with Blue Mountain. There is no adjustment for the six months ended June 30, 2020, because a full valuation allowance was recorded against the Company's net deferred tax assets.

NOTES TO UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS - Continued

Note 2. Mid-Continent Asset Sale Pro Forma Adjustments

- (a) Reflects cash proceeds received of approximately \$13 million.
- (b) Reflects elimination of outstanding accounts receivable trade, net and accounts payable and accrued expenses.
- (c) Reflects elimination of other current assets and other accrued liabilities.
- (d) Reflects elimination of oil and natural gas properties, property, plant and equipment, other noncurrent assets, asset retirement obligations and other noncurrent liabilities.
- (e) Reflects net impact recognized from the sale of these assets.
- (f) Reflects elimination of revenues and direct operating expenses associated with these assets.
- (g) Reflects elimination of depreciation, depletion and amortization expenses.
- (h) Reflects elimination of noncash impairment charge recorded during the six months ended June 30, 2020.
- (i) Reflects elimination of taxes, other than income taxes.
- (j) Reflects elimination of deferred income tax expense, determined using the Company's statutory tax rate, associated with recording a full valuation allowance against the Company's net deferred tax assets. There is no adjustment for the six months ended June 30, 2020, because a full valuation allowance was recorded against the Company's net deferred tax assets.